THREE ESSAYS IN OPERATIONS MANAGEMENT

by

Yunxia Zhu

APPROVED BY SUPERVISORY COMMITTEE:

Dr. Milind Dawande, Co-Chair

Dr. Chelliah Sriskandarajah, Co-Chair

Dr. Ganesh Janakiraman

Dr. Vijay S. Mookerjee

Dr. Shun-Chen Niu



© Copyright 2012

Yunxia Zhu

All Rights Reserved



Dedicated to my parents, my wife, and teachers.





THREE ESSAYS IN OPERATIONS MANAGEMENT

by

YUNXIA ZHU, B.S., M.S.

DISSERTATION

Presented to the Faculty of

The University of Texas at Dallas

in Partial Fulfillment

of the Requirements

for the Degree of

DOCTOR OF PHILOSOPHY IN

MANAGEMENT SCIENCE

THE UNIVERSITY OF TEXAS AT DALLAS

August 2012



UMI Number: 3523760

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent on the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI 3523760

Copyright 2012 by ProQuest LLC.

All rights reserved. This edition of the work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 - 1346



ACKNOWLEDGMENTS

I would like to first express my deepest gratitude to my advisor, Professor Milind Dawande. His dedication to research and focus on details have provided me a great example of a true researcher. He always encouraged me to provide my maximum efforts on conducting the research. The preciseness he sought in analysis and writing has pushed me to become a better researcher. As a mentor, he was always available whenever I needed help either with my research or with my life. His valuable advice immensely helped me in transforming myself to a good teacher and a better person. I am also indebted to my co-advisor, Professor Chelliah Sriskandarajah for his guidance, understanding and supervision. His desire of connecting the academic research with the practice in real life has motivated several topics of my research. It was an honor for me to work with him. I appreciate his constructive feedback and advice throughout my stay at UT Dallas.

I feel fortunate to have worked with Professor Vijay Mookerjee. When I doubted myself, he believed in me and provided his support. I would like to thank Professors Srinagesh Gavirneni and Vaidy Jayaraman for working with me and providing their advice and encouragements during my job search. I also want to thank Professors Ganesh Janakiraman and Shun-Chen Niu for serving in my dissertation committee and helping me during my PhD study.

I am grateful to all the professors in Operations Management area for their guidance and support during my stay in UT Dallas as well as during my job search. I am indebted to



Professor Yunzeng Wang who introduced me to UT Dallas. The journey for my PhD would not have begun without his guidance.

I would like to give thanks to my dear friends Tharanga Rajapakshe, Mili Mehrotra and all my colleagues at UT Dallas, for being with me during these years.

Last but not least, I thank my parents for giving me life and loving me and supporting me unconditionally. I would also like to give my special thanks to my wife Xuan Zhang for her love.

June 2012



PREFACE

This dissertation was produced in accordance with guidelines which permit the inclusion as part of the dissertation the text of an original paper or papers submitted for publication. The dissertation must still conform to all other requirements explained in the "Guide for the Preparation of Master's Theses and Doctoral Dissertations at The University of Texas at Dallas." It must include a comprehensive abstract, a full introduction and literature review, and a final overall conclusion. Additional material (procedural and design data as well as descriptions of equipment) must be provided in sufficient detail to allow a clear and precise judgment to be made of the importance and originality of the research reported.

It is acceptable for this dissertation to include as chapters authentic copies of papers already published, provided these meet type size, margin and legibility requirements. In such cases, connecting texts which provide logical bridges between different manuscripts are mandatory. Where the student is not the sole author of a manuscript, the student is required to make an explicit statement in the introductory material to that manuscript describing the student's contribution to the work and acknowledging the contribution of the other author(s). The signatures of the Supervising Committee which precede all other material in the dissertation attest to the accuracy of this statement.



THREE ESSAYS IN OPERATIONS MANAGEMENT

Publication No.

Yunxia Zhu, Ph.D. The University of Texas at Dallas, 2012

Supervising Professors: Dr. Milind Dawande, Co-Chair Dr. Chelliah Sriskandarajah, Co-Chair

This dissertation addresses three important problems in operations management.

The aim of the first problem is to highlight – and demonstrate via specific examples – the need for algorithmic results for some fundamental set-based notions on which search in social networks is expected to be prevalent. To this end, we argue that the concepts of *an influential set* and *a central set* that highlight, respectively, the specific role and specific location of a set are likely to be useful in practice. We formulate two specific search problems: the Elite Group Problem (EGP) and the Portal Problem (PP), that represent these two concepts and provide a variety of algorithmic results.

In the second paper, we introduce two new multi-period models – designed specifically to capture the operations of a medium-size Depository Institution (DI) – that emerge from the DI's objective to minimize the total cost incurred in managing the inventory of cash over a finite planning horizon. The Basic Model (BM) captures the DI's mode of operations if it



chooses not to locally reuse cash and, instead, incur the cross-shipping penalty. The Reuse Model (RM) represents the DI's actions when it locally recirculates cash. We show that the *Value of Local Reuse* for a DI, measured as the percentage cost saving between the optimal solutions of BM and RM, is substantial.

The third problem investigates the concept of *Industrial Symbiosis*, a resource-sharing strategy among co-located firms to engage traditionally separate industries in a collective approach that involves physical exchanges of materials, water, energy and by-products. Inspired by a real-world example of a paper-sugar symbiotic complex, we study the impact on a firm's operational decisions from implementing an industrial symbiotic system. Our focus is on understanding how the introduction of competition and nature of competition will affect the firm's willingness to implement the symbiotic system.



TABLE OF CONTENTS

ACKNOWLEDGMENTS	v
PREFACE	/ii
ABSTRACT	iii
LIST OF TABLES x	iii
LIST OF FIGURES x	iv
CHAPTER 1 INTRODUCTION	1
1.1 Structural Search and Optimization in Social Networks	1
1.2 Value of Local Cash Reuse: Inventory Models for Medium-Size Depository	
Institutions under the New Federal Reserve Policy	2
1.3 Process Innovation via an Industrial Symbiotic System: The Impact of Com-	
petition on the Willingness to Implement	3
CHAPTER 2 STRUCTURAL SEARCH AND OPTIMIZATION IN SOCIAL NET-	
WORKS	5
2.1 Synopsis	5
2.2 The Notions of Influential Sets and Central Sets	8
2.2.1 The Elite Group Problem (EGP) and The Size-Constrained Elite Group	
Problem (SCEGP) $\ldots \ldots $	10
2.2.2 The Portal Problem (PP) and The Exact-Size Portal Problem (ESPP)	13
2.3 Algorithmic Analysis	14
2.3.1 The Elite Group Problem (EGP) $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	15
2.3.2 The Portal Problem (PP) $\ldots \ldots \ldots$	28



2.4	Concl	usions and Future Research Directions	43
CHAP	ΓER 3	VALUE OF LOCAL CASH REUSE: INVENTORY MODELS FOR MED	IUM-
SIZ	E DEP	OSITORY INSTITUTIONS UNDER THE NEW FEDERAL RESERVE]
PO	LICY		45
3.1	Synop	osis	45
	3.1.1	The Closed-Loop Cash Supply Chain	46
	3.1.2	Literature Review and Our Contributions	51
3.2	The H	Basic Model (BM)	55
	3.2.1	Problem Formulation	55
	3.2.2	Two Structural Results for BM	61
	3.2.3	Forward Recursion Function of DP and its Computational Complexity	68
	3.2.4	A Generalized BM (BM^g)	70
3.3	The I	Reuse Model (RM): Inventory Decisions under the Fed's Recirculation	
	Policy	7	77
	3.3.1	Problem Description	77
	3.3.2	Computational Complexity and a Structural Property of RM	81
	3.3.3	Formulating RM as a Mixed-Integer Program (MIP) $\ldots \ldots \ldots$	86
3.4	Under	rstanding the Value of Local Reuse and Influences on the Volume of	
	Cross	-Shipping	90
	3.4.1	The Test Bed	91
	3.4.2	Results	93
3.5	A Rol	lling-Horizon Procedure for Real-Time Decisions	98
	3.5.1	A $\tau\text{-}\mathrm{Period}$ Rolling-Horizon Problem Based on Forecasted Data $\ .$.	99
	3.5.2	A Rolling-Horizon Procedure	101
	3.5.3	The Performance of Procedure ROLLING HORIZON	102
3.6	The F	Fed's Policy and a Socially-Optimal Mechanism: A Comparative Analysis	104
3.7	Concl	usions	108



CHAPTER 4 PROCESS INNOVATION VIA AN INDUSTRIAL SYMBIOTIC SYS-TEM: THE IMPACT OF COMPETITION ON THE WILLINGNESS TO IMPLE-MENT 112

4.1	Synop	sis	112
	4.1.1	A Real-World Implementation of a Symbiotic System	113
	4.1.2	Differences from Traditional Process Innovation	115
	4.1.3	Our Goals and Contributions	118
	4.1.4	Literature Review	121
4.2	Model	l Setting	124
	4.2.1	Types of Product Variants and Consumers	125
	4.2.2	Structure of Demand	127
	4.2.3	Cost-Side and Demand-Side Impacts of Implementing a Symbiotic Sys-	
		tem	129
4.3	Analy	zing the Impact of the Symbiotic System under Monopoly and Compe-	
	tition		133
	4.3.1	Monopoly Setting (Model M) $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	133
	4.3.2	The Challenges in the Model Setting under Competition $\ldots \ldots \ldots$	140
	4.3.3	$\label{eq:Presence} \mbox{Presence of a Competitor When All Green Consumers Are Flexible} .$	143
	4.3.4	Presence of a Competitor When All Green Consumers Are Dedicated	152
4.4	Under	standing the Willingness to Implement the Symbiotic System	162
	4.4.1	The Willingness in a Monopoly	163
	4.4.2	The Willingness under Competition for the Regular Variants	167
	4.4.3	A Comparative Look at the Shift in Willingness under Competition $% \mathcal{A}$.	171
	4.4.4	Simultaneously Benefiting the Firm and Its Consumers	176
4.5	Direct	tions for Future Work	187
BEEE	RENCE	q	188
			100

VITA



LIST OF TABLES

3.1	Parameter Settings to Quantify the Value of Local Reuse and the Sensitivity of the Volume of Cross-Shipping.	93
3.2	Impact of the Length of the Rolling Horizon (τ) and Variation Width (w) on the Performance of Procedure ROLLING HORIZON.	103
4.1	Valuations of the Three Types of Consumers for Regular and Green Variants, for an Arbitrary Value of $v, 0 \le v \le \alpha$.	127
4.2	The Market Demand when Both Regular and Green Variants are Available	128
4.3	The Optimal Results Under Scenario M2	136
4.4	Optimal Results for Firm 1's Unconstrained Problem, under Scenario M3	137
4.5	The Two Firms' Payoffs when Both can Provide Regular and Green Variants.	158
4.6	Willingness to Implement the Symbiotic System Under Monopoly, Competi- tion for the Regular Variant, and Competition for Both Regular and Green Variants	171
4.7	Equilibrium Results under Competition for Regular Variants, both Before and After the Implementation of the Symbiotic System. Here, q_i^* is the Total Production Quantity of Firm $i; i = 1, 2, \ldots, \ldots, \ldots$	177
4.8	Equilibrium Results under Competition for both Regular and Green Variants, both before and after the Implementation of the Symbiotic System.	183



LIST OF FIGURES

2.1	An "Influence" Network and an Elite Group	11
2.2	Optimal Portals in Two Simple Networks.	13
2.3	Optimal Elite Group for a Rooted Down-Tree and a Rooted Up-Tree	15
2.4	A Bipartite Graph G with a Balanced Biclique, and Its Bipartite Complement Graph G^c , Which is Used in the Proof of Theorem 2.3.3.	21
2.5	The Widget, a Directed Cycle with Length m , Used in the Proof of Theorem 2.3.3.	21
2.6	The Constructed Graph G' for $SCEGP$	22
2.7	Graph G' with Elite Group Set W	23
2.8	Optimal Portal in a Biclique.	30
2.9	Optimal Portal in a Balanced and Full Binary ($d=2$) Tree	37
3.1	Life Cycle of Physical Cash	46
3.2	Impact of Imbalance and Length of Planning Horizon on Value and Extent of Local Reuse	94
3.3	Impact of Per-Bundle Fit-Sorting Cost (r) and Per-Bundle Holding Cost (h) on Cross-Shipping.	98
3.4	The Relative Benefit of Social Plan under Two Scenarios	106
4.1	A Pictorial Illustration of the Paper-Sugar Symbiotic Industrial Complex	114
4.2	Distribution of Consumers' Valuation for the Regular Variant.	127
4.3	The Mutually Beneficial Use of Waste in a Symbiotic System. The Figure Shows the Production Cycle of Products P and S. The Notation is as Defined at the Start of Section 4.2.3.	131
4.4	Categorization of Equilibria Based on the Competitor's Variable Production Cost (c_2) , when $a > 1$, $c'_1 < \frac{\theta_1 \alpha}{2(1+\theta_1)}$, and $c'_1 < c_2 < \frac{1}{2} [\alpha + \frac{c'_1}{a(1-\theta_1)+\theta_1}]$	144



 xiv

4.5	Willingness in a Monopoly: Influence on the Decision to Implement of the Change in the (i) Variable Production Cost after Implementation $(c'_{1,p})$, and (ii) Proportion of Regular Consumers of Product P $(\theta_{1,p})$	165
4.6	Willingness under Competition for Regular Variants: Influence on the Decision to Implement of the Change in the Appreciation of the Green Variant of Product P (a_p) .	169
4.7	A Pictorial Representation of the Conditions in Theorems 4.4.7, 4.4.9, and 4.4.11.	182



CHAPTER 1

INTRODUCTION

1.1 Structural Search and Optimization in Social Networks

The explosive growth in the variety and size of social networks has focused attention on searching these networks for useful structures. Like the internet or the telephone network, the ability to efficiently search large social networks will play an important role in the extent of their use by individuals and organizations alike. However, unlike these domains, search on social networks is likely to involve measures that require a *set* of individuals to collectively satisfy some skill requirement or be tightly related to each other via some underlying social property of interest.

The aim of this paper is to highlight – and demonstrate via specific examples – the need for algorithmic results for some fundamental set-based notions on which search in social networks is expected to be prevalent. To this end, we argue that the concepts of *an influential set* and *a central set* that highlight, respectively, the specific role and specific location of a set are likely to be useful in practice. We formulate two specific search problems: the Elite Group Problem (EGP) and the Portal Problem (PP), that represent these two concepts and provide a variety of algorithmic results. We first demonstrate the relevance of EGP and PP across a variety of social networks reported in the literature. For simple networks (e.g., structured trees and bipartite graphs, cycles, paths, etc.), we show that an optimal solution to both EGP and PP is easy to obtain. Next, we show that EGP is polynomially solvable



on a general graph while PP is strongly NP-hard. Motivated by practical considerations, we also discuss (1) a size-constrained variant of EGP together with its penalty-based relaxation and (2) the solution of PP on balanced and full *d*-trees and general trees.

1.2 Value of Local Cash Reuse: Inventory Models for Medium-Size Depository Institutions under the New Federal Reserve Policy

The effective local reuse of physical cash by Depository Institutions (DIs) is the primary goal of the new cash recirculation policy of the Federal Reserve (Fed) of the United States. These guidelines, implemented since July 2007, encourage the reuse of cash by (1) penalizing a DI for the practice of *cross-shipping*: near-simultaneous deposit of used cash to – and withdrawal of fit cash from – the Fed and (2) offering a *custodial inventory* program that enables a DI to transfer fit cash to the Fed's books, but physically hold it within the DI's secured facility. The effective management of the inventory of cash under these new guidelines is both a challenging and important issue for DIs.

We introduce two new multi-period models – designed specifically to capture the operations of a medium-size DI – that emerge from the DI's objective to minimize the total cost incurred in managing the inventory of cash over a finite planning horizon. The Basic Model (BM) captures the DI's mode of operations if it chooses not to locally reuse cash and, instead, incur the cross-shipping penalty. Using two important structural properties, we provide a polynomial-time dynamic programming algorithm for BM. The Reuse Model (RM) represents the DI's actions when it locally recirculates cash. We first prove the hardness of RM and then develop an integer programming formulation. A comprehensive test bed – based



on our interaction with a leading secure-logistics provider – helps us to develop several useful insights into the relative impacts of the DI-specific parameters and the Fed's cross-shipping fee on the effective management of cash. In particular, we show that the *Value of Local Reuse* for a DI, measured as the percentage cost saving between the optimal solutions of BM and RM, is substantial and analyze the forces that influence the volume of cross-shipping. We also develop a rolling-horizon procedure to adapt the optimal solutions of BM and RM for obtaining near-optimal real-time solutions in the presence of a modest amount of uncertainty. Finally, we provide a comparative analysis of a DI's decisions under the Fed's mechanism and those under a socially-optimal mechanism.

1.3 Process Innovation via an Industrial Symbiotic System: The Impact of Competition on the Willingness to Implement

Industrial Symbiosis is defined as a resource-sharing strategy among co-located firms to engage traditionally separate industries in a collective approach that involves physical exchanges of materials, water, energy and by-products. Inspired by a real-world example of a paper-sugar symbiotic complex, we study the impact on a firm's operational decisions from implementing an industrial symbiotic system. The two products manufactured by the firm are symbiotically connected, in the sense that the waste from the manufacture of one product is used as a raw material for the second product, and vice-versa. We characterize the firm's operational optimal/equilibrium decisions for its two products – both in the presence and absence of a symbiotic system – under monopoly as well as under competition. The focus of our analysis is on understanding the behavior of the firm's "willingness" to implement.



Our models capture the supply-side (e.g., incurring a fixed cost and changes in variable production costs) as well as the demand-side influence of implementing the symbiotic system. The implementation enables the firm to label its two products as "green" (environmentfriendly). The consumers who have a relatively higher valuation for the green variants are further differentiated, based on their loyalty to the green variants. We consider two settings of Cournot competition – depending on whether or not competitors can produce the green variants – in the presence of multiple types of consumers and multiple variants. The difference in the firm's total profits before and after implementation is used as a metric to capture its willingness to implement the system. Three dominating forces that influence the firm's strategic decision to implement the system are (1) the proportion of the green consummers in the market, (2) consumers' appreciation for the green variants, and (3) changes in the variable production costs after implementation. When the firm's variable production costs remain similar before and after implementation, then (1) if the proportion of green consumers is high, then the arrival of competitors who only produce regular variants encourages the implementation of a symbiotic system, (2) if consumers' appreciation of the green variants is low, then the firm's willingness to implement is more (than in a monopoly) under competition from firms who produce both variants. Also of interest are situations when competition incentivizes the firm to implement and both the firm and consumers benefit from the implementation. We identify two such scenarios (one for each type of competition), when the fixed cost of implementing the system is modest.



CHAPTER 2

STRUCTURAL SEARCH AND OPTIMIZATION IN SOCIAL NETWORKS *

2.1 Synopsis

A social network represents a social structure as a set of definite relationships between the members – entities or groups – of a social system. In its most commonly used representation, a social network can be viewed as a network of nodes (individuals, organizations, web pages, etc.) related to one another using edges (friendship, commercial transactions, url links, etc.). Over the years, social networks have been used to analyze social phenomena in a wide variety of domains, including sociology, epidemiology, social psychology, economics, anthropology, history, and human geography (Scott 2000, Wasserman and Faust 1994, Brandes and Erlebach 2005). Often in social network analysis the interest is to explain individual or group behavior in the context of the larger social structure in which the individual or group is situated.

More recently, "social networking sites" such as Facebook (http://www.facebook.com) and Myspace (http://www.myspace.com) have proliferated on the internet and help users connect based on a wide range of interests and practices. While some sites support the maintenance of pre-existing social networks, others help strangers connect based on their shared interests

^{*}Accepted for Publication, Milind Dawande, Vijay Mookerjee, Chelliah Sriskandarajah, Yunxia Zhu, Structural Search and Optimization in Social Networks, INFORMS Journal on Computing, the Institute for Operations Research and the Management Sciences, 7240 Parkway Drive, Suite 300, Hanover, Maryland 21076 USA.



and/or activities. Some sites cater to diverse audiences while others attract people based on some shared identity (Boyd and Ellison 2007). Typically, the participants (players) of the network derive some utility from the network, for example, to find each other for exchanging ideas, solving problems, companionship, and so on.

It should be clear that, like any other network-based phenomenon such as the telephone or the internet, the ability of the individual or group to derive value depends on the ability to search the network for contacts. For example, searching the telephone network is facilitated by a phone directory, browsing the internet requires a browser and a search engine, and so on. Many researchers believe that the advent of the web browser and search engine was most influential to the explosive growth of the internet. By analogy, it can be proposed that the utility of social networks to individuals and organizations will also depend on the ability to search the networks of interest for useful structures. For example, a participant in Facebook may want to discuss a topic of interest and may need to call upon a selected subset of friends to join the discussion. In the Open Source community, individual developers form a social network by virtue of having worked on common projects. In such a community, a developer or a firm may want to create a project-team of members with certain specialized skills and access to resources.

Searching a social network often creates search problems that are different from those encountered in other network phenomena like the web or the telephone network. In the web, the typical nature of search is to provide the user with a set of web sites that match based upon a list of search terms. There is usually no requirement that the web sites returned by the search engine satisfy some complex relationship to one another, other than, of course,



the trivial relationship that they must all match (to varying degrees) with the list of search terms.

On the other hand, search problems in a social network can be more complex. In particular, the search results may often need to satisfy a *set* measure. For example, in extracting a project-team from a larger network, it may be important that the set of developers that are returned collectively satisfy some skill requirements, but, in addition, are tightly related to one another by virtue of having worked on common projects. With the improvement in computing technology, the data and the tools needed to identify the network of interest are readily available. From a technical perspective, when the results of a search need to meet (or exceed) a specified set measure (specifically, a non-additive measure), the search often becomes combinatorial in nature. Search problems in social networks therefore provide a challenging ground for researchers interested in applying graph-theoretic, algorithmic methods to the area. Our interest in this study stems from the new problems and opportunities that are likely to arise for the use of graph-theoretic methods to solve interesting search problems in social networks.

The remainder of the paper is organized as follows. In Section 2.2, we argue that two set-based notions – influential sets and central sets – are likely to provide a fundamental structural basis for important search problems arising in a variety of practical social networks, and introduce two optimization problems – EGP and PP – corresponding to these two notions. Section 2.3 investigates the complexities of these problems on several special graphs as well as on general graphs. Section 2.4 concludes the paper and provides directions for future research.



2.2 The Notions of Influential Sets and Central Sets

Given the significance of search in social networks and, consequently, the need for efficient algorithms, an important question arises naturally: What are some fundamental set-based notions on which search in social networks is expected to be prevalent? Traditionally, in social network analysis, two fundamental properties of individual members – their *location* and their *role* in the network – have proven to be fundamental. This is natural, since these two properties provide insights into the groupings and interactions in the network. Accordingly, for individual members of a social network, network centrality measures, including *Degree Centrality, Closeness Centrality*, and *Betweenness Centrality*, have been heavily investigated and used (see, e.g., Freeman 1979; Brandes and Erlebach 2005, Chapters 3-5). For set-based search too, structures and measures that highlight the specific role or specific location of a set are likely to be the most useful in practice. The need and use of such set-based measures has already been documented in recent studies. For example, the notions of group (or set) betweeness and group degree centralities are discussed in Chapter 4 of Carrington et al. (2005) and in Everett and Borgatti (1999).

The motivation to study the role played by members in a network has to do with understanding the influence a member can potentially cast over other members in the network. Such notions of influence exerted by a single member can intuitively be extended to the influence a *set* of members can potentially exert over the rest of the group. A set of influential members may be useful to identify for a variety of reasons, often having to do with wanting to promote an idea, product, or message to other members of the network. For example, a firm may wish to advertise a new product or service and use an influential group



of members to help in this cause (Hill et al. 2006). Similarly, a welfare organization may want to disseminate ideas of social importance within a community of interacting members and use an influential set of members for spreading the message in an effective and timely manner. Another purpose to study influential groups is often to identify a set of members who possess specialized knowledge or information pertaining to a specific domain, namely, the key *experts* in the group. For example, it may be important to identify a set of expert oncologists for devising an informed, yet balanced plan of action to treat a difficult case. Here, a *set* of experts may be especially relevant to consult to eliminate or reduce bias as well as to surface fresh perspectives that can aid in problem solving.

The motivation to study the location of a member (or a set of members) is subtly different from that of examining member roles. Location is essentially a topological characteristic that has to do with a member or a set of members acting to facilitate contact between other interacting members of the network. A centrally located member is *well connected*, or, in other words, has better access to other members by virtue of acting as a conduit that allows exchanges and flows of information or ideas in the network. A central location does not necessarily imply influence, neither does an influential member necessarily need to be centrally located. Indeed, recent research in Reality Mining (Pentland 2004, Greene 2008, Hesseldahl 2008) and interaction within social networks reveals significant distinctions between these two concepts. For example, managers who may be influential within a business organization usually do not play a central role in the routing of communications between teams (Gloor et al. 2007, Thompson 2008). The players central for communication could, instead, be less influential employees. The question arises: what property does location convey that is



useful to a problem solver? One benefit of identifying centrally located members is that it provides one with an understanding of the paths that are heavily used in the network so that sufficient resources can be made available at these locations to avoid communication bottlenecks from occurring. An interesting variant is one where the problem solver may *want* to thwart communication: the activities of a terrorist group may be significantly impaired by striking at locations or members that are central to the flow of communication within the network (Erickson 1981).

The twin notions of influence and centrality admit a variety of interpretations, depending on the context of the social network under consideration. Accordingly, there can be several meaningful measures to evaluate "good" influential and central sets. For example, to measure the centrality of a set of vertices, the classical measure of *betweenness centrality* of a single vertex has been extended to group betweenness centrality (Everett and Borgatti 1999) and *co-betweenness centrality* (Kolaczyk et al. 2009). Along this theme, we propose two specific measures: one for an influential set and the other for a central set. We now describe two optimization problems that correspond to these two measures, discuss their origins, and provide examples of social networks where these problems are relevant.

2.2.1 The Elite Group Problem (EGP) and The Size-Constrained Elite Group Problem (SCEGP)

Technical Definition

<u>INSTANCE</u>: *n* players; an "influence" social network represented by a directed graph G(V, A), |V| = n, in which the nodes represent the players and the set of arcs represent pairwise



influences pertaining to a social property: a directed arc (i, j) indicates that i is influenced by j. For SCEGP, a positive integer $k \leq n$ is also given.

SOLUTION OF EGP: A set $W \subseteq V$ such that there does not exist a directed arc $(i, j) \in A$ with $i \in W, j \notin W$.

<u>SOLUTION OF SCEGP</u>: Same as EGP, with the additional requirement that $|W| \leq k$. <u>OBJECTIVE FUNCTION</u>: Maximize the total number of directed arcs, γ_W , incident on any node in W from nodes in $V \setminus W$. More precisely, the *score* γ_W is defined as follows: $\gamma_W = \sum_{i \notin W, j \in W} a_{ij}$, where $a_{ij} = 1$, if $(i, j) \in A$; 0 otherwise.

Note that in a graph G(V, A), there is at least one feasible solution for EGP, namely the complete set of nodes V, with score $\gamma_V = 0$. Also, observe that adding more nodes to an elite group does not necessarily increase the number of directed arcs into the group. If a node, say j, is added to an elite group W, then to obtain $\gamma_{W \cup \{j\}}$, we (i) add to γ_W the number of arcs from $V \setminus \{W \cup \{j\}\}$ to j and (ii) subtract the number of arcs from node j to the nodes in W. Thus, depending on this tradeoff, $\gamma_{W \cup \{j\}}$ can be either larger or smaller or the same as γ_W .



Figure 2.1. An "Influence" Network and an Elite Group.



Origin and Applications

The notion of an "elite" group originated from efforts to examine and understand social behavior within a close-knit community. In the 1980s, Sociologist Li Fan analyzed the giving (and receiving) of gifts between the residents of a Mongolian town (Wellman et al. 2001), and found that one (elite) block of residents received gifts from the others but only exchanged gifts among each other. Thus, as a set, this group of residents only received gifts from the other members of the town.

Another example of the notion of an elite group occurs in the analysis of the advice-seeking behavior of the members of a school, reported in Hawe and Ghali (2008). Here, the social network revealed that, together, the Principal, the Vice-Principal, and some key technical staff, form a group with the properties that (1) most of the other staff members seek advice from one or more members of this group and (2) the members of the group typically seek advice only from (one or more) members within the group. Thus, to influence opinion within the community in general, it may be beneficial to first convince this group of individuals.

In the context of social network analysis, the members of an elite group can be regarded as key players or opinion leaders. For instance, if a member of a community is often consulted by other members on (say) health issues, then she can be regarded as a key player (an "elite" member) in the opinion-seeking network of that community (Borgatti 2006). Another example is the co-sponsorship network in the United States Senate (Fowler 2006). In this network, the prominent senators typically receive a significant amount of co-sponsorship. Thus, the set of these prominent senators constitute an (approximate) elite group.



2.2.2 The Portal Problem (PP) and The Exact-Size Portal Problem (ESPP)

Technical Definition

<u>INSTANCE</u>: *n* players; a connected, undirected graph G = (V, E), |V| = n, in which the nodes represent the players and edges represent the pairwise connections between the players; a positive integer $k \leq n$.

SOLUTION: For PP, a set $Q \subseteq V$ such that $|Q| \leq k$. For ESPP, a set $Q \subseteq V$ such that |Q| = k.

<u>OBJECTIVE FUNCTION</u>: Maximize r(Q), defined as follows:

$$r(Q) = \frac{BC(Q)}{\binom{n-|Q|}{2}} \text{ and } BC(Q) = \sum_{s \notin Q, t \notin Q, s \neq t} \frac{\sigma_{st}(Q)}{\sigma_{st}}$$

where σ_{st} is the total number of shortest paths from node s to node t; $s, t \in V \setminus Q, s \neq t$, and $\sigma_{st}(Q)$ is the number of shortest paths from node s to node t which have at least one node in set Q as an internal node.



Figure 2.2. Optimal Portals in Two Simple Networks.

Previous Work and Applications

PP is a natural extension of the popular Betweenness Centrality (BC) measure (Freeman



1979, Scott 2000) for individual nodes (members) of a social network; for k = 1, an optimal solution to PP is a node with the highest BC. Everett and Borgatti (1999) extend the notion of BC to groups, and illustrate the measure on a few examples. For a *given* set of nodes Q, Puzis et al. (2007) provide a polynomial-time algorithm to compute the nonnormalized measure BC(Q) (referred to as "Group Betweenness Centrality"). They prove that the problem of obtaining a set with the highest Group Betweenness Centrality (i.e., $\max_{Q \subseteq V} BC(Q)$ is NP-hard and also propose a simple heuristic. Note that the behavior of our normalized measure r(Q) can be fundamentally different from that of BC(Q). In general, the set which maximizes the Group Betweenness Centrality may not necessarily be an optimal solution of our model, and vice-versa. Puzis et al. (2007) discuss an interesting application of a network of computers in which a limited number of virus-cleaning devices need to be placed at a subset of nodes (computers) to prevent the spread of viruses. To maximize the utility of the devices, it is beneficial to place them at the nodes of a portal of an appropriate size. Another interesting application where a portal may need to be identified is in a disease-outbreak network. For example, Klovdahl et al. (2001) describe a TB-outbreak network and motivate the need to identify the critical members in this network to control the spread of the disease. Everett and Borgatti (1999) discuss the interaction network of animals (monkeys) and use the notion of a portal to determine a socially central set of animals.

2.3 Algorithmic Analysis

We now analyze EGP and PP. For a search problem, a basic question is that of its computational complexity. For simple networks, an optimal solution to both problems is easy to



obtain. For EGP, we first illustrate this and then identify a structural property of an elite group that can help in reducing the size of the underlying graph. Then, we show that EGP is polynomially solvable for a general network. Next, motivated by practical considerations, we introduce a size-constrained version of EGP together with its penalty-based relaxation and show that both are strongly NP-hard. For PP, we first show that PP is strongly NP-hard on a general graph. We then consider several special graphs on which PP is polynomially solvable. Finally, we discuss a heuristic for general trees.

2.3.1 The Elite Group Problem (EGP)

Given a directed graph G(V, A), recall that an elite group is a set $W \subseteq V$ such that there does not exist any directed arc $(i, j) \in A$ with $i \in W$, $j \notin W$. The objective of EGP is to maximize the total number (or score), γ_W , of directed arcs incident on the nodes in W. For some simple networks, it may be straightforward to prove the optimality of a specific elite group. Rooted up- and down-trees are especially useful networks to study because they represent hierarchically organized structures, e.g., reporting relationships in a department, natural taxonomies, etc. (Cross and Parker 2004).



Figure 2.3. Optimal Elite Group for a Rooted Down-Tree and a Rooted Up-Tree.



Observation 2.3.1 If the graph G is a rooted down-tree (i.e., each node in G, except the root, has a unique predecessor and all arcs in G are directed downwards from the root to the leaf nodes. See Figure 2.3), then the elite group W^* consisting of all the leaf nodes of G is an optimal elite group. If the graph G is a rooted up-tree (i.e., each node in G, except the root, has a unique successor and all arcs in G are directed upwards from the leaf nodes towards the root. See Figure 2.3), then the elite group W^* consisting of all non-leaf nodes of G is an optimal elite group.

Proof: First, note that the root is not included in an optimal elite group; for otherwise, each node of G is in the elite group and the score is 0, which is clearly a non-optimal solution for any non-trivial rooted down-tree G. Consider an optimal elite group W which contains a non-leaf node t such that the unique predecessor of t is not in W. Note that all descendants of t are also in W. Let $n_t \ge 1$ be the number of direct descendants of t in G. Then, removing t from W results in a feasible elite group $W' = W - \{t\}$ with score $\gamma_{W'} = \gamma_W + (n_t - 1) \ge \gamma_W$. Continuing, we can similarly remove all non-leaf nodes from W without decreasing the score to obtain an elite group consisting only of leaf nodes. Thus, there exists an optimal elite group W^* consisting only of leaf nodes. Finally, note that W^* must contain *all* leaf nodes. This follows since including a leaf node strictly increases the score of an elite group.

The proof for a rooted up-tree is similar.

Our next result helps us "shrink" the strongly connected components (e.g. directed cycles) in G to single nodes in our search for an elite group. We will use this result later in the proof of Theorem 2.3.3.



Proof: Consider a strongly connected component C which includes nodes v_1, v_2, \ldots, v_n . Suppose, without loss of generality, $v_1 \in W$. Since there is a directed path from v_1 to v_2 , node v_2 must belong to W as well. Continuing this argument, nodes v_3, v_4, \cdots, v_n must belong to W. Similarly, if, say, $v_1 \in \overline{W}$, then $\{v_2, v_3, \ldots, v_n\} \subseteq \overline{W}$. For otherwise, if $v_j \in W$ for some $j \in \{2, 3, \ldots, n\}$, then $v_1 \in W$. The result follows.

Note that there are many polynomial algorithms to find a strongly connected component (if one exists) in a graph. If G contains a strongly connected component C, then, by using Observation 2.3.2, we can shrink C into a single node. An arc in the original graph between a node $v \in V \setminus C$ and a node of C is represented in the *shrunk graph* as between v and the (shrunk) node representing the component. Thus, in the shrunk graph, we use a separate new arc to represent each arc between a node in $V \setminus C$ and a node of C in the original graph. Therefore, in general, the shrunk graph becomes a multigraph since there may be parallel arcs between two nodes. We can continue this type of shrinking until there is no non-trivial strongly connected component in the modified shrunk graph. Consequently, we can assume without loss of generality that the network is a Directed Acyclic Graph (DAG). Therefore, EGP translates into finding a sink set of maximum indegree in a DAG. The following result follows immediately from Observation 2.3.2.

Lemma 2.3.1 There is a one-to-one correspondence between elite groups in G and elite groups in the shrunk graph: For every elite group W in G, we can get one in the shrunk



graph with the same score by taking the nodes of the shrunk graph corresponding to all the strongly connected components in W. Conversely, for every elite group W' in the shrunk graph, we can get one in G with the same score by taking the nodes in all the strongly connected components corresponding to the nodes in W'.

Next, we show that EGP is polynomially solvable.

Theorem 2.3.2 The EGP is polynomially solvable.

Proof: For $j \in V$, define $\pi_j \in \{0, 1\}$ as follows:

$$\pi_j = \begin{cases} 1, & \text{node } j \text{ belongs to the elite group } W_j \\ 0, & \text{otherwise.} \end{cases}$$

Then, an integer programming (IP) formulation for EGP is as follows:

$$\max \sum_{\substack{(i,j) \in A \\ s.t.}} (\pi_j - \pi_i)$$

$$s.t. \quad \pi_i - \pi_j \leq 0, \quad \forall \ (i,j) \in A$$

$$\pi_i \in \{0,1\}, \quad \forall \ i \in V$$

For a directed arc (i, j), if node *i* is in *W* (i.e., $\pi_i = 1$), then node *j* must also be in *W* (i.e., $\pi_j = 1$). Otherwise, if $\pi_i = 0$, then $\pi_j \in \{0, 1\}$. This is enforced by the first constraint. In the objective function, $(\pi_j - \pi_i)$ represents the contribution of arc (i, j) to γ_W : if nodes *i* and *j* are both in *W* or both in *V**W* (i.e., $\pi_i - \pi_j = 0$), then the contribution is 0. If node *i* is in *V**W* and node *j* is in *W* (i.e., $\pi_i = 0, \pi_j = 1$), then the contribution is 1. The constraints of the IP can be written as $A\pi \leq 0$, where *A* is the node-arc incidence matrix of *G* and $\pi \in \{0, 1\}^{|V|}$. It is well-known that the node-arc incidence matrix of a *directed* graph is totally unimodular (see, e.g., Hoffman and Kruskal 1956, Nemhauser and Wolsey 1988).



Thus, the linear programming relaxation of the above IP results in an integer optimum. The result follows.

Note that the shrinking of strongly connected components (Lemma 2.3.1) maintains the total unimodularity of the constraint matrix of the IP above. Thus, the size of a network containing strongly connected components can be reduced before formulating the EGP. The objective function of the IP above is to maximize the number of arcs directed into the elite group; instead, if we change the objective function to maximize a weighted linear combination of arcs directed into the elite group, the modified problem remains polynomially solvable.

Remark 1: (penalizing the size of the elite group) Note that EGP does not impose any constraint on the cardinality (i.e., the number of nodes) of the elite group. In Section 2.3.1, we consider a hard constraint on the cardinality. An alternative is to impose a "soft" constraint by imposing a penalty p > 0 on the cardinality. In this case, the objective function in the IP above changes to $\max \sum_{(i,j) \in A} (\pi_j - \pi_i) - p \sum_{i \in V} \pi_i$. Since this modified objective is linear and the constraint matrix is totally unimodular, the modified problem remains polynomially solvable.

Size-Constrained Elite Group Problem (SCEGP)

Typically, the purpose of identifying an elite group is to use the members of this group to effectively influence the other members of the social network. Thus, for practicability in managing this subsequent task, the size of an elite group may need to be restricted. Motivated by this requirement, Theorem 2.3.3 discusses the complexity of the *Size-Constrained*


Elite Group Problem (SCEGP), defined as follows: Given a positive integer $k \leq n$, find an optimal elite group $W \subseteq V$ with $|W| \leq k$.

Theorem 2.3.3 The decision problem corresponding to SCEGP is strongly NP-Complete.

Proof: The strongly NP-Complete problem which we use in our reduction is the Balanced Biclique Problem (Garey and Johnson 1979), defined as follows.

Balanced Biclique Problem (BBP)

Instance. An undirected Bipartite Graph $G = (U \cup V, E)$, with |U| = |V| = n. A positive integer $k \leq n$.

Solution. An induced subgraph $G_1 \subseteq G$ such that $G_1 = (U_1 \cup V_1, E_1), U_1 \subseteq U, V_1 \subseteq V, |U_1| = |V_1| = k, E_1 \subseteq E$, and $u_1 \in U_1, v_1 \in V_1$ implies that $\{u_1, v_1\} \in E_1$. The size of the biclique is 2k.

Given an arbitrary instance of BBP specified by G, we construct an instance of SCEGP on a related graph G'. The construction of G' is done in two steps. First, we obtain G^c , the bipartite complement graph of G. Then, we add two additional node sets O and S, extend each node in U into a directed cycle, and give directions to all edges to get G'. We now explain our construction and illustrate with an example of G in Figure 2.4:

Step 1. Get G^c , the bipartite complement graph of G (see Figure 2.4).

<u>Step 2</u>. We add two node sets O and S consisting, respectively, of n^3 and n^2 nodes. The nodes of O (resp., S) form a directed cycle. There is a directed arc from each node $o_i \in O$ to each node in U. There is a directed arc from each node in V to each node $s_i \in S$. Let $m = n + n^2$. Next, we extend each node $u_i \in U$ into a length m directed cycle



20



Figure 2.4. A Bipartite Graph G with a Balanced Biclique, and Its Bipartite Complement Graph G^c , Which is Used in the Proof of Theorem 2.3.3.



Figure 2.5. The Widget, a Directed Cycle with Length m, Used in the Proof of Theorem 2.3.3.

 C_i by adding m - 1 additional nodes $(u_{i,1}, u_{i,2}, \dots, u_{i,m-1})$ (see Figure 2.5). Let $U' = \{u_i, u_{i,1}, u_{i,2}, \dots, u_{i,m-1} | u_i \in U, i = 1, 2, \dots, n\}$. The edges between O and U' are directed from O to U', those between U' and V are directed from U' to V, and those between V and S are directed from V to S. The construction of G' is now complete (see Figure 2.6). Let $N = O \cup U' \cup V \cup S$. On G', consider the following decision question for SCEGP:

<u>DECISION QUESTION</u>: Let $t = km + (n - k) + n^2$ and $D = kn^3 + kn^2$. Does there exist an elite group W in G' such that $|W| \le t$, and $\gamma_W \ge D$?

Note that the construction of the decision problem from the given instance of the BBP is polynomially bounded. That is, the total number of nodes in G' is bounded by a polynomial in n, as is the time necessary to construct a description of the input of the decision problem.





Figure 2.6. The Constructed Graph G' for SCEGP.

The decision problem is clearly in class NP. We now show that the decision question has an affirmative answer if and only if the original graph G contains a balanced biclique of size 2k (i.e, $|U_1| = |V_1| = k$).

⇒ Suppose $U_1 \cup V_1$ is a balanced biclique of size 2k in G. Let $U_2 = U \setminus U_1, V_2 = V \setminus V_1$. In G', let $U'_1 = \{C_i | u_i \in U_1\}, U'_2 = \{C_i | u_i \in U_2\}, W = U'_1 \cup V_2 \cup S, \overline{W} = O \cup U'_2 \cup V_1$ (see Figure 2.7). We now show that the set W is an elite group that provides an affirmative answer to the decision question.

First we need to prove the set W is a valid elite group in G', i.e., there is no arc from W to \overline{W} . Since $U_1 \cup V_1$ is a biclique of G, then there is no arc from U'_1 to V_1 in G'. Since G is bipartite, there is no arc between U'_1 and U'_2 . Also, by construction, there is no arc from U'_1 to O. Thus, there is no arc from U'_1 to \overline{W} . Similarly, there is no arc from V_2 to \overline{W} and from S to \overline{W} . Thus, W is a valid elite group.

Next, observe that $|W| = |U'_1| + |V_2| + |S| = km + (n - k) + n^2 = t$. Finally, note that γ_W is the number of arcs from \overline{W} to W. The number of arcs from O to U'_1 (respectively, V_1 to





Figure 2.7. Graph G' with Elite Group Set W.

S) is kn^3 (respectively, kn^2). Also, the number of arcs from U'_2 to V_2 is nonnegative. Thus, $\gamma_W \ge kn^3 + kn^2 = D$. The result follows.

 \Leftarrow Suppose W is an elite group in G' with $|W| \leq t$ and $\gamma_W \geq D$. Let $\overline{W} = N \setminus W$. The following claims characterize the set W.

Claim 2.3.1 In G', the nodes in C_i either all belong to W or all belong to \overline{W} . Similarly, the nodes in S (respectively, O) either all belong to W or all belong to \overline{W} .

Proof: The nodes in C_i (respectively, S, O) form a directed cycle. The result follows from Observation 2.3.2.

Claim 2.3.2 Each node in O must belong to \overline{W} . Similarly, each node in S must belong to W.

Proof: Suppose a node in O belongs to W. Then, from Claim 2.3.1, each node in O belongs to W. Also, from the definition of elite group, each node in U' must belong to W. Consequently $|W| \ge |O| + |U'| = n^3 + nm$. Since $n \ge 2$ and $n \ge k$, we have $n^3 > n^2 + n > N^2$



 $n^2 + n - k$ and $nm \ge km$. So $n^3 + nm > (n^2 + n - k) + km$, which implies |W| > t. This contradicts the assumption that $|W| \le t$. Thus, each node in O must belong to \overline{W} .

Suppose a node in S belongs to \overline{W} . Then, from Claim 2.3.1, each node in S belongs to \overline{W} . Also, each node in V must belong to \overline{W} . As shown above, each node in O is in \overline{W} . Thus, only a subset $Q' \subseteq U'$ can belong to W. Let $Q = U \cap W$. Note that |W| = |Q'| = m|Q|. Since $m = n + n^2$ and $|W| \leq t = n^2 + km + n - k = km + m - k = (k + 1)m - k$, we have $|W| = m|Q| \leq (k + 1)m - k$, so $|Q| \leq k$. Thus $\gamma_W = n^3|Q| \leq n^3k < kn^3 + kn^2 = D$, which contradicts the assumption that $\gamma_W \geq D$. Thus, each node in S belongs to W.

As a consequence of Claim 2.3.2, we have $W = U'_1 \cup V_2 \cup S$ and $\overline{W} = O \cup U'_2 \cup V_1$. Let $U_1 = \{u_i | C_i \in U'_1\}.$

Claim 2.3.3 $|U_1| = k$.

Proof: We first show that $|U_1| \leq k$. Suppose $|U_1| \geq k+1$, then $|W| \geq |U'_1| = |U_1|m \geq (k+1)m = km+m$. Since $m = n+n^2 > (n-k)+n^2$, we have $|W| \geq km+m > km+n-k+n^2 = t$, which contradicts the assumption that $|W| \leq t$. Thus, $|U_1| \leq k$.

Next, we show that $|U_1| \ge k$. Suppose $|U_1| \le k - 1$. Let $|V_1| = h$. Then, $|V_2| = |V| - |V_1| = n - h$. Recall that γ_W is the number of arcs from \overline{W} to W.

The number of arcs from O to U'_1 (resp., from V_1 to S and from U'_2 to V_2) is $n^3|U_1| \le n^3(k-1)$ (resp., hn^2 and $\le n|V_2| = n(n-h)$). Thus $\gamma_W \le n^3(k-1) + hn^2 + n(n-h) = kn^3 - n^3 + n^2 + h(n^2 - n)$. Since $n^2 - n > 0$ and $0 \le h \le n$, $(n^2 - n)h$ reaches its maximum when h = n. Thus $kn^3 - n^3 + n^2 + h(n^2 - n) \le kn^3 - n^3 + n^2 + n(n^2 - n) = kn^3 < kn^3 + kn^2 = D$. Thus, $\gamma_W < D$, contradicting the assumption that $\gamma_W \ge D$. Thus, $|U_1| \ge k$. The result follows. \Box



Claim 2.3.4 $|V_1| \ge k$.

Proof: Note that $|W| = |U'_1| + |V_2| + |S| = km + |V_2| + n^2 \le t = km + (n-k) + n^2$. Thus, $|V_2| \le n-k$. Since $|V_1| = n - |V_2|$, we have $|V_1| \ge k$.

Note that $U'_1 \subseteq W$, $V_1 \subseteq \overline{W}$. Then, from the definition of an elite group, there is no arc from U'_1 to V_1 in G'. Since G' is the bipartite complement graph of G, there is an edge between each node in U_1 and each node in V_1 in G. Since $|U_1| = k$, $|V_1| \ge k$, there exists at least one balanced biclique of size 2k in G. This concludes the proof of Theorem 2.3.3.

Relaxing the Structure of the Elite Group

Due to the combined requirements of cardinality and structure, the size-constrained elite group problem (SCEGP) of Section 2.3.1 is not guaranteed to always have a non-trivial feasible solution. However, in practice, we may sometimes prefer to identify a non-empty group of players who can influence a large number of players outside the group, but are also influenced by a few outsiders. To enable such solutions, we relax the constraint that forbids directed arcs from nodes of an elite group W to $V \setminus W$. Instead, for each arc, we impose a penalty p > 0 that is specified as part of the input. The objective function is the number of incoming arcs into W minus p times the number of arcs coming out of W. We refer to this problem as the Size-Constrained Elite Group Problem with Penalties (SCEGPP).

Technical Definition

<u>INSTANCE</u>: *n* players; an "influence" social network represented by a directed graph G(V, A), |V| = n, in which the nodes represent the players and the set of arcs represent pairwise



influences pertaining to a social property: a directed arc (i, j) indicates that i is influenced by j. A positive integer $k \leq n$. A positive number p.

SOLUTION: A set $W \subseteq V$ such that $|W| \leq k$.

<u>OBJECTIVE FUNCTION</u>: Maximize γ_W , the number of arcs from $V \setminus W$ to W minus p times the number of arcs from W to $V \setminus W$. That is, $\gamma_W = \sum_{i \notin W, j \in W} a_{ij} - p \sum_{i \in W, j \notin W} a_{ij}$, where $a_{ij} = 1$, if $(i, j) \in A$; 0 otherwise.

IP Formulation

For $j \in V$, we define $\pi_j \in \{0, 1\}$ as follows:

 $\pi_j = \begin{cases} 1, & \text{node } j \text{ belongs to the elite group } W; \\ 0, & \text{otherwise.} \end{cases}$

For $(i, j) \in A$, we define $x_{ij} \in \{0, 1\}$ and $y_{ij} \in \{0, 1\}$ as follows:

$$\begin{aligned} x_{ij} &= \begin{cases} 1, & \text{arc } (i,j) \text{ is from } V \setminus W \text{ to } W; \\ 0, & \text{otherwise.} \end{cases} \\ y_{ij} &= \begin{cases} 1, & \text{arc } (i,j) \text{ is from } W \text{ to } V \setminus W; \\ 0, & \text{otherwise.} \end{cases} \end{aligned}$$

An IP formulation for SCEGPP is as follows:

$$\max \gamma_W = \sum_{(i,j) \in A} (x_{ij} - py_{ij})$$

s.t. $\pi_i - \pi_j = y_{ij} - x_{ij}, \quad \forall (i,j) \in A$
 $y_{ij} + x_{ij} \leq 1, \quad \forall (i,j) \in A$
 $\sum_{i \in V} \pi_i \leq k,$



$$x_{ij}, y_{ij} \in \{0, 1\}, \quad \forall (i, j) \in A$$

 $\pi_i \in \{0, 1\}, \quad \forall i \in V$

If node $i \in W$ and node $j \in V \setminus W$ (i.e., $\pi_i = 1, \pi_j = 0$), then the first constraint enforces that $y_{ij} = 1, x_{ij} = 0$. Similarly, if node $i \in V \setminus W$ and node $j \in W$ (i.e., $\pi_i = 0, \pi_j = 1$), then $y_{ij} = 0, x_{ij} = 1$. For $\pi_i = \pi_j = 0$ or $\pi_i = \pi_j = 1$, we have $y_{ij} - x_{ij} = \pi_i - \pi_j = 0$ and, hence, $y_{ij} = 0, x_{ij} = 0$ due to the second constraint. Thus, in the objective function, $(x_{ij} - py_{ij})$ represents the contribution of arc (i, j) to γ_W : (i) if arc (i, j) is from $V \setminus W$ to W, then the contribution is 1, (ii) if arc (i, j) is from W to $V \setminus W$, then the contribution is -p, (iii) if arc (i, j) is from W to W (or from $V \setminus W$ to $V \setminus W$), then the contribution is 0. The following theorem discusses the computational complexity of SCEGPP.

Theorem 2.3.4 The decision problem corresponding to SCEGPP is strongly NP-Complete.

Proof: We again use BBP (Section 2.3.1) in our reduction. The notation is as defined in Section 2.3.1. The construction of the graph G' is exactly the same as in the proof of Theorem 2.3.3. The decision question, however, is different.

DECISION QUESTION: Let $t = km + (n-k) + n^2$, $p = n^4 + 3n^3 + 3n^2$, and $D = kn^3 + kn^2$. Does there exist an elite group W in G' such that $|W| \le t$, and $\gamma_W \ge D$?

The decision problem is clearly in class NP. It is easy to see that the decision question has an affirmative answer if and only if the original graph G contains a balanced biclique of size 2k (i.e, $|U_1| = |V_1| = k$).

 \implies This part is exactly as same as in the proof of Theorem 2.3.3.



 \Leftarrow Suppose W is an elite group in G' with $|W| \leq t$ and $\gamma_W \geq D$.

Claim 2.3.5 In G', there does not exist any arc from W to \overline{W} .

Proof: If there exists at least one arc from W to \overline{W} , then $\gamma_W \leq \sum_{(i,j)\in A} x_{ij} - p$. Since $\sum_{(i,j)\in A} x_{ij} \leq |G'| \leq p$, we have $\gamma_W \leq 0$, which contradicts $\gamma_W \geq D$.

With Claim 2.3.5, the remainder of the argument is the same as in the proof of Theorem 2.3.3.

2.3.2 The Portal Problem (PP)

Given a connected, undirected graph G(V, E) and a positive integer k, recall from Section 2.2 that an optimal portal is a set $Q \subseteq V, |Q| \leq k$ such that r(Q) is maximized.

As mentioned earlier, a portal is a natural extension to a set-based measure of the notion of Betweenness Centrality (BC) for a single node. For k = 1, PP reduces to the well-known Betweenness Centrality Problem, which is polynomially solvable (Everett and Borgatti 1999). Thus, PP is polynomially solvable when k = 1. However, for higher values of k, finding an optimal solution is often a challenging task. The primary difficulty is that the measure r(Q)is *non-additive*. In other words, BCs of two distinct nodes in Q cannot, in general, be simply added when computing r(Q). This is obvious, since a specific path between nodes i and j, $i, j \in V \setminus Q$, with two or more internal nodes in Q is counted only once in the computation of r(Q).

We first show that PP and ESPP are strongly NP-hard. An efficient, polynomial-time algorithm for obtaining an optimal solution on general graphs is, therefore, unlikely. Then,



we address special graphs (bicliques and balanced and full d-trees). Finally, we analyze a heuristic for general trees.

Proof of Hardness of PP and ESPP

Puzis et al. (2007) use the Vertex Cover Problem (Garey and Johnson 1979) to show the hardness of the non-normalized measure BC(Q) (Section 2.2.2), which is the numerator of our measure r(Q). Furthermore, given G and k, the variant considered in Puzis et al. (2007) requires that the solution has exactly k nodes. The strongly NP-Complete problem which we use in our reduction is the Independent Set Problem (Garey and Johnson 1979).

Independent Set Problem (ISP)

Instance. A connected, undirected graph G = (V, E); a positive integer $k \leq |V|$.

Solution. A set of nodes, $I \subseteq V, |I| \ge k$, such that no two nodes in I are connected by an edge in E.

Theorem 2.3.5 The decision problem corresponding to PP is strongly NP-Complete.

Proof: Given an arbitrary instance of ISP, specified by G(V, E), we consider the following decision problem:

<u>DECISION QUESTION</u>: Does there exist a portal $Q \subseteq V$ in G(V, E) such that $|Q| \leq |V| - k$ and $r(Q) \geq 1$?

Note that the decision problem is clearly in class NP. We now show that ISP has an affirmative answer if and only if the above decision question has an affirmative answer.



Suppose I^* is an independent set in G with at least k^* nodes. Let $Q^* = V \setminus I^*$. Then, $|Q^*| \leq |V| - k^*$. From the definition of an independent set, it follows that all paths in G between any two nodes in I^* have at least one node in Q^* as an internal node. Thus, $r(Q^*) = 1$ and the decision question has an affirmative answer. Conversely, if there exists $Q \subseteq V$ with $|Q| \leq |V| - k$ and $r(Q) \geq 1$, then the set $V \setminus Q$ is an independent set of at least k nodes.

Corollary 2.3.6 The decision problem corresponding to ESPP is strongly NP-Complete.

Results on Specific Families of Graphs

We now discuss two specific families of graphs: Bicliques and Balanced and Full d-Trees.

Bicliques:

Let $G = (U \cup V, E)$ be a biclique: $n_1 = |U| \leq |V| = n_2$ and $u \in U, v \in V$ implies that $\{u, v\} \in E$. The size of the biclique is $n_1 + n_2$. Let $Q_1, Q_2 \subseteq U \cup V$. If $|Q_1 \cap U| = |Q_2 \cap U|$ and $|Q_1 \cap V| = |Q_2 \cap V|$, then $BC(Q_1) = BC(Q_2)$. Thus, for $Q \subseteq U \cup V$, the objective function r(Q) depends only on two numbers: $k_1 = |Q \cap U|$ and $k_2 = |Q \cap V|$. Theorem 2.3.7 (resp., Corollary 2.3.11) provides an optimal solution to PP (resp., ESPP).



Figure 2.8. Optimal Portal in a Biclique.



Theorem 2.3.7 Let $G = (U \cup V, E)$ be a biclique with $n_1 = |U| \le |V| = n_2$. Let $Q \subseteq U \cup V$. Let $k_1 = |Q \cap U|, k_2 = |Q \cap V|$. Then,

- (i) For 1 ≤ k ≤ n₁ − 1, any set Q which satisfies k₁ = k and k₂ = 0 is an optimal solution of PP.
- (ii) For $k \ge n_1$, then Q = U is an optimal solution of PP.

Proof: The result of Theorem 2.3.7 follows from the proofs of Lemmas 2.3.8, 2.3.9, and 2.3.10.

Lemma 2.3.8 Let $G = (U \cup V, E)$ be a biclique with $n_1 = |U| \le |V| = n_2$. If $k \ge n_1$, then Q = U is an optimal solution of PP.

Proof: Let Q = U. Since $k \ge n_1$, we have $|Q| = |U| = n_1 \le k$. Let G'(Q) denote the induced subgraph obtained by removing all the nodes in Q from G. Thus, we have G'(Q) = V. For any two nodes in V, there exist n_1 shortest paths (in G) between them. Each shortest path has exactly one node in U as an internal node. Thus,

$$BC(Q) = \begin{pmatrix} |V| \\ 2 \end{pmatrix} \frac{n_1}{n_1} = \begin{pmatrix} n_2 \\ 2 \end{pmatrix}.$$

Also,

$$\begin{pmatrix} n_1 + n_2 - |Q| \\ 2 \end{pmatrix} = \begin{pmatrix} n_2 \\ 2 \end{pmatrix}.$$

Therefore, r(Q) = 1 and Q = U is an optimal solution of PP.

To obtain an optimal solution of PP for $1 \le k \le n_1 - 1$, we first obtain an optimal solution of the corresponding instance of ESPP in the following lemma.



Lemma 2.3.9 Let $G = (U \cup V, E)$ be a biclique with $n_1 = |U| \le |V| = n_2$. For $1 \le k \le 1$ $n_1 - 1$, any set Q with $k_1 = |Q \cap U| = k$ and $k_2 = |Q \cap V| = 0$ is an optimal solution of ESPP.

Proof: Since n_1, n_2 and k are given, the value $\begin{pmatrix} n_1 + n_2 - k \\ 2 \end{pmatrix}$ is fixed. Thus, maximizing

r(Q) is equivalent to maximizing BC(Q). We have

$$BC(Q) = \begin{pmatrix} n_1 - k_1 \\ 2 \end{pmatrix} \frac{k_2}{n_2} + \begin{pmatrix} n_2 - k_2 \\ 2 \end{pmatrix} \frac{k_1}{n_1}$$

Using $k_2 = k - k_1$, we obtain

$$BC(Q) = \binom{n_1 - k_1}{2} \frac{k - k_1}{n_2} + \binom{n_2 - (k - k_1)}{2} \frac{k_1}{n_1}$$
$$= \frac{n_1(n_1 - k_1)(n_1 - k_1 - 1)(k - k_1) + n_2(n_2 - k + k_1)(n_2 - k + k_1 - 1)k_1}{2n_1n_2}$$

Let $g(k_1)$ represent the numerator of BC(Q). Thus, $g(k_1) = n_1(n_1 - k_1)(n_1 - k_1 - 1)(k - k_1)(n_1 - k_1)(n_1 - k_1)(n_1 - k_1)(k - k_1)(n_1 - k$ k_1) + $n_2(n_2 - k + k_1)(n_2 - k + k_1 - 1)k_1$. In the following claim, we show that for $0 \le k_1 \le k$, $g(k_1)$ reaches its maximum at $k_1 = k$.

Claim 2.3.6 Let $n_1 \le n_2$ and $1 \le k \le n_1 - 1$. For $0 \le k_1 \le k$, $g(k_1) = n_1(n_1 - k_1)(n_1 - k_2)(n_1 - k_2)(n_1 - k_2)(n_2 - k_2)(n_1 - k_2)(n_2 - k_2)(n$ $k_1 - 1(k - k_1) + n_2(n_2 - k + k_1)(n_2 - k + k_1 - 1)k_1$ reaches its maximum at $k_1 = k$.

Proof: By defining $a = n_2 - n_1$, $b = n_1(2n_1 + k - 1) + n_2(2n_2 - 2k - 1)$, $c = n_1(-n_1^2 - 2k - 1)$ $2kn_1 + n_1 + k$) + $n_2(n_2^2 - 2kn_2 - n_2 + k^2 + k)$, and $e = kn_1^2(n_1 - 1)$, we can rewrite $g(k_1)$ as $ak_1^3 + bk_1^2 + ck_1 + e$. Although k_1 can only take integral values, it is convenient to assume $g(k_1)$ as a continuous function of k_1 ($0 \le k_1 \le k$) for the purpose of this proof. We have two forms of $g(k_1)$ based on the value of a.



- (i) For n₁ < n₂, we have a > 0. Thus, g(k₁) is a cubic function of k₁. Let g'(k₁) represent the first-order derivative of g(k₁) with respect to k₁. Thus, g'(k₁) = dg(k₁)/dk₁ = 3ak₁² + 2bk₁ + c has at most two real roots, which correspond to the critical points of g(k₁). Since a > 0, g'(k₁) is convex. Let Δ = (2b)² 4(3a)c represent the discriminant of g'(k₁).
 - (1) For $\Delta \leq 0$, we have $g'(k_1) \geq 0$. Thus, $g(k_1)$ monotonically increases with k_1 and reaches its maximum at $k_1 = k$ for $0 \leq k_1 \leq k$.
 - (2) For $\Delta > 0$, $g'(k_1)$ has two real roots: \bar{k}_1 and \hat{k}_1 . Without loss of generality, we assume $\bar{k}_1 < \hat{k}_1$. Thus, $\bar{k}_1 = \frac{-2b-\sqrt{\Delta}}{6a}$, $\hat{k}_1 = \frac{-2b+\sqrt{\Delta}}{6a}$. Let $g''(k_1) = \frac{d^2g(k_1)}{dk_1^2} = 6ak_1 + 2b$. Then, we have $g''(\bar{k}_1) = -\sqrt{\Delta} < 0$, $g''(\hat{k}_1) = \sqrt{\Delta} > 0$. Thus, $g(k_1)$ achieves its local maximum at \bar{k}_1 and achieves its local minimum at \hat{k}_1 . Therefore, the maximum of $g(k_1)$ for $0 \le k_1 \le k$ has three possible values: g(0), g(k), or $g(\bar{k}_1)$ (if $0 \le \bar{k}_1 \le k$).

Since $k \le n_1 - 1 < n_2 - 1$, we have $2n_2 - 2k - 1 > 0$. Since $n_1 > 0, n_2 > 0$, and $2n_1 + k - 1 > 0$, we have $b = n_1(2n_1 + k - 1) + n_2(2n_2 - 2k - 1) > 0$. Since a > 0 and b > 0, we have $\bar{k}_1 = \frac{-2b - \sqrt{\Delta}}{6a} < 0$. Thus, since \bar{k}_1 is outside the range [0, k], the maximum of $g(k_1)$ for $0 \le k_1 \le k$ is either g(0) or g(k). Also, we have $g(0) = n_1^2(n_1 - 1)k$ and $g(k) = n_2^2(n_2 - 1)k$. Since $n_1 < n_2$, we have g(0) < g(k). Therefore, for $0 \le k_1 \le k$, $g(k_1)$ reaches its maximum at $k_1 = k$.

(ii) If $n_1 = n_2$, we have a = 0, $b = n_1(4n_1 - k - 2)$, $c = n_1(-4kn_1 + k^2 + 2k)$, and $e = n_1^2(n_1 - 1)k$. Since $k \le n_1 - 1$ and $n_1 \ge 1$, we have $4n_1 - k - 2 > 0$. Thus,



 $b = n_1(4n_1 - k - 2) > 0$. Therefore, $g(k_1) = bk_1^2 + ck_1 + e$ is a quadratic function of k_1 . Since b > 0, $g(k_1)$ is convex. Thus, the maximum of $g(k_1)$ for $0 \le k_1 \le k$ is either g(0) or g(k). Also, we have $g(0) = n_1^2(n_1 - 1)k = g(k)$. Therefore, for $0 \le k_1 \le k$, $g(k_1)$ reaches its maximum at $k_1 = k$.

This completes the proof of Claim 2.3.6.

Thus, given an instance of ESPP with n_1, n_2 , and k, the function $g(k_1)$ reaches its maximum at $k_1 = k$. Consequently, r(Q) is maximized for any set Q with $k_1 = k, k_2 = 0$. This completes the proof of Lemma 2.3.9.

Lemma 2.3.9 provides an optimal solution of ESPP for $1 \le k \le n_1 - 1$. In our next result, we show that the optimal value r(Q) of ESPP increases with |Q| for $1 \le |Q| \le n_1 - 1$. Thus, given $k, 1 \le k \le n_1 - 1$, an optimal solution of PP can be obtained by solving the corresponding instance of ESPP.

Lemma 2.3.10 Let $G = (U \cup V, E)$ be a biclique with $n_1 = |U| \le |V| = n_2$. Let Q^* denote an optimal solution of an instance of PP defined by G and a positive integer $k, 1 \le k \le n_1 - 1$. Then, $|Q^*| = k$. Consequently, an optimal solution of PP can be obtained by solving the instance of ESPP corresponding to G and k.

Proof: Let \hat{Q} be an optimal solution of an instance of ESPP defined by G and a positive integer \hat{k} . Thus, we have $|\hat{Q}| = \hat{k}$. We next show $r(\hat{Q})$ increases with \hat{k} . From Lemma 2.3.9, we have $\hat{k}_1 = |\hat{Q} \cap U| = \hat{k}$ and $\hat{k}_2 = |\hat{Q} \cap V| = 0$. Thus,

$$BC(\hat{Q}) = \binom{n_1 - \hat{k}}{2} \frac{0}{n_2} + \binom{n_2}{2} \frac{\hat{k}}{n_1} = \frac{n_2(n_2 - 1)\hat{k}}{2n_1},$$

المتسارات

$$r(\hat{Q}) = \frac{BC(\hat{Q})}{\left(\begin{array}{c} n_1 + n_2 - \hat{k} \\ 2 \end{array}\right)}.$$

For $1 \leq \hat{k} \leq n_1 - 1$, the value $BC(\hat{Q})$ increases with \hat{k} , and $\begin{pmatrix} n_1 + n_2 - \hat{k} \\ 2 \end{pmatrix}$ decreases

with \hat{k} . Thus, $r(\hat{Q})$ increases with \hat{k} . Since Q^* is an optimal solution of an instance of PP defined by G and k, we have $r(Q^*) = \max_{1 \le \hat{k} \le k} r(\hat{Q})$. Since $r(\hat{Q})$ increases with \hat{k} , we have $|Q^*| = k$, $|Q^* \cap U| = k$, and $|Q^* \cap V| = 0$. The result follows.

Together, Lemmas 2.3.8, 2.3.9, and 2.3.10, complete the proof of Theorem 2.3.7. ■ We also summarize the solution of ESPP.

Corollary 2.3.11 Let $G = (U \cup V, E)$ be a biclique with $n_1 = |U| \le |V| = n_2$. Let $Q \subseteq U \cup V$. Let $k_1 = |Q \cap U|, k_2 = |Q \cap V|$.

- 1. For $1 \le k \le n_1 1$, any set Q which satisfies $k_1 = k$ and $k_2 = 0$ is also an optimal solution of ESPP.
- 2. For $n_1 \leq k \leq n_1 + n_2 1$, then any set Q which satisfies $k_1 = n_1$ and $k_2 = k n_1$ is an optimal solution of ESPP.

Balanced and Full *d*-Trees:

Given a tree G(V, E) and $Q \subseteq V$, let G'(Q) denote the induced subgraph obtained by removing all the nodes in Q from G. In general, G'(Q) is a forest with disjoint trees as its connected components. Since G is a tree, there is a unique path in G connecting any two distinct nodes s and t in $V \setminus Q$; thus, $\sigma_{st} = 1$ (see Section 2.2.2). We first define some



notation for a general tree G(V, E):

- n: the number of nodes in G (i.e., n = |V|).
- k: the number of nodes in Q (i.e., k = |Q|).
- *l*: the number of connected components in G'(Q).
- A_i : the *i*th connected component in G'(Q), $i = 1, 2, \cdots, l$.
- a_i : the size (i.e., the number of nodes) of component A_i , $i = 1, 2, \dots, l$.

Consider a connected component, say A_i , of G'(Q). In G, there is a unique path from any node in A_i to each node in every other connected component in G'(Q). Thus,

$$BC(Q) = \sum_{s \notin Q, t \notin Q, s \neq t} \frac{\sigma_{st}(Q)}{\sigma_{st}} = \sum_{1 \le i < j \le l} a_i a_j$$
(2.1)

Since $\sum_{i=1}^{l} a_i = |V| - |Q| = n - k$, we have

$$BC(Q) = \frac{(n-k)^2 - \sum_{i=1}^l a_i^2}{2}$$
(2.2)

Thus, for fixed n and k, maximizing BC(Q) is equivalent to minimizing $\sum_{i=1}^{l} a_i^2$. We next illustrate the solution of this problem for balanced and full d-trees.

On a rooted balanced and full *d*-tree, each node (except the leaf nodes) has *d* distinct successors, each node (except root) has a unique predecessor. All leaf nodes have the same distance (height) to the root node. For a *d*-tree, if we remove any node other than the root node and leaf nodes, we will add *d* more connected components into the remaining graph. So if we remove *k* nodes from a *d*-tree, we will have at most l = dk + 1 connected components left.

Theorem 2.3.12 Let G be a balanced and full d-tree with height $h \ge 2$. For an instance of PP defined by G and a positive integer k, let $t = \min\{\lceil h/2 \rceil, \lfloor \log_d k \rfloor\}$ and let \overline{Q} denote the



36



Figure 2.9. Optimal Portal in a Balanced and Full Binary (d=2) Tree.

set of nodes on the tth level of G. Then, \overline{Q} provides an asymptotically maximal solution to PP, with $r(\overline{Q}) \ge (1 - \frac{1}{d^{t+1}})$.

Proof: First, since the t^{th} level of a balanced and full *d*-tree has d^t nodes, $|\bar{Q}| = d^t \leq k$. Note that $G'(\bar{Q})$ has exactly $l = d|\bar{Q}| + 1$ connected components. Of these, we have (i) $d|\bar{Q}|$ identical components, say $A_i, i = 1, 2, ..., d|\bar{Q}|$, each with $\frac{d^{h-t}-1}{d-1}$ nodes. Thus, $a_1 = a_2 = \cdots = a_{d|\bar{Q}|} = \frac{d^{h-t}-1}{d-1}$, and (ii) one component, say A_l , with $\frac{d^t-1}{d-1}$ nodes. Thus, $a_l = \frac{d^t-1}{d-1}$.

From Equation 2.1, we have

$$BC(\bar{Q}) = \sum_{1 \le i < j \le l} a_i a_j$$

$$= d|\bar{Q}|a_l a_1 + a_1^2 \begin{pmatrix} d|\bar{Q}| \\ 2 \end{pmatrix}$$

$$= d^{t+1} \frac{d^t - 1}{d - 1} \frac{d^{h-t} - 1}{d - 1} + \frac{(d^{h-t} - 1)^2}{(d - 1)^2} \frac{d^{t+1}(d^{t+1} - 1)}{2}$$

$$= \frac{d^{t+1}(d^{h-t} - 1)(d^{h+1} - d^{h-t} - d^{t+1} + 2d^t - 1)}{2(d - 1)^2}.$$
Also, $\binom{n - |\bar{Q}|}{2} = \binom{\frac{d^{h+1} - 1}{d - 1} - d^t}{2}}{2} = \frac{(d^{h+1} - d^{t+1} + d^t - 1)(d^{h+1} - d^{t+1} + d^t - d^{t+1} + d^t - d^t)}{2(d - 1)^2}.$



Thus,

$$BC(\bar{Q}) - (1 - \frac{1}{d^{t+1}}) \begin{pmatrix} n - |\bar{Q}| \\ 2 \end{pmatrix}$$

= $\frac{1}{2d^{t+1}(d-1)^2} [d^h(d^{t+3} - d^{t+2} + 2d^{t+1} - d^2 - d) + d - d^t + d^{2t} - d^{2t+1} + 2d^{2t+2} - d^{2t+3} - d^{3t+1}]$

Since $t = \min\{\lceil h/2 \rceil, \lfloor \log_d k \rfloor\}$ and $\lceil h/2 \rceil \le (h+1)/2$, we have $t \le \lceil h/2 \rceil \le (h+1)/2$. Thus, $d^h \ge d^{2t-1}$. Since $(d^{t+3} - d^{t+2} + 2d^{t+1} - d^2 - d) \ge 0$, we have

$$BC(\bar{Q}) - (1 - \frac{1}{d^{t+1}}) \begin{pmatrix} n - |\bar{Q}| \\ 2 \end{pmatrix} \geq \frac{1}{2d^{t+1}(d-1)^2} [d^{2t-1}(d^{t+3} - d^{t+2} + 2d^{t+1} - d^2 - d) \\ + d - d^t + d^{2t} - d^{2t+1} + 2d^{2t+2} - d^{2t+3} - d^{3t+1}] \\ = \frac{(d^t - d)(d^{2t}(d(d-2) + 2) - 1)}{2d^{t+1}(d-1)^2} \\ \geq 0$$

Thus,

$$r(\bar{Q}) = \frac{BC(\bar{Q})}{\left(\begin{array}{c} n - |\bar{Q}| \\ 2 \end{array}\right)} \ge (1 - \frac{1}{d^{t+1}}).$$

Since $t = \min\{\lceil h/2 \rceil, \lfloor \log_d k \rfloor\}$, the ratio $r(\bar{Q}) \to 1$ with an increase in the size of G and k. Thus, the resulting family of solutions \bar{Q} can be referred to as an asymptotically maximal family. Note that the solution \bar{Q} is not necessarily optimal. Thus, in cases where \bar{Q} is not optimal, there can exist a solution that is superior to \bar{Q} .



Remark 2: The solutions of PP and ESPP on paths, cycles, and cliques, are straightforward to obtain. Therefore, these results are listed without proofs as follows.

Theorem 2.3.13 Let G be a path $v_1 cdot v_2 cdot \dots cdot v_n$ and consider an instance of ESPP specified by G and k. Let $\mu = \frac{n-k}{k+1}$, $c = n-k-(k+1)\lfloor\mu\rfloor$. Then, Q^* is an optimal solution of ESPP if and only if $G'(Q^*)$ has exactly c connected components of size $\lfloor\mu\rfloor + 1$ and exactly (k+1-c)connected components of size $\lfloor\mu\rfloor$.

Thus, given an explicit description of the path $v_1 - v_2 - \ldots - v_n$ and a positive integer k, an optimal solution of ESPP is $Q^* = \{v_{i(\lfloor \mu \rfloor + 2)}, i = 1, 2, \cdots, c\} \cup \{v_{c(\lfloor \mu \rfloor + 2) + j(\lfloor \mu \rfloor + 1)}, j = 1, 2, \cdots, k - c\}.$ The optimal objective function value is $r(Q^*) = BC(Q^*) / \binom{n-k}{2}$, where

$$BC(Q^*) = \frac{(n-k)^2 - c(\lfloor \mu \rfloor + 1)^2 - (k+1-c)(\lfloor \mu \rfloor)^2}{2}.$$

An optimal solution of PP is also easy to obtain: we simply solve ESPP for each $\bar{k} \leq k$. Since $k \leq n$, this requires time polynomial in the size of the input (assuming that the path is explicitly specified in the input).

Theorem 2.3.14 Let G be a cycle $v_1 cdot v_2 cdot \dots cdot v_n cdot v_1$ and consider an instance of ESPP specified by G and k. Let $\mu = \frac{n-k}{k}$, $c = n - k - k\lfloor \mu \rfloor$. Then, Q^* is an optimal solution of ESPP if and only if $G'(Q^*)$ has exactly c connected components of size $\lfloor \mu \rfloor + 1$ and exactly (k - c)connected components of size $\lfloor \mu \rfloor$.

Thus, given an explicit description of the cycle $v_1 - v_2 - \dots - v_n - v_1$ and a positive integer $k \ge 2$, an optimal solution of ESPP is $Q^* = \{v_{i(\lfloor \mu \rfloor + 2)}, i = 1, 2, \dots, c\} \cup \{v_{c(\lfloor \mu \rfloor + 2) + j(\lfloor \mu \rfloor + 1)}, j = 1, 2, \dots, c\}$



 $1, 2, \dots, k-c$. The optimal objective function value is $r(Q^*) = BC(Q^*) / \binom{n-k}{2}$, where

$$BC(Q^*) = \frac{(n-k)^2 - c(\lfloor \mu \rfloor + 1)^2 - (k-c)(\lfloor \mu \rfloor)^2}{2}.$$

We can easily obtain an optimal solution of PP by solving ESPP for each $\bar{k} \leq k$. Since $k \leq n$, this requires time polynomial in the size of the input (assuming that the cycle is explicitly specified in the input).

Let G = (V, E) be a clique and let $Q \subseteq V$. For any two nodes $s, t \in V \setminus Q$, the unique shortest path in G between s and t is of length 1 and exists in G'(Q). Thus, no shortest path between any two nodes in $V \setminus Q$ has an internal node in Q. Thus, $\sigma_{st}(Q) = 0$. It follows that $r(Q) = BC(Q) \equiv 0$ for any $Q \subseteq V$. In other words, any subset of nodes is an optimal solution for PP and ESPP.

General Trees

An important question that arises naturally is that of the solution of PP on a general tree. The tree structure occurs frequently in real-world social networks. Wein (2009) describes the milk supply chain as a tree. The author argues that for a potential terrorist attack, it is enough to introduce a small amount of toxin (botulinum) at a few key nodes of the tree. We believe that such nodes can be identified by finding an optimal or near-optimal portal in the tree. In Perer and Wilson (2007), the authors discuss the underground distribution network of steroids among players of Major League Baseball. Investigators have used this network to determine the role of each individual in the distribution of steroids. Again, knowledge



of a good portal in this network should help identify key members. Interfirm collaboration networks, studied in Schilling and Phelps (2007), are approximately trees. In their search for firms with a higher innovative output, the authors find a set of nodes that is a near-optimal portal. In Hanaki et al. (2007), the authors argue that locally, a large and sparse random network often resembles a pure branching tree.

The computational complexity of PP for a tree is an open problem. We now present a simple heuristic (Theorem 2.3.15) to find a portal in a tree and obtain a lower bound on its performance. To describe the heuristic, we need the following labeling procedure:

Labeling Procedure

Input: A tree G(V, E).

Initialization: Let i = 0.

Step 1: Select all the leaf nodes of G, label them as being on Level *i*, and include them in set S(i). Let $G = G \setminus S(i)$.

Step 2: If $G = \emptyset$, terminate; otherwise, let i = i + 1 and go to Step 1.

We record the highest level we get in this labeling procedure as h, and refer to it as the *height* of the tree. Let n[j] = |S(j)|, the number of nodes in Level j. It is easy to see that $n[0] \ge n[1] \ge n[2] \ge \ldots \ge n[h]$ and $n[h] \in \{1, 2\}$.

Theorem 2.3.15 Let G be a tree with height $h \ge 2$. For an instance of PP defined by G(V, E) and a positive integer $k \ge 2$,

1. If $n[1] \le k$, then set $\bar{t} = 1$.



2. If n[1] > k, then set \overline{t} such that $n[\overline{t} - 1] > k, n[\overline{t}] \le k$.

Let $t = \max\{\lceil h/2 \rceil, \bar{t}\}$. Let \bar{Q} denote the set of nodes on the level t of G. Then, \bar{Q} provides a solution to PP with $r(\bar{Q}) \geq \frac{b(b-1)t^2+2bt(h-t)}{(n-b)(n-b-1)}$, where b = n[t] and n = |V|. Moreover, this bound is tight.

Proof: Since \bar{Q} includes all the nodes on level t of G, we have $|\bar{Q}| = n(t)$. Since $t = \max\{\lceil h/2 \rceil, \bar{t}\} \ge \bar{t}$, we have $n(t) \le n[\bar{t}] \le k$. Thus, $|\bar{Q}| \le k$. Note that $G'(\bar{Q})$ has $l \ge b+1$ connected components. Of these, we have (i) at least b components, say $A_i, i = 1, 2, \ldots, b$, each with at least t nodes: from level 0 to level t-1, A_i has at least one node from each level. Thus, $a_i = |A_i| \ge t, i = 1, 2, \ldots, b$, and (ii) one component, say A_l , with at least h-t nodes: from level t + 1 to level h, A_l has at least one node from each level. Thus, $a_l = |A_l| \ge h - t$. From (2.1) defined above, we have

$$BC(\bar{Q}) = \sum_{1 \le i < j \le l} a_i a_j$$

$$\ge \sum_{1 \le i < j \le b} a_i a_j + \sum_{1 \le i \le b} a_i a_l$$

$$\ge \binom{b}{2} t^2 + bt(h-t)$$

$$= \frac{b(b-1)}{2} t^2 + bt(h-t).$$

Also, $\binom{n-|\bar{Q}|}{2} = \frac{(n-b)(n-b-1)}{2}$. Thus, we have

$$r(\bar{Q}) = \frac{BC(\bar{Q})}{\binom{n-|\bar{Q}|}{2}} \ge \frac{b(b-1)t^2 + 2bt(h-t)}{(n-b)(n-b-1)}$$

To show the tightness of the bound, let G be a path of length 4. Thus, n = 5, h = 2, n[0] = n[1] = 2, n[2] = 1. For k = 2, if we apply the heuristic, we have \bar{Q} as all the nodes



on Level 1 of G. Then, $t = 1, b = n[1] = 2, \frac{b(b-1)t^2 + 2bt(h-t)}{(n-b)(n-b-1)} = 1$, which implies that \overline{Q} is an optimal solution to PP on G.

When G is a balanced and full d-tree, the procedure in Theorem 2.3.15 is the same as the one in Theorem 2.3.12, which was shown to provide an asymptotically maximal solution to PP.

2.4 Conclusions and Future Research Directions

The ability to find useful structures in social networks will undoubtedly benefit their users as well as other stakeholders – the businesses that use these networks and the sites that host them. Unlike the internet, structural search on social networks is set-based and offers a rich variety of interesting combinatorial optimization problems. In this paper, our effort is to identify and analyze specific instances of such problems. We consider two problems – the Elite Group Problem (EGP) and the Portal Problem (PP) – derived, respectively, from the notions of influence and centrality. We demonstrate the relevance of these problems on a variety of social networks and show the following results: (i) the basic EGP is polynomially solvable while its size-constrained variant is strongly NP-hard. We also show the hardness of a penalty-based relaxation of the size-constrained version. (ii) PP is strongly NP-hard. We discuss the solution of PP on several special networks – bicliques, balanced and full *d*-trees, paths, cycles, and cliques – and propose a heuristic for general trees.

In the industry, the focus, thus far, has been on developing "social search engines" to search social media and user-generated content, e.g., Social Mention (http://www.social



mention.com), Twitter (http://search.twitter.com), and Delver (http://www.delver.com). Some networks do facilitate simple search, e.g., MySpace allows a user to find other users with similar interests. However, to our knowledge, there is little or no sophisticated structural search available to ordinary users of social networks. Since this type of search is typically combinatorial in nature, the resulting problems are expected to be challenging. One idea is to provide an easy-to-use modeling language to enable members to specify complex, constrained search and then use sophisticated solvers (e.g., CPLEX) or heuristics to solve the resulting problems. Another possibility is to develop a repository – that could evolve over time – of efficient algorithms for the typical combinatorial searches that users specify. The notions of an elite group and a portal studied in this paper are extensions to set-based measures of, respectively, indegree and betweenness centralities for individual members of a social network. Similarly, useful structures based on extensions of other popular centralities, e.g., the more general degree centrality or closeness centrality (Carrington et al. 2005), could also be investigated. Applications of such set-based measures have been discussed for several social networks (see, e.g., Cattani and Ferriani 2008, Owen-Smith et al. 2002, Morselli and Giguere 2006).

The ideas of search developed in this paper naturally flow into other operational problems of interest. One such problem is targeted online advertising. For example, in the Twitter network, it is possible for one member to "follow" another, suggesting a directed link in the network. The identification of an elite group within Twitter could, therefore, be used to target promotional material to members of this group. For instance, an advertisement could be targeted using keywords exchanged by two members during a conversation on Twitter.



CHAPTER 3

VALUE OF LOCAL CASH REUSE: INVENTORY MODELS FOR MEDIUM-SIZE DEPOSITORY INSTITUTIONS UNDER THE NEW FEDERAL RESERVE POLICY ¹

3.1 Synopsis

Unlike most consumer products, a significant portion of physical cash is recirculated and reused in an economy. The Federal Reserve System (Fed) of the United States provides currency services (e.g., separating cash into notes that are fit and unfit for recirculation) to depository institutions (DIs; e.g., banks) who, in turn, meet the public's demand for cash. Over the years, the Fed's analysis has suggested that DIs are relying on the Fed's cash processing services to recirculate a substantial amount of physical cash that the DIs should have recirculated themselves (Federal Reserve 2003, 2006). As identified by the Fed, it is often the case that cash deposited to the Fed by a DI is reordered by the same DI in the same denomination within one business week. From the point of view of a DI, this practice – referred to as *cross-shipping* – made sense (prior to 2007) since the Fed's cash processing services were free. Therefore, it was convenient and economical for a DI to let the Fed clean the cash, i.e., separate it into fit and unfit notes. This practice, however, progressively and significantly increased the Fed's expenses to process cash and manage its currency circulation

¹**Reprinted by permission**, Yunxia Zhu, Milind Dawande, Chelliah Sriskandarajah, Value of Local Cash Reuse: Inventory Models for Medium-Size Depository Institutions under the New Federal Reserve Policy, Manufacturing & Service Operations Management, Volume 13, Number 4, 2011, Copyright (2011), the Institute for Operations Research and the Management Sciences, 7240 Parkway Drive, Suite 300, Hanover, Maryland 21076 USA.



operations (Blacketer and Evetts 2004). With the aim of encouraging DIs to "locally" clean and reuse cash to ultimately reduce the societal cost of providing currency to the public, the Fed introduced its new currency recirculation policy in July 2007. The details of the Fed's policy will be discussed in Section 3.1.1.



Figure 3.1. Life Cycle of Physical Cash.

In its physical form, cash can be classified into four categories (Figure 3.1). The Bureau of Engraving and Printing (BEP) prints *new cash* on the Fed's request. After its introduction into the economy, this cash is referred to as *used cash*. Over time, the quality of used cash deteriorates. Used cash that is deemed suitable to remain in circulation is referred to as *fit cash*. The remainder is *unfit cash*, which is typically worn, soiled, or torn, and is destroyed by the Fed. In this paper, we only address the management of physical notes. Coins are not considered since they are less perishable, have lesser value, and the guidelines and challenges in managing their use are markedly different from those of notes. We now describe the structure of the U.S. cash supply chain.

3.1.1 The Closed-Loop Cash Supply Chain

As in Rajamani et al. (2006), it is convenient to describe the cash supply chain as a closedloop supply chain using its 4-P components: Product, Players, Process, and Policy.



• Product:

The Fed's new currency recirculation policy applies only to the \$10 and \$20 notes (Federal Reserve 2006). Between these two denominations, \$20 notes form an overwhelmingly dominant portion of the currency in circulation because of their heavy use in ATMs. Consequently, DIs are primarily interested in recirculating \$20 notes (Federal Reserve 2006). Our analysis, therefore, assumes currency of a single denomination, namely \$20. Both the transportation and transactions of cash between DIs and the Fed occur in integer multiples of cash *bundles*, where a bundle is 1000 notes of the same denomination.

• Players & Process:

The Fed acts as the exclusive wholesaler, that orders the production of cash from the BEP and controls the release of cash into the economy through DIs. The DIs act as retailers, providing cash to end customers and collecting deposits from them. Individuals and Commerce are the end customers, who deposit used cash to and/or withdraw fit cash from the DIs through various channels such as ATMs, bank branches, and businesses. Besides these three primary players, third-party logistic providers (3PLPs) such as Brink's, Inc., Dunbar Armored, and Loomis, Fargo & Co., also play an important role in this supply chain. In addition to collecting cash and transporting it between the DIs and the Fed, the 3PLPs are also involved in the processing of cash, e.g., counting and sorting cash into fit and unfit notes. Before July 2007, the Fed accepted deposits of used cash from DIs, fit-sorted that currency (i.e., separated it into fit and unfit cash), removed unfit cash from circulation, and provided fit cash to DIs.



The DIs simply sorted the collected used cash by denomination and then forwarded it to the Fed. After the implementation of the Fed's policy, in an ideal situation, the DIs locally fit-sort the used cash and use the resulting fit cash to satisfy (partially or fully) their customers' demand. From the systemic view, the local reuse of fit cash by DIs can be an influential practice because a significant fraction – about 65-75%, based on the Fed's documentation (Federal Reserve 2010b) and our conversations with a third-party provider – of used cash is fit cash. Thus, a major percentage of the movement of used cash (resp., fit cash) to (resp., from) the Fed can be avoided and, consequently, help in supporting the Fed's eventual goal of minimizing the cost of providing currency to the public. Our analysis in this paper assumes that the percentage of fit cash in the used cash is 75%.

The inventory models in this paper specifically capture the operations of DIs that handle a medium volume of currency. Such DIs face a weekly demand of fit cash that is typically between 1000 to 2000 bundles. The weekly demand at DIs that handle large volumes typically averages more than 5000 bundles (Dawande et al. 2010). Due to the relatively low volume of physical cash, medium DIs may prefer to engage the services of 3PLP for fit-sorting and managing custodial inventory (to be defined shortly). This has been anticipated by the Fed as well (Federal Reserve 2003). Furthermore, due to the relatively high percentage variation (as compared to large DIs) in the weekly transaction of cash, such DIs are expected to engage the services of the 3PLP on an as-required basis (as opposed to a daily or weekly service, which is the typical case with large DIs). In turn, since service requests from such a DI can be irregular, a



3PLP typically uses a fixed-fee structure: a fixed-cost for ordering transportation, a setup cost for fit-sorting in addition to a per-bundle fit-sorting cost, and a setup cost for withdrawing cash from custodial inventory. Since the deposit (if any) into custodial inventory occurs immediately after fit-sorting (for which a fee is charged), 3PLPs typically do not include a separate fee for such deposits.

• Policy:

As mentioned earlier, DIs had been overusing (prior to 2007) the cash processing services of the Fed via the practice of cross-shipping. There were two main reasons for DIs to cross-ship. First, the cost of fit-sorting – regardless of whether the DI invests in the infrastructure itself or uses a 3PLP – discourages DIs from locally reusing cash: it requires sophisticated labor and expensive machines. On the other hand, the Fed provided this service for free. Second, for a DI, currency held on account at the Fed can earn interest (by lending to other institutions via the Federal Funds Market), while currency held in its own vault cannot. Therefore, holding cash incurs a cost of lost opportunity for a DI. Hence, it was natural for a DI to minimize the volume of cash held in its own vault while still satisfying customers' demand. To achieve this most efficiently, DIs frequently deposited/withdrew currency to/from the Fed, which resulted in cross-shipping.

The concerns over cross-shipping, and the consequent increase in the amount spent on its currency management operations, motivated the Fed to introduce the new currency recirculation policy in July 2007. The policy has two primary components: a recirculation fee on cross-shipped currency and the custodial inventory program.



Recirculation Fee: (Federal Reserve 2006) [Federal Reserve System] proposes to establish a recirculation fee to provide further incentive for deposition institutions to recirculate currency. Depository institutions would pay the fee when they deposit [fit or non-fit-sorted cash] and order the same denomination within the same business week [within] a Reserve Bank zone or subzone. The fee would not be [activated by] deposits of...[unfit] cash.

Currently, the Fed charges a per-bundle cross-shipping penalty of \$5. The volume of cross-shipping is calculated as the lesser of (1) the fit cash in the used-cash deposit and (2) the fit-cash withdrawal in the same week. Thus, the current recirculation fee is the volume (i.e., number of bundles) of cross-shipping multiplied by five.

Custodial Inventory Program: (Federal Reserve 2006) A custodial inventory enables an institution to transfer currency to the Federal Reserve Bank's books, but physically hold the currency within their secured facility, thereby reducing the investment cost of holding currency long enough to recirculate it to customers. The Custodial Inventory program is applicable only to \$10 and \$20 notes.

The program encourages fit-sorting by DIs as the currency held in custodial inventory is eligible to be lent to other institutions (and is, hence, an earning asset) and is more readily available to meet customer demand than fit cash ordered from the Fed. Note that the program does not allow a DI to deposit cash withdrawn from the Fed into custodial inventory. In other words, only the cash that has been fit-sorted by the DI is eligible to be deposited into custodial inventory.¹



Finally, it is important to note that a DI may choose to not fit-sort cash and, consequently, not participate in the custodial inventory program. Instead, the DI can choose to pay recirculation fee if it indulges in cross-shipping.

3.1.2 Literature Review and Our Contributions

There are three broad domains of research that are relevant to our work. We now briefly review them and outline our contributions along each direction.

Cash Supply Chain: Since the Fed's recirculation policy has been recently implemented, there are only a few studies that analyze the operational issues resulting from the policy. Under the simplifying assumption that both the demand of fit cash and the deposit of used cash satisfy week-to-week periodicity, both Geismar et al. (2007) and Mehrotra et al. (2010) derive characteristics of optimal policies to manage the inventory of cash. In these papers, periodicity means that the demand (resp., deposit) of fit-cash (resp., used-cash) on Day *i* (say, Monday) is exactly the same *every* week. The periodicity assumption is typically reasonable for (large) DIs that transact a high volume of physical cash each week. Our models are more general in that they allow for arbitrary demands over a finite planning horizon. Furthermore, the analyses in both Geismar et al. (2007) and Mehrotra et al. (2010) are largely based on 2-week cyclic

¹Apart from the description above, the Fed's policy includes some other operational details such as a *de minimis* exemption on cross-shipped cash. The *de minimis* exemption allows DIs to cross-ship a small volume of cash per quarter without any penalty. Moreover, the Fed intends to periodically review the *de minimis* exemption and may modify it at any time (Federal Reserve 2010c). Therefore, we have chosen not to include this exemption in our analysis. There are also some additional rules on operating custodial inventory. These details are secondary in nature and can be ignored for the purpose of assessing the overall impact of the Fed's policy.



policies. Such policies can be significantly suboptimal for medium-size DIs since such DIs typically need to schedule irregular and infrequent service requests due to relatively low volumes of cash transactions. Moreover, as mentioned earlier, our focus on medium-size DIs necessitates a significantly different cost structure for transportation and fit-sorting. We assume a fixed-cost for ordering transportation and a setup cost for fit-sorting in addition to a per-bundle fit-sorting cost, while both Geismar et al. (2007) and Mehrotra et al. (2010) assume simple per-bundle transportation and fit-sorting costs.

Rajamani et al. (2006) discuss the Fed's cash recirculation guidelines and describe the U.S. cash supply chain as a closed-loop supply chain. Dawande et al. (2010) examine the efficacy of the Fed's policy to coordinate the cash supply chain.

Our Contributions: As mentioned above, the assumptions used – and the solutions derived – in both Geismar et al. (2007) and Mehrotra et al. (2010) are not suitable for medium-size DIs. Therefore, our attempt in this paper has been to study inventory models that specifically capture the operations of medium-size DIs. We analyze two multi-period models motivated by a DI's aim to minimize the total cost incurred in managing the inventory of cash over a finite planning horizon. From a technical perspective, the non-periodicity and the different cost structure makes these models fundamentally different from those in Geismar et al. (2007) and Mehrotra et al. (2010). Therefore, a much more sophisticated analysis is required.

• Remanufacturing and Reuse: Guide (2000) surveys production planning and control activities at remanufacturing firms and discusses seven complicating characteristics



with respect to traditional manufacturing firms. Among them, the issue of balancing product returns (in our case, used-cash deposits by the customers) with demand (in our case, fit-cash demand by the customers) is related to our analysis in this paper. Guide and van Wassenhove (2001) develop a framework for analyzing the profitability of reuse activities in remanufacturing. In our model, under the Fed's new policy, DIs self-select to implement the fit-sorting (i.e., reuse) activity if it is economically attractive for them. For false-failure returns, Ferguson et al. (2006) design contracts that incentivize a retailer to increase her effort and reduce the number of returns. We expand on the conceptual similarity between false-failure returns and reuse of cash in Section 3.7.

Our Contributions: Our work studies a novel closed-loop supply chain, namely the supply chain of physical cash. The structure and operations of the cash supply chain, e.g., the different classes of cash inventories (fit, used, and unfit cash) and the ability to convert one type of cash (used cash) into another (fit cash) via fit-sorting, make the setting quite unique. The Fed's recirculation policy is an attempt to improve the efficiency of the "reverse" channel in the cash supply chain.

• Lot-Sizing Models: The problems studied in this paper are related to the classical lot-sizing literature. Since the literature in this domain is quite extensive, we avoid providing a detailed review and, instead, only mention a few seminal studies: Wagner and Whitin (1958), DeMatteis and Mendoza (1968), Balintfy (1964), and Joneja (1990). Some excellent review papers in the lot-sizing domain are Robinson et al. (2009), Drexl and Kimms (1997), Kuik et al. (1994), and Salomon et al. (1991).



Our Contributions: We address new and challenging lot-sizing models that naturally arise in a real-world setting. The uniqueness of our models stems from the combined presence of the following aspects: (1) the different classes of cash inventories (used-, fit-, and unfit cash) and the partial convertibility of used cash into fit cash, via the process of fit-sorting, (2) the novel notion of cross-shipping, the non-linear definition of cross-shipping penalty, and the Fed's mechanism (i.e., a recirculation fee that is applicable only in the presence of cross-shipping), and (3) the fixed/variable costs charged by the secure-logistics provider together with the cross-shipping costs.

Given the Fed's policy, a DI has two options: it can either (i) continue to rely on the Fed for fit-sorting and incur cross-shipping penalties, or (ii) undertake fit-sorting and reuse (some or all of) its used-cash deposits. The Basic Model (BM), developed in Section 3.2, captures a DI's mode of operations if it chooses the first option, i.e., it implements neither fit-sorting nor the custodial inventory option. Our main result in this section is a polynomial-time dynamic programming algorithm for BM. Section 3.3 discusses the reuse model (RM), in which a DI chooses the second option and implements both the components of the Fed's policy. We prove that the decision problem corresponding to RM is NP-complete and formulate RM as a mixed-integer program. In Section 3.4, via a comprehensive numerical study, we evaluate the value of local reuse for a DI and develop insights into the impacts of the DI-specific parameters and the Fed's cross-shipping fee on the effective management of physical cash. Our primary conclusion is that the saving for a DI from implementing the two components of the Fed's policy as well as the extent of local recirculation are substantial. Section 3.5 develops a rolling-horizon procedure to adapt the decisions from our models for real-time



solutions. In Section 3.6, we compare a DI's decisions under the Fed's policy with those under a socially-optimal mechanism. Finally, in Section 3.7, we summarize our results, discuss the limitations of our analysis, and outline future research directions.

3.2 The Basic Model (BM)

The Basic Model (BM) captures a DI's mode of operations if it implements neither fit-sorting nor the custodial inventory option and, instead, chooses to incur the cross-shipping penalty. We first provide a dynamic programming (DP) formulation of BM. Two important structural results (Theorems 3.2.1 and 3.2.3) allow us to significantly reduce the complexity of solving BM and show (in Theorem 3.2.5) that BM can be solved in time $O(T^3)$, where T is the length of the planning horizon. A generalized version of BM (referred to as BM^g), in which the initial and ending inventories of both fit-cash and used-cash can be arbitrary non-negative values, is discussed in Section 3.2.4.

3.2.1 Problem Formulation

As mentioned earlier, our objective is to minimize – over a given planning horizon – a DI's total cost of managing the inventory of cash, which includes (1) the ordering cost charged by the 3PLP, (2) the cross-shipping cost charged by the Fed and (3) the holding cost incurred by the DI. Ideally, an infinite horizon setting would be most appropriate, since DIs do operate under such an assumption. However, since (1) the periodicity assumption is not appropriate for medium-size DIs and (2) our interest is in precisely specifying the optimal weekly decisions under arbitrary (fit-cash) demands and (used-cash) deposits, we would need


an explicit description of this data over an infinite horizon. On the other hand, considering a single-period (i.e., one week) model defeats any attempt to plan over a reasonable horizon. Therefore, as a tradeoff, we develop our solutions for an arbitrary finite horizon. In practice, DIs typically have good visibility for about one quarter into the future. Therefore, a choice of about 15 to 20 weeks would be appropriate for medium-size DIs to plan their decisions on managing physical cash inventory.

The planning horizon consists of T periods. The Fed charges its cross-shipping penalty if both withdrawal (of fit cash) and deposit (of used cash) occur within the same week. Therefore, it is natural for a DI to plan its transportation activities (corresponding to these withdrawals and deposits) on a per-week basis. For a medium-size DI, another auxiliary reason for choosing the length of a period as one week is that the total daily volume of transactions of physical cash is not high enough to warrant daily inventory decisions. As mentioned earlier, due to the low volume of cash, 3PLPs typically prefer a fixed-fee structure, which again supports the choice of per-week decisions on transportation, fit-sorting, and custodial inventory.

Next, note that a DI maintains precise records of transactions – both deposits and withdrawals – of physical cash for a considerable duration of time, for these are audited by the Fed. Thus, a DI has access to accurate and rich data on deposits and withdrawals. For medium-size DIs, while substantial week-to-week variations in the total weekly deposits and withdrawals are common, a relatively accurate (as compared to typical retail businesses) forecast can be projected, based on this data, over a modest planning horizon of, say, the next 10 to 15 weeks. Furthermore, as we will show later in Section 3.5, even in the presence



56

of a considerable amount of variation between the forecasted and realized values of weekly deposits and withdrawals, the decisions based on forecasted data can be efficiently adapted to obtain near-optimal real-time solutions.

For the basic model, the **Timeline of the Activities** conducted by the DI is as follows.

- At the end of Period t, the inventories of used cash and fit cash are counted by the DI and decisions on (1) the quantity of used cash to be deposited to the Fed and (2) the quantity of fit cash to be withdrawn from the Fed are taken.
- The 3PLP transports the cash between the DI and the Fed. By the beginning of the Period t + 1, the DI receives the fit cash withdrawn (if any) from the Fed.
- During Period t + 1, the DI uses the fit cash inventory to satisfy demand (from customers) and collects the used cash deposited (by customers).

Notation

We now introduce the notation used for defining and analyzing the DP.

Parameters:

- T Number of periods in the planning horizon.
- s Transportation ordering cost, which is a fixed fee charged by the 3PLP.
- h Per bundle per period holding cost incurred by the DI.
- *e* Per bundle cross-shipping cost charged by the Fed.
- $\rho \qquad \mbox{The percentage of fit cash in the used cash. In our computations (Sections 3.4, 3.5, and 3.3.3), we set $\rho = 0.75$. }$
- d_t^u The projected deposit of used cash (in bundles) by the DI's customers during Period t, t = 1, 2, ..., T.
- d_t^f The projected demand of fit cash (in bundles) for the DI's customers during Period t, t = 1, 2, ..., T.



Decision Variables:

- x_t^u The quantity of used cash (in bundles) deposited to the Fed at the end of Period t, $t = 0, 1, \dots, T$.
- x_t^f The quantity of fit cash (in bundles) withdrawn from the Fed at the end of Period t, t = 0, 1, ..., T.
- X the solution vector $(x_0^u, x_0^f, x_1^u, x_1^f, \dots, x_T^u, x_T^f)$ of BM, which includes all the decisions made during the planning horizon.

Other Notation:

- I_t^u The inventory of used cash (in bundles) at the end of Period t, t = 0, 1, ..., T. I_t^f The inventory of fit cash (in bundles) at the end of Period t, t = 0, 1, ..., T.
- v_t $(I_t^u, I_t^f, x_t^u, x_t^f)$, the vector that denotes the state of the system and the decisions made at the end of Period t.
- $G_t(I_t^u, I_t^f)$ The minimum cost incurred during periods 1 through t, given that, at the end of Period t, the inventory of used cash is I_t^u and the inventory of fit cash is I_t^f . H_t The holding cost incurred during Period t.
- O_t The ordering cost charged by the 3PLP for the transportation of cash at the end of Period t-1. Without loss of generality, this cost is charged to Period t.
- R_t The cross-shipping cost (if any) incurred on the transaction of used- and fitcash between the DI and the Fed at the end of Period t - 1. Without loss of generality, this cost is charged to Period t.
- C_t The total cost incurred during Period t. Thus, $C_t = H_t + O_t + R_t$.
- C(X) The total cost incurred during the entire *T*-period planning horizon for a solution *X* of BM.

A Dynamic Programming Formulation

We now define a dynamic programming formulation for BM. We assume that the initial and ending inventory for used cash and fit cash are both 0. Thus, we have the following boundary conditions:

$$x_0^u = I_0^u = I_0^f = 0, x_0^f > 0; x_T^u = I_T^u, x_T^f = I_T^f = 0.$$

Note that in reality, the initial and ending cash positions of a DI are never zero. The Fed requires DIs to hold a minimum of one day of average daily payments of fit cash in their



vaults (Federal Reserve, 2006). Since this amount, which is essentially a safety stock, is regulated by the Fed, it is not affected by the DIs' decisions. Therefore, we only account for fit cash kept over and above this amount.

Dynamic Program (DP):

STAGE: Each period. There are T stages, t = 1, 2, ..., T.

STATE: The inventories of used cash and fit cash at the end of period t, (I_t^u, I_t^f) .

DECISION: The quantity of used cash deposited to the Fed and the quantity of fit cash withdrawn from the Fed at the end of period t, (x_t^u, x_t^f) .

Objective: $\min_{I_T^u} G_T(I_T^u, 0)$

Cost Calculation:

Given $v_{t-1} = (I_{t-1}^u, I_{t-1}^f, x_{t-1}^u, x_{t-1}^f)$ (i.e., the state of the system and the decisions made at the end of Period t-1), three types of costs are incurred in Period t:

1. Holding cost (H_t)

After the 3PLP transports the cash between the DI and the Fed, the inventory of used cash (resp., fit cash) at the beginning of Period t is $I_{t-1}^u - x_{t-1}^u$ (resp., $I_{t-1}^f + x_{t-1}^f$). At the end of period t, the inventory of used cash (resp., fit cash) is $I_t^u = I_{t-1}^u - x_{t-1}^u + d_t^u$ (resp., $I_t^f = I_{t-1}^f + x_{t-1}^f - d_t^f$). In our models, the duration of each period is one week; the week-to-week demands and deposits are deterministic, but non-stationary. The daily demand rate *within* each week is assumed to be constant. Thus, the holding cost of used cash (resp., fit cash) for Period t is calculated as the product of the per-bundle



per-period holding cost and the average of the inventories of used cash (resp., fit cash) at the beginning and the end of Period t. Thus, the holding cost of the DI during Period t is:

$$H_t(v_{t-1}) = 0.5h[(I_{t-1}^u - x_{t-1}^u) + (I_{t-1}^u - x_{t-1}^u + d_t^u) + (I_{t-1}^f + x_{t-1}^f) + (I_{t-1}^f + x_{t-1}^f - d_t^f)].$$

2. Ordering cost (O_t)

If, at the end of period t - 1, the DI either deposits used cash or withdraws fit cash, the 3PLP charges a fixed cost s during Period t.

$$O_t(v_{t-1}) = \begin{cases} 0, & \text{if } x_{t-1}^u = x_{t-1}^f = 0\\ s, & \text{otherwise.} \end{cases}$$

3. Cross-shipping cost (R_t)

If, at the end of Period t - 1, the DI both deposits used cash and withdraws fit cash, the Fed charges a cross-shipping fee during Period t. Recall from Section 3.1 that the per-bundle cross-shipping fee of e is charged on the minimum of the fit-cash withdrawn and the fit cash in the used-cash deposited. Since ρ is the percentage of fit cash in the used cash, we have

$$R_t(v_{t-1}) = e \min\{\rho x_{t-1}^u, x_{t-1}^f\}.$$

Thus, the total cost incurred during Period t is:

$$C_t(v_{t-1}) = H_t(v_{t-1}) + O_t(v_{t-1}) + R_t(v_{t-1}).$$

Note that given a solution $X = (x_0^u, x_0^f, x_1^u, x_1^f, \dots, x_T^u, x_T^f)$ of BM, the value of v_{t-1} is determined. To simplify our exposition in the proofs of Theorems 3.2.1 and 3.2.3, given a solution



X of BM, we simply use $O_t(X)$ (resp., $R_t(X)$, $H_t(X)$, and $C_t(X)$) to represent $O_t(v_{t-1})$ (resp., $R_t(v_{t-1})$, $H_t(v_{t-1})$, and $C_t(v_{t-1})$).

It is easy to see that both the state space and the number of possible decision vectors of the DP defined above are exponential in the input size of an instance of BM. We now derive two important structural results for BM that can help us significantly reduce the state space of the DP, and consequently obtain a polynomial-time algorithm for BM.

3.2.2 Two Structural Results for BM

Theorem 3.2.1 identifies a characteristic of an optimal solution regarding the withdrawals of fit-cash: there exists an optimal solution in which fit cash is withdrawn only when the fit-cash inventory becomes zero. Theorem 3.2.3 establishes a property of used-cash deposits: there exists an optimal solution in which a DI will always deposit either none or all of its used-cash inventory.

Before we introduce our results, it is instructive to compare and contrast the management of inventory in our model with that in single-product lot-sizing problems and in jointreplenishment problems (e.g., Wagner and Whitin, 1958; Joneja, 1990). Used cash and fit cash of the same denomination can be considered as two classes of the same product, with one *partially convertible* (i.e., 75% convertible) into the other via fit-sorting. The costs of managing the inventories for these two classes are inter-dependent in a somewhat complicated manner. On the one hand, if used-cash is deposited and fit-cash is ordered together in the same period, then one realizes savings as there is only one transportation order as



opposed to two if these two actions occur in two different periods. However, an additional cross-shipping penalty is incurred if both these actions occur together in the same period. Furthermore, the cross-shipping cost is proportional to the minimum of the volumes of the used-cash deposited and the fit-cash withdrawn. Clearly, there is a need to jointly manage the inventories of the two classes in BM. To further highlight our unique setting in the joint replenishment realm, note that the DI's transactions (of used- and fit-cash) with the Fed at the end of a period impact the inventories of the two classes in two opposite ways: the inventory of fit-cash increases and that of used-cash decreases after the transaction.

Theorem 3.2.1 There exists an optimal solution, say X, of BM such that either $x_t^f = 0$ or $I_t^f = 0$ for t = 1, 2, ..., T.

Proof: Consider an arbitrary solution X of BM and let t be the earliest period such that both $x_t^f > 0$ and $I_t^f > 0$. Next, let t - k be the latest period before t such that $x_{t-k}^f > 0$ and $I_{t-k}^f = 0, 1 \le k \le t$. The existence of such a period is guaranteed since we have $x_0^f > 0$ and $I_0^f = 0$. Thus, t - k has at least one possible value, namely 0. It follows immediately from the definitions of t - k and t that there is no withdrawal of fit cash by the DI (from the Fed) between these two periods. Thus, we have $x_{t-k}^f > I_t^f$.

Let $\delta = I_t^f$. We now define another solution \bar{X} of BM as follows. Let $\bar{x}_i^u = x_i^u, i = 1, 2, ..., T$; $\bar{x}_{t-k}^f = x_{t-k}^f - \delta; \bar{x}_t^f = x_t^f + \delta; \bar{x}_i^f = x_i^f, i \in \{0, 1, ..., T\} \setminus \{t - k, t\}$. The only differences between \bar{X} and X are the volumes of fit-cash withdrawals at the end of Periods t - k and t. Thus, the difference between the total ordering costs (resp., cross-shipping costs) of these two solutions includes only the differences of the ordering costs (resp., cross-shipping costs)



during Period t - k + 1 and Period t + 1. Hence,

$$O(\bar{X}) - O(X) = O_{t-k+1}(\bar{X}) - O_{t-k+1}(X) + O_{t+1}(\bar{X}) - O_{t+1}(X),$$
$$R(\bar{X}) - R(X) = R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) + R_{t+1}(\bar{X}) - R_{t+1}(X).$$

Since $x_{t-k}^f > 0, \bar{x}_{t-k}^f > 0$, we have $O_{t-k+1}(\bar{X}) - O_{t-k+1}(X) = s - s = 0$. Since $x_t^f > 0, \bar{x}_t^f > 0$, we have $O_{t+1}(\bar{X}) - O_{t+1}(X) = s - s = 0$. Thus, $O(\bar{X}) - O(X) = 0$. As compared to solution X, the total inventory of fit cash held by solution \bar{X} between the end of Period t - kand the end of Period t is less by an amount $\delta > 0$. Thus, $H(\bar{X}) - H(X) = -\delta hk$ and $\bar{I}_t^f = I_t^f - \delta = 0$. Hence, we have $C(\bar{X}) - C(X) = H(\bar{X}) - H(X) + O(\bar{X}) - O(X) + R(\bar{X}) - R(X) = -\delta hk + 0 + R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) + R_{t+1}(\bar{X}) - R_{t+1}(X)$. The possible values of $R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) = e(\min\{\rho x_{t-k}^u, \bar{x}_{t-k}^f\} - \min\{\rho x_{t-k}^u, x_{t-k}^f\})$ are as follows:

(A₁) If
$$\rho x_{t-k}^u \leq \bar{x}_{t-k}^f$$
, we have $R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) = 0$.

(A₂) If
$$\bar{x}_{t-k}^f < \rho x_{t-k}^u < x_{t-k}^f$$
, we have $R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) = e(\bar{x}_{t-k}^f - \rho x_{t-k}^u) > -e\delta$.

(A₃) If
$$\rho x_{t-k}^u \ge x_{t-k}^f$$
, we have $R_{t-k+1}(\bar{X}) - R_{t-k+1}(X) = -e\delta$.

Next, we consider the possible values of $R_{t+1}(\bar{X}) - R_{t+1}(X) = e(\min\{\rho x_t^u, \bar{x}_t^f\} - \min\{\rho x_t^u, x_t^f\})$:

- (B₁) If $\rho x_t^u \leq x_t^f$, we have $R_{t+1}(\bar{X}) R_{t+1}(X) = 0$.
- (B₂) If $x_t^f < \rho x_t^u < \bar{x}_t^f$, we have $R_{t+1}(\bar{X}) R_{t+1}(X) = e(\rho x_t^u x_t^f) < e\delta$.
- (B₃) If $\rho x_t^u \ge \bar{x}_t^f$, we have $R_{t+1}(\bar{X}) R_{t+1}(X) = e\delta$.

Thus, we have nine possible combinations $(A_i, B_j), i = 1, 2, 3; j = 1, 2, 3$. For each of the five combinations $(A_1, B_1), (A_2, B_1), (A_3, B_1), (A_3, B_2)$, and (A_3, B_3) , we have $R_{t-k+1}(\bar{X})$ –



 $R_{t-k+1}(X) + R_{t+1}(\bar{X}) - R_{t+1}(X) \le 0$. Since $-\delta hk < 0$, we have $C(\bar{X}) - C(X) < 0$. Thus, \bar{X} is a better solution than X.

Let $\gamma = x_t^f$. We define another solution \hat{X} of BM as follows: $\hat{x}_i^u = x_i^u, i = 1, 2, ..., T$; $\hat{x}_{t-k}^f = x_{t-k}^f + \gamma; \hat{x}_t^f = 0; \hat{x}_i^f = x_i^f, i \in \{0, 1, ..., T\} \setminus \{t - k, t\}$. Thus, $H(\hat{X}) - H(X) = \gamma hk$. Since $x_{t-k}^f > 0, \hat{x}_{t-k}^f > 0$, we have $O_{t-k+1}(\hat{X}) - O_{t-k+1}(X) = s - s = 0$. Since $x_t^f > 0, \hat{x}_t^f = 0$, we have $O_{t+1}(\hat{X}) - O_{t+1}(X) \leq s - s = 0$. Thus, we have $C(\hat{X}) - C(X) = H(\hat{X}) - H(X) + O(\hat{X}) - O(X) + R(\hat{X}) - R(X) \leq \gamma hk + 0 + R_{t-k+1}(\hat{X}) - R_{t-k+1}(X) + R_{t+1}(\hat{X}) - R_{t+1}(X)$. Under each of the four combinations $(A_1, B_2), (A_1, B_3), (A_2, B_2),$ and (A_2, B_3) , we have $R_{t-k+1}(\hat{X}) - R_{t-k+1}(\hat{X}) = 0, R_{t+1}(\hat{X}) - R_{t+1}(X) = -e\gamma$. Thus, $C(\hat{X}) - C(X) \leq \gamma hk - e\gamma$. We now investigate these 4 combinations individually.

- (i) (A_1, B_2) : When $\rho x_{t-k}^u \leq \bar{x}_{t-k}^f$ and $x_t^f < \rho x_t^u < \bar{x}_t^f$, we have $C(\bar{X}) C(X) = -\delta hk + e(\rho x_t^u x_t^f)$. If $C(\bar{X}) C(X) < 0$, then \bar{X} is a better solution than X. If $C(\bar{X}) C(X) \geq 0$, then we have $-\delta hk + e(\rho x_t^u x_t^f) \geq 0$. Since $0 < \rho x_t^u x_t^f < \delta$, we have e > hk. Then, $C(\hat{X}) - C(X) \leq \gamma hk - e\gamma = \gamma (hk - e) < 0$. Therefore, \hat{X} is a better solution than X.
- (ii) (A_1, B_3) : When $\rho x_{t-k}^u \leq \bar{x}_{t-k}^f$ and $\rho x_t^u \geq \bar{x}_t^f$, we have $C(\bar{X}) C(X) = -\delta hk + e\delta$. If $C(\bar{X}) C(X) < 0$, then \bar{X} is a better solution than X. If $C(\bar{X}) C(X) \geq 0$, then we have $-\delta hk + e\delta \geq 0$. Thus, $e \geq hk$ and, therefore, $C(\hat{X}) C(X) \leq \gamma hk e\gamma = \gamma(hk e) \leq 0$. Consequently, \hat{X} is at least as good as X.
- (iii) (A_2, B_2) : When $\bar{x}_{t-k}^f < \rho x_{t-k}^u < x_{t-k}^f$ and $x_t^f < \rho x_t^u < \bar{x}_t^f$, we have $C(\bar{X}) C(X) = -\delta hk + e(\bar{x}_{t-k}^f \rho x_{t-k}^u) + e(\rho x_t^u x_t^f)$. If $C(\bar{X}) C(X) < 0$, then \bar{X} is a better solution



than X. If $C(\bar{X}) - C(X) \ge 0$, then we have $-\delta hk + e(\bar{x}_{t-k}^f - \rho x_{t-k}^u) + e(\rho x_t^u - x_t^f) \ge 0$. Since $0 < (\bar{x}_{t-k}^f - \rho x_{t-k}^u) + (\rho x_t^u - x_t^f) < \delta$, we have e > hk. Therefore, $C(\hat{X}) - C(X) \le \gamma hk - e\gamma = \gamma (hk - e) < 0$. Again, \hat{X} is a better solution than X.

(iv)
$$(A_2, B_3)$$
: When $\bar{x}_{t-k}^f < \rho x_{t-k}^u < x_{t-k}^f$ and $\rho x_t^u \ge \bar{x}_t^f$, we have $C(\bar{X}) - C(X) = -\delta hk + e(\bar{x}_{t-k}^f - \rho x_{t-k}^u) + e\delta$. If $C(\bar{X}) - C(X) < 0$, \bar{X} is a better solution than X . Otherwise, if $C(\bar{X}) - C(X) \ge 0$, then we have $-\delta hk + e(\bar{x}_{t-k}^f - \rho x_{t-k}^u) + e\delta \ge 0$. Since $0 < \bar{x}_{t-k}^f - \rho x_{t-k}^u + \delta < \delta$, we have $e > hk$. Therefore, $C(\hat{X}) - C(X) \le \gamma hk - e\gamma = \gamma(hk - e) < 0$ and, hence, \hat{X} is a better solution than X .

Thus, if in a solution X of BM, there exists some t such that $x_t^f > 0$ and $I_t^f > 0$, we can obtain another solution (either \bar{X} of \hat{X}) which satisfies either $x_t^f = 0$ or $I_t^f = 0$, without increasing the total cost. Finally, if there exist multiple periods that satisfy $x_t^f > 0$ and $I_t^f > 0$, the argument above can be repeated – starting from the earliest such period – for each of these periods, to obtain an optimal solution which satisfies either $x_t^f = 0$ or $I_t^f = 0$ for t = 1, 2, ... T. The result follows.

Theorem 3.2.1 immediately implies the following result.

Corollary 3.2.2 There exists an optimal solution X of BM such that when $x_t^f > 0$, we have $x_t^f = \sum_{j=1}^k d_{t+j}^f$, for some $k \in \{1, 2, ..., T - t\}$. Also, $I_t^f = -d_t^f + \sum_{j=0}^k d_{t+j}^f$ for some $k \in \{0, 1, ..., T - t\}$. Thus, I_t^f has (T - t + 1) possible values.

While Theorem 3.2.1 examined the decisions on the withdrawal of fit cash from the Fed, the following result analyzes the decisions on the deposits of used cash to the Fed.



Theorem 3.2.3 There exists an optimal solution, say X, of BM such that either $x_t^u = 0$ or $x_t^u = I_t^u$ for every t = 1, 2, ..., T.

Proof: Let X be an arbitrary solution of BM and t be the latest period with $0 < x_t^u < I_t^u$. Then, let t + k be the first period after t such that $x_{t+k}^u = I_{t+k}^u$, $1 \le k \le T - t$. Since $x_T^u = I_T^u > 0$, note that t + k has at least one possible value, namely T. Also, we have $x_{t+k}^u > I_t^u - x_t^u$ since the definitions of t and t + k imply that no used-cash deposit occurs between these two periods.

Let $\delta = I_t^u - x_t^u > 0$. We now define another solution of BM, \bar{X} . This solution is the same as X, except for the values of the deposits of used cash at the end of Period t and Period t + k. Let $\bar{x}_t^u = I_t^u = x_t^u + \delta$, $\bar{x}_{t+k}^u = x_{t+k}^u - \delta$. Since $x_t^u > 0$, $\bar{x}_t^u > 0$, and $x_{t+k}^u > 0$, $\bar{x}_{t+k}^u > 0$, we have $O_{t+1}(\bar{X}) - O_{t+1}(X) = s - s = 0$ and $O_{t+k+1}(\bar{X}) - O_{t+k+1}(X) = s - s = 0$. Thus, $C(\bar{X}) - C(X) = -\delta hk + R_{t+1}(\bar{X}) - R_{t+1}(X) + R_{t+k+1}(\bar{X}) - R_{t+k+1}(X)$. Also, $R_{t+1}(\bar{X}) - R_{t+1}(X) = e(\min\{\rho x_t^u + \rho \delta, x_t^f\} - \min\{\rho x_t^u, x_t^f\}) \le e\rho\delta$, and $R_{t+k+1}(\bar{X}) - R_{t+k+1}(X) = e(\min\{\rho x_{t+k}^u - \rho \delta, x_{t+k}^f\} - \min\{\rho x_{t+k}^u, x_{t+k}^f\}) \le 0$. It follows that $C(\bar{X}) - C(X) \le -\delta hk + e\rho\delta = (-hk + e\rho)\delta$.

If $hk > e\rho$, we have $C(\bar{X}) < C(X)$. Thus, \bar{X} is a better solution than X.

Otherwise, if $hk \leq e\rho$, then there are three possibilities:

- 1. If $\rho x_t^u \ge x_t^f$, then, $R_{t+1}(\bar{X}) R_{t+1}(X) = 0$. Since $R_{t+k+1}(\bar{X}) R_{t+k+1}(X) \le 0$, we have $C(\bar{X}) C(X) \le 0 \delta hk < 0$, which implies \bar{X} is a better solution than X.
- 2. If $\rho x_{t+k}^u \leq x_{t+k}^f$, then $R_{t+k+1}(\bar{X}) R_{t+k+1}(X) = -e\rho\delta$. Since $R_{t+1}(\bar{X}) R_{t+1}(X) \leq e\rho\delta$,



we have $C(\bar{X}) - C(X) \leq e\rho\delta - e\rho\delta - \delta hk < 0$, which implies \bar{X} is a better solution than X.

3. If $\rho x_t^u < x_t^f$ and $\rho x_{t+k}^u > x_{t+k}^f$, we let $\gamma = x_t^u$, and consider another solution \hat{X} of BM, defined as follows: $\hat{x}_i^f = x_i^f, i = 1, 2, \dots, T$; $\hat{x}_t^u = 0$; $\hat{x}_{t+k}^u = x_{t+k}^u + \gamma$; $\hat{x}_i^u = x_i^u, i \in \{1, 2, \dots, T\} \setminus \{t, t+k\}$. Since $x_t^u > 0, \hat{x}_t^u = 0$, we have $O_{t+1}(\hat{X}) - O_{t+1}(X) \leq s - s = 0$. Since $x_{t+k}^u > 0, \hat{x}_{t+k}^u > 0$, we have $O_{t+k+1}(\hat{X}) - O_{t+k+1}(X) = s - s = 0$. Thus, we have $C(\hat{X}) - C(X) \leq \gamma hk + R_{t+1}(\hat{X}) - R_{t+1}(X) + R_{t+k+1}(\hat{X}) - R_{t+k+1}(X)$. We have $R_{t+1}(\hat{X}) - R_{t+1}(X) = -e\rho\gamma$, $R_{t+k+1}(\hat{X}) - R_{t+k+1}(X) = 0$. Therefore, $C(\hat{X}) - C(X) \leq \gamma hk - e\rho\gamma = (hk - e\rho)\gamma$. Since $hk \leq e\rho$, we have $C(\hat{X}) - C(X) \leq 0$, which implies \hat{X} is at least as good as X.

To summarize, if in a solution X of BM, there exists some t such that $0 < x_t^u < I_t^u$, we can always find another solution which satisfies either $x_t^u = 0$ or $x_t^u = I_t^u$, without increasing the total cost. If there exist multiple periods with $0 < x_t^u < I_t^u$, then the argument above can be repeated for each such period, starting from the latest of these periods. This concludes the proof.

The following result follows immediately from Theorem 3.2.3.

Corollary 3.2.4 There exists an optimal solution of BM for which $I_t^u = \sum_{j=1}^k d_{t+1-j}^u$ for some $k \in \{1, 2, ..., t\}$. Thus, I_t^u has t possible values.

Together, Theorems 3.2.1 and 3.2.3 allow us to significantly reduce the enumeration of the states when searching for an optimal solution of BM. We now introduce the forward recursion function of DP.



3.2.3 Forward Recursion Function of DP and its Computational Complexity

Recall from Section 3.2.1 that (i) $G_t(I_t^u, I_t^f)$ is the minimum cost incurred during periods 1 through t, given that, at the end of Period t, the inventory of used cash is I_t^u and the inventory of fit cash is I_t^f and (ii) $C_t(I_{t-1}^u, I_{t-1}^f, x_{t-1}^u, x_{t-1}^f)$ is the total cost incurred during Period t.

From Theorem 3.2.1 and the relationship $I_t^f = I_{t-1}^f + x_{t-1}^f - d_t^f$, to have a fit-cash inventory of I_t^f at the end of Period t, one of the following must hold at the end of Period t-1: (i) We have fit-cash inventory of 0 and order $I_t^f + d_t^f$ fit-cash from the Fed; i.e., $I_{t-1}^f = 0$ and $x_{t-1}^f = I_t^f + d_t^f$ or (ii) We have fit-cash inventory of $I_t^f + d_t^f$ and do not place an order for fit-cash from the Fed; i.e., $I_{t-1}^f = I_t^f + d_t^f$, $x_{t-1}^f = 0$. Since $I_0^u = I_0^f = x_0^u = 0$, $x_0^f > 0$, we have $I_1^u = d_1^u$. Thus, for t = 1, the forward recursion function is:

$$G_1(d_1^u, I_1^f) = C_1(0, 0, 0, I_1^f + d_1^f)$$
(3.1)

For $2 \le t \le T$, the forward recursions have the following forms, depending on the value of I_t^u :

1. If $I_t^u = d_t^u$, then the used-cash inventory is 0 at the beginning of Period t, which in turn implies that, at the end of Period t - 1, we deposit all the used-cash inventory to the Fed (i.e., $x_{t-1}^u = I_{t-1}^u$). Thus, in this case, the forward recursion function is:

$$G_t(d_t^u, I_t^f) = \min_{I_{t-1}^u} \begin{cases} G_{t-1}(I_{t-1}^u, I_t^f + d_t^f) + C_t(I_{t-1}^u, I_t^f + d_t^f, I_{t-1}^u, 0), \\ G_{t-1}(I_{t-1}^u, 0) + C_t(I_{t-1}^u, 0, I_{t-1}^u, I_t^f + d_t^f) \end{cases}$$
(3.2)

2. If $I_t^u > d_t^u$, then the used-cash inventory is positive at the beginning of Period t. From Theorem 3.2.3, we have $x_{t-1}^u = 0$. Consequently, we have $I_{t-1}^u = I_t^u - d_t^u$ and the



forward recursion function is:

$$G_t(I_t^u, I_t^f) = \min \begin{cases} G_{t-1}(I_t^u - d_t^u, I_t^f + d_t^f) + C_t(I_t^u - d_t^u, I_t^f + d_t^f, 0, 0), \\ G_{t-1}(I_t^u - d_t^u, 0) + C_t(I_t^u - d_t^u, 0, 0, I_t^f + d_t^f) \end{cases}$$
(3.3)

Theorem 3.2.5 The computational complexity of DP is $O(T^3)$.

Proof: (i) For t = 1, the forward recursion function is Equation 3.1. From Corollary 3.2.2, I_1^f has T possible values. Thus, there are T possible values of $C_1(0, 0, 0, I_1^f + d_1^f)$.

(ii) When $2 \le t \le T$, from Corollary 3.2.2, I_t^f has (T - t + 1) possible values. Also, the values of I_t^f and I_t^u are independent of each other. We have two possibilities:

- (1) The forward recursion function is Equation 3.2. Then, from Corollary 3.2.4, I_{t-1}^u has t-1 possible values. Thus, for each state defined by Equation 3.2, there are 2(t-1) possible combinations of $G_{t-1}(I_{t-1}^u, I_{t-1}^f) + C_t(I_{t-1}^u, I_{t-1}^f, x_{t-1}^u, x_{t-1}^f)$. Since there are (T-t+1) states defined by Equation 3.2, the total number of possible combinations of $G_{t-1}(I_{t-1}^u, I_{t-1}^f, x_{t-1}^u)$ for over all these states is 2(t-1)(T+1-t).
- (2) The forward recursion function is Equation 3.3. For each state defined by Equation 3.3, there are 2 possible combinations of G_{t-1}(I^u_{t-1}, I^f_{t-1}) + C_t(I^u_{t-1}, I^f_{t-1}, x^u_{t-1}, x^f_{t-1}). From Corollary 3.2.4, there are t possible values of I^u_t. Among them, (t − 1) values satisfy I^u_t > d^u_t. Thus, the number of states defined by Equation 3.3 is (t − 1)(T + 1 − t). Consequently, there are a total of 2(t − 1)(T + 1 − t) possible combinations of G_{t-1}(I^u_{t-1}, I^f_{t-1}) + C_t(I^u_{t-1}, I^f_{t-1}), x^d_{t-1}, x^d_{t-1}, x^d_{t-1}) over all the states defined by Equation 3.3.

Over all values of t (i.e., t = 1, 2, ..., T), the total number, $\xi(T)$, of possible combinations of $G_{t-1}(I_{t-1}^u, I_{t-1}^f) + C_t(I_{t-1}^u, I_{t-1}^f, x_{t-1}^u, x_{t-1}^f)$ that the DP evaluates for a T-period instance of



BM is, $T + \sum_{t=2}^{T} [2(t-1)(T+1-t) + 2(t-1)(T+1-t)] = \frac{2}{3}T^3 + \frac{1}{3}T$. Since each combination can be evaluated in time O(1), the computational complexity of DP is $O(T^3)$.

The primary implication of Theorem 3.2.5 is that if a DI decides not to implement fitsorting (and, therefore, also cannot exploit the custodial inventory option), then its optimal inventory decisions can be obtained efficiently.

3.2.4 A Generalized BM (BM^g)

The original BM (referred to here as BM^o, to avoid confusion) has the following assumptions: $I_0^f = 0, I_0^u = 0, I_T^f = 0, x_T^u = I_T^u, d_t^f > 0, d_t^u > 0, t = 1, 2, ..., T$. However, in practice, the DI may want to have some extra fit-cash inventory above the minimum amount required by the Fed (see Section 3.2.1) at the end of a planning horizon, i.e., $I_T^f > 0$. Consequently, fit-cash inventory can be positive at the beginning of the next planning horizon, i.e., $I_0^f > 0$. For practical reasons, we assume a DI's purpose of carrying fit-cash inventory over planning horizons is to satisfy customers' demand. Thus, we assume that the DI will not deposit its initial inventory of fit-cash back to the Fed. Similarly, we only consider the non-trivial case in which the DI needs to withdraw a positive amount of fit-cash during the planning horizon, i.e., $I_0^f < \sum_{t=1}^T d_t^f + I_T^f$. In this section, we study a generalized version of BM (BM^g) in which the initial and ending inventories of both fit cash and used cash are non-negative, i.e., $I_0^f \ge 0, I_0^u \ge 0, I_T^f \ge 0$, and $I_T^u \ge 0$. We first describe an intermediate problem (referred to as BM^e) which is derived from BM^g, and shares some structural similarities with BM^o. Next, we prove that BM^g and BM^e share the same optimal solution. Finally, we show that BM^e can be solved in polynomial time.



Given an instance of BM^g with $\hat{d}_t^f > 0$, $\hat{d}_t^u > 0$, t = 1, 2, ..., T, $\hat{I}_0^f \ge 0$, $\hat{I}_0^u \ge 0$, $\hat{I}_T^f \ge 0$, we now describe the procedure to convert it to an instance of BM^c: First, we copy the fit-cash demand of the first T - 1 periods from BM^g to BM^c, and set the fit-cash demand of the last period in BM^c as the sum of the fit-cash demand of the last period in BM^g and the ending inventory of fit-cash, i.e., $d_T^f = \hat{d}_T^f + \hat{I}_T^f$. Second, we determine the value of t_1 , the latest period until which the DI does not have to withdraw fit-cash from the Fed and, instead, use initial inventory to satisfy customers' demand. The value of t_1 is determined as follows: if $\hat{I}_0^f < d_1^f$, then $t_1 = 0$; otherwise, t_1 is defined by $\sum_{t=1}^{t_1} d_t^f \le \hat{I}_0^f < \sum_{t=1}^{t_1+1} d_t^f$. Finally, we make the following updates: $I_0^f = 0$, $I_T^f = 0$; $d_t^f = 0$, $t = 1, 2, ..., t_1$; $d_{t_1+1}^f = \sum_{t=1}^{t_1+1} \hat{d}_t^f - \hat{I}_0^f$; $d_t^f = \hat{d}_t^f, t_1 + 2 \le t \le T - 1$; $d_T^f = \hat{d}_T^f + \hat{I}_T^f$. Note that during the transformation, we only alter the demands of fit-cash and the initial and ending inventories of fit-cash. The used-cash deposits in BM^c are the same as in BM^g.

The following two claims characterize the relationship between the solutions of BM^g and BM^c from two aspects: feasibility and optimality.

Claim 3.2.1 Each feasible solution to an instance of BM^g is a feasible solution to the corresponding instance of BM^c , and vice-versa.

Proof: A feasible solution to an instance of BM^g , X, satisfies the following conditions:

$$\hat{I}_t^f = \hat{I}_0^f + \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t \hat{d}_i^f \ge 0, 1 \le t \le T - 1,$$
(3.4)

$$\hat{I}_{0}^{f} + \sum_{i=0}^{I-1} x_{i}^{f} - \sum_{i=1}^{I} \hat{d}_{i}^{f} - \hat{I}_{T}^{f} = 0, \qquad (3.5)$$

$$x_t^u \le \hat{I}_t^u = \sum_{i=1}^t \hat{d}_i^u - \sum_{i=0}^{t-1} x_i^u, 1 \le t \le T,$$
(3.6)



$$x_i^f \ge 0, x_i^u \ge 0, i = 0, 1, \dots, T.$$
 (3.7)

Similarly, the feasibility conditions to an instance of the corresponding BM^c are:

$$I_t^f = \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t d_i^f \ge 0, 1 \le t \le T - 1,$$
(3.8)

$$\sum_{i=0}^{T-1} x_i^f - \sum_{i=1}^T d_i^f = 0, \qquad (3.9)$$

$$x_t^u \le I_t^u = \sum_{i=1}^t d_i^u - \sum_{i=0}^{t-1} x_i^u, 1 \le t \le T,$$
(3.10)

$$x_i^f \ge 0, x_i^u \ge 0, i = 0, 1, \dots, T.$$
 (3.11)

 $\implies \text{Assume } \hat{X} \text{ is a feasible solution to an instance of BM}^g. \text{ Then } \hat{X} \text{ satisfies Constraints 3.4} \\ - 3.7. \text{ Since } d_i^u = \hat{d}_i^u, i = 1, 2, \dots, T, \text{ it is clear that } \hat{X} \text{ satisfies Constraints 3.10 and 3.11.} \\ \text{For } 1 \leq t \leq t_1, \text{ from the definition of } t_1, \text{ we have } \sum_{i=1}^t d_i^f = 0. \text{ From Constraint 3.7, we have } \\ \hat{x}_i^f \geq 0, i = 0, 1, \dots, T. \text{ Thus, Constraint 3.8 holds for } 1 \leq t \leq t_1. \text{ When } t = t_1 + 1, \text{ we have } \\ -\sum_{i=1}^{t_1+1} d_i^f = 0 - d_{t_1+1}^f = \hat{I}_0^f - \sum_{i=1}^{t_1+1} \hat{d}_i^f. \text{ For } t_1 + 2 \leq t \leq T - 1, \text{ since } \hat{d}_t^f = d_t^f, \text{ we have } \\ -\sum_{i=1}^t d_i^f = \hat{I}_0^f - \sum_{i=1}^t \hat{d}_i^f. \text{ Since } \hat{X} \text{ satisfies Constraint 3.4, it also satisfies Constraint 3.8} \\ \text{for } t_1 + 1 \leq t \leq T - 1. \text{ Finally, for } t = T, \text{ we have } -\sum_{i=1}^T d_i^f = -\sum_{i=1}^{T-1} d_i^f - d_T^f = \\ (\hat{I}_0^f - \sum_{i=1}^{T-1} \hat{d}_i^f) - (\hat{d}_T^f + \hat{I}_T^f) = \hat{I}_0^f - \sum_{i=1}^T \hat{d}_i^f - \hat{I}_T^f. \text{ Since Constraint 3.5 holds, } \hat{X} \text{ satisfies Constraint 3.9. Therefore, } \hat{X} \text{ satisfies Constraints 3.8-3.11.} \end{cases}$

 \Leftarrow Assume \bar{X} is a feasible solution to an instance of BM^c. Then \bar{X} satisfies Constraints 3.8-3.11. Since $d_i^u = \hat{d}_i^u$, i = 1, 2, ..., T, Constraints 3.6 and 3.7 are satisfied by \bar{X} . For $1 \le t \le t_1$, from the definition of t_1 , we have $\hat{I}_0^f - \sum_{i=1}^t \hat{d}_i^f \ge 0$ and $\sum_{i=1}^t d_i^f = 0$. Therefore, since \bar{X} satisfies Constraint 3.8, it satisfies Constraint 3.4 for $1 \le t \le t_1$. For $t_1 + 1 \le t \le T$, the argument is similar to that in the first part of the proof; we avoid repeating it for brevity.



Therefore, \bar{X} is also a feasible solution to the corresponding instance of BM^g. The result follows.

While Claim 3.2.1 shows that BM^{g} and BM^{c} share the same feasible set, the claim below analyzes the difference between the total costs incurred in these two instances for the same feasible solution.

Claim 3.2.2 Assume X is a feasible solution to an instance of BM^g . Let $C^g(X)$ (resp., $C^c(X)$) represent the total cost corresponding to X in BM^g (resp., the corresponding instance of BM^c). Then, we have $C^g(X) - C^c(X) = \frac{1}{2}h[(2t_1+1)\hat{I}_0^f + \hat{I}_T^f - 2\sum_{t=1}^{t_1}\sum_{i=1}^t \hat{d}_i^f]$, a quantity that is independent of X.

Proof: For X, the ordering cost (resp., cross-shipping cost) incurred in BM^g and the corresponding instance of BM^c is the same. Thus, the only difference between $C^g(X)$ and $C^c(X)$ is the holding cost for fit cash. Recall from Section 3.2.1 that the holding cost for fit-cash during Period t is $\frac{1}{2}h(I_{t-1}^f + x_{t-1}^f + I_t^f)$. Thus, we have

$$C^{g}(X) - C^{c}(X) = \frac{1}{2}h\sum_{t=1}^{T} [(\hat{I}_{t-1}^{f} + x_{t-1}^{f} + \hat{I}_{t}^{f}) - (I_{t-1}^{f} + x_{t-1}^{f} + I_{t}^{f})]$$

$$= \frac{1}{2}h\sum_{t=1}^{T} (\hat{I}_{t-1}^{f} - I_{t-1}^{f}) + \frac{1}{2}h\sum_{t=1}^{T} (\hat{I}_{t}^{f} - I_{t}^{f})$$

$$= \frac{1}{2}h(\hat{I}_{0}^{f} + \hat{I}_{T}^{f}) + h\sum_{t=1}^{T-1} (\hat{I}_{t}^{f} - I_{t}^{f}).$$

Since $\hat{I}_t^f = \hat{I}_0^f + \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t \hat{d}_i^f$, $I_t^f = I_0^f + \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t d_i^f$, $t = 1, 2, \dots, T-1$, we have $\hat{I}_t^f - I_t^f = \hat{I}_0^f + \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t \hat{d}_i^f - (I_0^f + \sum_{i=0}^{t-1} x_i^f - \sum_{i=1}^t d_i^f) = \hat{I}_0^f - \sum_{i=1}^t \hat{d}_i^f + \sum_{i=1}^t d_i^f$, $t = 1, 2, \dots, T-1$.



For $1 \le t \le t_1$, we have $\hat{I}_t^f - I_t^f = \hat{I}_0^f - \sum_{i=1}^t \hat{d}_i^f$. For $t = t_1 + 1$, we have

$$\begin{aligned} \hat{I}_{t_{1}+1}^{f} - I_{t_{1}+1}^{f} &= \hat{I}_{0}^{f} - \sum_{i=1}^{t_{1}+1} \hat{d}_{i}^{f} + \sum_{i=1}^{t_{1}+1} d_{i}^{f} \\ &= \hat{I}_{0}^{f} - \sum_{i=1}^{t_{1}+1} \hat{d}_{i}^{f} + \sum_{i=1}^{t_{1}} d_{i}^{f} + d_{t_{1}+1}^{f} \\ &= \hat{I}_{0}^{f} - \sum_{i=1}^{t_{1}+1} \hat{d}_{i}^{f} + 0 + (\sum_{i=1}^{t_{1}+1} \hat{d}_{i}^{f} - \hat{I}_{0}^{f}) = 0. \end{aligned}$$

For $t_1 + 2 \le t \le T - 1$, since $\hat{d}_t^f = d_t^f$ and $\hat{I}_{t_1+1}^f = I_{t_1+1}^f$, we have $\hat{I}_t^f - I_t^f = 0$.

Combining the above three results, we have

$$C^{g}(X) - C^{c}(X) = \frac{1}{2}h(\hat{I}_{0}^{f} + \hat{I}_{T}^{f}) + h\sum_{t=1}^{t_{1}}(\hat{I}_{0}^{f} - \sum_{i=1}^{t}\hat{d}_{i}^{f}) = \frac{1}{2}h[(2t_{1}+1)\hat{I}_{0}^{f} + \hat{I}_{T}^{f} - 2\sum_{t=1}^{t}\sum_{i=1}^{t}\hat{d}_{i}^{f}],$$

a quantity that is independent of X.

🕻 للاستشارات

The following theorem follows immediately from the above two claims.

Theorem 3.2.6 Every optimal solution to an instance of BM^g is an optimal solution to the corresponding instance of BM^c , and vice-versa.

Thus, to solve an instance of BM^g , we can solve the corresponding instance of BM^c . Next, we compare BM^c with BM^o and derive some structural properties of BM^c . In both BM^c and BM^o , we have $I_0^f = 0$, $I_T^f = 0$. However, in BM^c , the fit-cash demands of the first t_1 periods are all zero. This violates the assumption of $d_t^f > 0$, t = 1, 2, ..., T, which is required for BM^o . However, the following result shows that the property described in Theorem 3.2.1 (Section 3.2.2) continues to hold for BM^c .

Theorem 3.2.7 There exists an optimal solution, say X, of BM^c such that either $x_i^f = 0$ or $I_i^f = 0$ for i = 0, 1, ..., T.



Proof: Consider an arbitrary solution X of BM^c and let t_0 be the first period the DI withdraws fit-cash from the Fed. We first observe that $0 \le t_0 \le t_1$. Suppose $t_0 \ge t_1 + 1$, then we have $x_i^f = 0$ for $0 \le i \le t_1$, and $I_{t_1+1}^f = I_0^f + \sum_{i=0}^{t_1} x_i^f - \sum_{i=1}^{t_1+1} d_i^f = -d_1^f < 0$, which contradicts the assumption that X is an optimal solution of BM^c. Hence, $0 \le t_0 \le t_1$. It is easy to see that either $x_i^f = 0$ or $I_i^f = 0$ for $i = 0, 1, \ldots, t_0$.

Next we show that either $x_i^f = 0$ or $I_i^f = 0$ for $i = t_0 + 1, t_0 + 2, ..., T$. Let t be the earliest period such that both $x_t^f > 0$ and $I_t^f > 0$. Obviously, $t \ge t_0 + 1$. Let t - k be the latest period before t such that $x_{t-k}^f > 0$ and $I_{t-k}^f = 0, 1 \le k \le t$. The existence of such a period is guaranteed since we have $x_{t_0}^f > 0$ and $I_{t_0}^f = 0$. Thus, t - k has at least one possible value, namely t_0 . It follows immediately from the definitions of t - k and t that there is no withdrawal of fit cash by the DI (from the Fed) between these two periods. Thus, we have $x_{t-k}^f > I_t^f$. The remaining part of the proof is similar to that in the proof of Theorem 3.2.1. We, therefore, avoid repeating it for brevity.

Theorem 3.2.7 immediately implies the following result.

Corollary 3.2.8 There exists an optimal solution X of BM^c such that

- (i) We have $x_{t_0}^f > 0$ for some $t_0 \in \{0, 1, \dots, t_1\}$ and $x_t^f = 0 \ \forall t \in \{0, 1, \dots, t_1\} \setminus \{t_0\}$. Also, $x_{t_0}^f = \sum_{j=1}^k d_{t_1+j}^f$ for some $k \in \{1, 2, \dots, T - t_1\}$. For $0 \le t \le t_1$, we have $I_t^f = 0$ or $I_t^f = \sum_{j=1}^k d_{t_1+j}^f$ for some $k \in \{1, 2, \dots, T - t_1\}$. I_t^f has $(T - t_1 + 1)$ possible values.
- (ii) For $t_1 + 1 \le t \le T$, if $x_t^f > 0$, we have $x_t^f = \sum_{j=1}^k d_{t+j}^f$ for some $k \in \{1, 2, ..., T t\}$. Also, $I_t^f = -d_t^f + \sum_{j=0}^k d_{t+j}^f$ for some $k \in \{0, 1, ..., T - t\}$. Thus, I_t^f has (T - t + 1) possible values.



In the original BM (BM^o), we require DI to deposit all the used-cash inventory at the end of planning horizon, i.e., $x_T^u = I_T^u$. However, in BM^c, x_{T-1}^u is last decision made by DI for used-cash deposit. Next through three claims, we show in Theorem 3.2.9 that the property shown in Theorem 3.2.3 still holds for BM^c.

Claim 3.2.3 There exists an optimal solution, say X, of BM^c such that either $x_{T-1}^u = 0$ or $x_{T-1}^u = I_{T-1}^u$.

Proof: Let X be an solution which satisfies $0 < x_{T-1}^u < I_{T-1}^u$. Let $\delta = I_{T-1}^u - x_{T-1}^u$. We define another solution of BM^c, \bar{X} . Except for $\bar{x}_{T-1}^u = x_{T-1}^u + \delta$, all the other decisions of \bar{X} share the same value as in X. Thus, $O_T(\bar{X}) - O_T(X) = 0$, $R_T(\bar{X}) - R_T(X) = e(\min\{\rho x_{T-1}^u + \rho\delta, x_{T-1}^f\}) - \min\{\rho x_{T-1}^u, x_{T-1}^f\}) \le e\rho\delta$. It follows that $C(\bar{X}) - C(X) \le -\delta h + e\rho\delta$.

If $h > e\rho$, then $C(\bar{X}) - C(X) < 0$. Otherwise, we have $h \le e\rho$. If $\rho x_{T-1}^u \ge x_{T-1}^f$, then $R_T(\bar{X}) - R_T(X) = 0$, $C(\bar{X}) - C(X) < 0$. If $\rho x_{T-1}^u < x_{T-1}^f$, we consider another solution \hat{X} , which is the same as X, except for $\hat{x}_{T-1}^u = 0$. Thus, $O_T(\hat{X}) - O_T(X) \le 0$, $R_T(\hat{X}) - R_T(X) = 0 - e\rho x_{T-1}^u$. It follows that $C(\bar{X}) - C(X) \le x_{T-1}^u(h - e\rho) \le 0$. The results follow.

Using the similar method, we can prove the following claim.

Claim 3.2.4 There exists an optimal solution, say X, of BM^c such that if $x_{T-1}^u = 0$, then either $x_t^u = 0$ or $x_t^u = I_t^u$ for every t = 1, 2, ..., T.

Proof: Let X be an solution which satisfies $x_{T-1}^u = 0$ and $0 < x_t^u < I_t^u$, $0 \le t < T - 1$. By using a similar method as in the proof of Claim 3.2.3, we can obtain a solution no worse than X by only changing x_t^u to either 0 or I_t^u . We omit the details here for brevity.



The proof of following claim is similar to that of Theorem 3.2.3.

Claim 3.2.5 There exists an optimal solution, say X, of BM^c such that if $x_{T-1}^u = I_{T-1}^u$, then either $x_t^u = 0$ or $x_t^u = I_t^u$ for every t = 1, 2, ..., T.

Combining Claims 3.2.3, 3.2.4, 3.2.5, we have the following theorem.

Theorem 3.2.9 There exists an optimal solution, say X, of BM^c such that either $x_i^u = 0$ or $x_i^u = I_i^u$ for i = 0, 1, ..., T - 1.

We can now use a forward recursion similar to that described in Section 3.2.3 to optimize BM^c . The computational complexity of the corresponding DP for BM^c remains $O(T^3)$. Our next task is to discuss the scenario when the DI implements the Fed's policy.

3.3 The Reuse Model (RM): Inventory Decisions under the Fed's Recirculation Policy

Under the (local) Reuse Model (RM), a DI can recirculate cash by using the fit-sorting and custodial inventory options. We first describe the model in Section 3.3.1. Then, in Section 3.3.2, we show that the decision problem corresponding to RM is NP-complete and establish a structural property for RM. Finally, in Section 3.3.3, we provide a mixed-integer programming formulation of RM that is used in Section 3.4.

3.3.1 Problem Description

Structurally, RM is a much more complicated model than BM. There are four additional parameters (defined below) in RM. Moreover, for each period, RM has four additional decision



variables and two more types of inventories. Under RM, the **Timeline of the Activities** conducted by the DI and the 3PLP is significantly different than that in BM and is described below.

- At the end of Period t, the DI takes the following actions (in the given order) (i) counts the inventories of used cash, unfit cash, fit cash, and custodial inventory, (ii) decides the quantity of used cash to fit-sort. As mentioned earlier in Section 3.1, after fit-sorting (if any), about 75% of the fit-sorted used cash is fit cash and the remaining 25% is unfit cash, (iii) decides the quantity of fit-sorted cash to be placed in custodial inventory. Thus, the remainder of the fit-sorted cash will be directly placed in the DI's fit-cash inventory, (iv) decides the quantities of used cash and unfit cash to be deposited to the Fed, the quantity of fit cash to be withdrawn from the Fed, and the quantity of fit cash to be withdrawn from custodial inventory.
- After notified by the DI about the decisions above, the 3PLP takes the following actions (in the given order) by the beginning of Period t + 1: (i) fit-sorts the used cash and places the resulting fit cash (resp., unfit cash) in custodial inventory and/or fit-cash inventory (resp., unfit cash inventory) based on the DI's preference, (ii) if required, transports the used cash and/or unfit cash from the DI to the Fed and withdraws fit cash from the Fed. By the beginning of the Period t + 1, the DI receives the fit cash (if any) withdrawn from the Fed.
- During Period t+1, the DI uses the fit-cash inventory as well as custodial inventory to satisfy demand (from customers) and collects the used cash deposited (by customers).



Note: In RM, the DI maintains two types of fit cash inventories: the regular fit cash in its own vault and the fit cash in custodial inventory (managed by the 3PLP). We use I_t^f (to

retain the same notation as in BM) to denote the former and I_t^c to denote the latter.

Notation

For consistency, the basic notation is the same as that defined for BM in Section 3.2.1. We,

therefore, define only the additional notation required for RM.

Parameters:

s_r	Setup cost for fit-sorting.
s_c	Setup cost for withdrawal from custodial inventory.
h_c	Per bundle per period custodial inventory holding cost.
r	Per bundle fit-sorting cost.

Decision Variables:

The quantity of unfit cash (in bundles) deposited to the Fed at the end of
Period $t, t = 0, 1,, T$.
The quantity of used cash (in bundles) fit-sorted at the end of Period $t, t =$
$0, 1, \ldots, T.$
The quantity of fit cash (in bundles) deposited to custodial inventory at the
end of Period $t, t = 0, 1, \dots, T$.
The quantity of fit cash (in bundles) from custodial inventory needed to satisfy
fit-cash demand during Period $t + 1, t = 0, 1,, T - 1$. This decision is made
at the end of Period t. For convenience of notation, we let $x_T^{c-} = 0$.
The solution vector $(x_t^u, x_t^f, x_t^n, x_t^r, x_t^{c+}, x_t^{c-}), t = 0, 1, \dots, T$, which includes all
the decisions made during the planning horizon.

Other Notation:

y_t^o	$y_t^o = 0$, if $x_t^u = x_t^f = x_t^n = 0$, $t = 0, 1,, T$; $y_t^o = 1$, otherwise.
y_t^c	$y_t^c = 1$, if $x_t^{c-} > 0$, $t = 0, 1, \dots, T$; $y_t^c = 0$, otherwise.
y_t^r	$y_t^r = 1$, if $x_t^r > 0$, $t = 0, 1,, T$; $y_t^r = 0$, otherwise.
I_t^n	The inventory of unfit cash (in bundles) at the end of Period $t, t = 0, 1,, T$.
I_t^c	The custodial inventory (in bundles) at the end of Period $t, t = 0, 1,, T$.
H_t^c	The cost charged by the 3PLP for holding custodial inventory during Period t .
O_t^c	The ordering cost charged by the 3PLP for withdrawal from custodial inventory
	during Period t .



F_t	The cost charged by the 3PLP for fit-sorting cash at the end of Period $t - 1$.
	Without loss of generality, this cost is charged to Period t .
C_t	The total cost incurred during Period t. Thus, $C_t = H_t + H_t^c + O_t + O_t^c + R_t + F_t$.

Cost Calculation

The total cost over the planning horizon is the sum of the following six costs:

1. Holding cost for used cash, (regular) fit cash, and unfit cash: $H = \sum_{t=0}^{T} [0.5h(2I_t^u - d_t^u) + 0.5h(2I_t^f + d_t^f - x_{t-1}^{c-}) + hI_t^n].$

Note: Since the inventory of used cash at the end of Period t is I_t^u , and the deposit of used cash during Period t is d_t^u , retrospectively, at the beginning of Period t, the inventory is $I_t^u - d_t^u$. Thus, the average inventory of used cash during Period t is $0.5(2I_t^u - d_t^u)$. Similarly, there is I_t^f regular fit cash at the end of Period t and the customers withdraw d_t^f during Period t. Since the withdrawal of x_{t-1}^{c-} from custodial inventory is also used to satisfy the customers' demand, retrospectively, at the beginning of Period t, the fit-cash inventory is $I_t^f + d_t^f - x_{t-1}^{c-}$. Thus, the average inventory of fit cash during Period t is $0.5(2I_t^f + d_t^f - x_{t-1}^{c-})$. Since there is no change to unfit-cash inventory during a period, the average inventory of used cash during Period t is I_t^n .

2. Custodial inventory holding cost: $H^c = \sum_{t=0}^T 0.5h_c(2I_t^c + x_{t-1}^{c-}).$

Note: Since the custodial inventory at the end of Period t is I_t^c , and x_{t-1}^{c-} custodial inventory is withdrawn during Period t to satisfy the customers' demand, retrospectively, at the beginning of Period t, the custodial inventory is $I_t^c + x_{t-1}^{c-}$. Thus, the average custodial inventory during Period t is $0.5(2I_t^c + x_{t-1}^{c-})$.



- 3. Ordering cost: $O = \sum_{t=0}^{T} sy_t^o$.
- 4. Custodial inventory setup cost: $O^c = \sum_{t=0}^{T} s_c y_t^c$.
- 5. Cross-shipping cost: $R = \sum_{t=0}^{T} e \min\{\rho x_t^u, x_t^f\}.$
- 6. Fit-sorting cost: $F = \sum_{t=0}^{T} (s_r y_t^r + r x_t^r).$

3.3.2 Computational Complexity and a Structural Property of RM

From the precise description above, it is intuitive that the decisions in RM are significantly more complex than those in BM. The following result corroborates this intuition by formally establishing the hardness of RM.

Theorem 3.3.1 The decision problem corresponding to RM is NP-Complete.

Proof: The NP-Complete problem which we use in our reduction is the PARTITION PROB-LEM (Garey and Johnson, 1979).

PARTITION PROBLEM (PP)

INSTANCE: A set of positive integers $A = \{a_1, a_2, \dots, a_n\}$, a positive integer B, such that $\sum_{i=1}^{n} a_i = 2B$.

SOLUTION: A subset $A_1 \subset A$, such that $\sum_{a_i \in A_1} a_i = B$.

Given an arbitrary instance of PP, we construct an instance of RM as follows. Let T = n+2. $d_1^u = \frac{4}{3}B, d_2^u = d_3^u = \ldots = d_{n+2}^u = 0$. $d_1^f = d_2^f = 0, d_3^f = a_1, d_4^f = a_2, \ldots, d_{n+2}^f = a_n$. Let $s_r = 0, r = 0, h_c = 0, s = s_c = 1, h \ge 2n+3$. For the constructed instance, we consider the following decision question.



<u>DECISION QUESTION</u>: Let $D = \frac{7}{6}hB + n + 1$. Does there exist a solution X for RM such that $C(X) \le D$?

The construction of the decision problem from the given instance of PP is polynomially bounded. The decision question, therefore, is in class NP. We now show that the decision question has an affirmative answer if and only if the set A has a subset A_1 such that $\sum_{a_i \in A_1} a_i = B$.

 \Leftarrow Let $A_1 \subset A$ be such that $\sum_{a_i \in A_1} a_i = B$. We let $n_1 = |A_1|$, the number of elements in set A_1 . Consider the following solution of the constructed instance of RM: $X = \{x_1^r = \frac{4}{3}B; x_1^{c+} = B; x_1^n = \frac{1}{3}B; x_{1+i}^f = a_i, a_i \in A_1; x_{1+i}^{c-} = a_i, a_i \in A \setminus A_1; \text{ all other decision variables}$ are 0}. The solution X can be interpreted as follows. At the end of Period 1, the DI fit-sorts all the used cash it collects during that period, places all the resulting fit cash in custodial inventory, and deposits the unfit cash to the Fed. For $t \in \{3, 4, \ldots, n+2\}$, the DI withdraws just enough fit cash to satisfy the demand of that period either entirely from the Fed at the end of Period t - 1 or entirely from custodial inventory during Period t.

For the solution X, the holding cost is incurred from two sources: the holding cost for used cash during Period 1 and the holding cost for fit cash during the periods when we use withdrawal from the Fed to satisfy fit-cash demand. Thus, $H(X) = \frac{1}{2}h(d_1^u + \sum_{t=2}^{n+2} x_t^f) =$ $\frac{1}{2}h(\frac{4}{3}B + B) = \frac{7}{6}hB$. Since $h_c = 0$, we have $H^c(X) = 0$. Also, R(X) = 0, F(X) = 0, O(X) = $s(1 + n_1) = 1 + n_1$, $O^c(X) = s_c(n - n_1) = (n - n_1)$. Thus, the total cost of X is:

$$C(X) = H(X) + H^{c}(X) + R(X) + F(X) + O(X) + O^{c}(X)$$

= $\frac{7}{6}hB + 0 + 0 + 0 + (1 + n_{1}) + (n - n_{1}) = \frac{7}{6}hB + n + 1 = D$



 \implies Suppose X is a solution of RM such that $C(X) \leq D$. The following claims help us to characterize the required set A_1 .

Claim 3.3.1 $x_t^u = 0, t = 0, 1, \dots, n+2.$

Proof: Since the sum of all used cash deposits over the planning horizon is an obvious upper bound on the total fit-sorted cash, we have $\sum_{t=0}^{n+1} x_t^r \leq \sum_{t=1}^{n+2} d_t^u = \frac{4}{3}B$. The fit cash placed in custodial inventory has to necessarily come from fit-sorting; thus, $\sum_{t=0}^{n+1} x_t^{c-} \leq \sum_{t=0}^{n+1} x_t^{c+} \leq \frac{3}{4} \sum_{t=0}^{n+1} x_t^r = B$. Also since the demand of fit cash has to be satisfied by withdrawals either from the Fed or from custodial inventory, we have $\sum_{t=0}^{n+1} x_t^f + \sum_{t=0}^{n+1} x_t^{c-} = \sum_{t=1}^{n+2} d_t^f = \sum_{i=1}^n a_i = 2B$. Consequently, we have $\sum_{t=0}^{n+1} x_t^f = 2B - \sum_{t=0}^{n+1} x_t^{c-} \geq 2B - B = B$.

When we withdraw x_t^f fit cash from the Fed at the end of Period t, we incur a holding cost of at least $\frac{1}{2}hx_t^f$ during Period t + 1. Similarly, we incur a holding cost of at least $\frac{1}{2}hd_1^u$ for used cash during Period 1. Thus,

$$C(X) \ge \frac{1}{2}h\sum_{t=0}^{n+1} x_t^f + \frac{1}{2}hd_1^u = \frac{1}{2}h\sum_{t=0}^{n+1} x_t^f + \frac{2}{3}hB$$

Assume $\sum_{t=0}^{n+1} x_t^f > B$. Then, we have $\sum_{t=0}^{n+1} x_t^f \ge B + 1$ since all decision variables are integral. Since $h \ge 2n+3$, we have

$$C(X) \ge \frac{2}{3}hB + \frac{1}{2}h\sum_{t=0}^{n+1} x_t^f \ge \frac{2}{3}hB + \frac{1}{2}hB + \frac{1}{2}h > \frac{7}{6}hB + n + 1 = D,$$

which contradicts $C(X) \leq D$. Therefore, $\sum_{t=0}^{n+1} x_t^f = B$. Then, $\sum_{t=0}^{n+1} x_t^{c-} = 2B - \sum_{t=0}^{n+1} x_t^f = B$. B. Consequently, $\sum_{t=0}^{n+1} x_t^{c+} = B$ and $\sum_{t=0}^{n+1} x_t^r = \frac{4}{3}B$. Since $\sum_{t=0}^{n+1} x_t^r + \sum_{t=0}^{n+2} x_t^u = \sum_{t=1}^{n+2} d_t^u$, we have $\sum_{t=0}^{n+2} x_t^u = \frac{4}{3}B - \frac{4}{3}B = 0$. Thus, $x_t^u = 0, t = 0, 1, \dots, n+2$.

Claim 3.3.2 $x_0^f = x_1^f = x_0^{c-} = x_1^{c-} = 0; x_1^r = \frac{4}{3}B, x_t^r = 0, t = 2, 3, \dots, n+2; x_1^{c+} = B, x_t^{c+} = 0, t = 2, 3, \dots, n+2; x_1^n = \frac{1}{3}B, x_1^n = 0, t = 2, 3, \dots, n+2.$



Proof: From Claim 3.3.1, we have $C(X) \ge \frac{2}{3}hB + \frac{1}{2}h\sum_{t=0}^{n+1} x_t^f = \frac{7}{6}hB$. Moreover, a holding cost of at least $\frac{1}{2}hI_t^f$ (resp., $\frac{1}{2}hI_t^u$, $\frac{1}{2}hI_t^n$) is incurred on the fit-cash inventory (resp., used-cash inventory, unfit-cash inventory) during Period t. Thus,

$$C(X) \ge \frac{7}{6}hB + \frac{1}{2}h(\sum_{t=1}^{n+2}I_t^f + \sum_{t=1}^{n+2}I_t^n + \sum_{t=2}^{n+2}I_t^u).$$

Using an argument similar to that in the proof of Claim 6, we have $\sum_{t=1}^{n+2} I_t^f + \sum_{t=1}^{n+2} I_t^n + \sum_{t=2}^{n+2} I_t^u = 0$. Thus, $I_t^f = 0, t = 1, 2, ..., n+2$; $I_t^n = 0, t = 1, 2, ..., n+2$; $I_t^u = 0, t = 1, 2$

Since $I_1^f = I_2^f = 0$, $d_1^f = d_2^f = 0$, there is no withdrawal of fit cash either from the Fed at the end of Periods 0 and 1 or from custodial inventory during the Periods 1 and 2, i.e., $x_0^f = x_1^f = x_0^{c^-} = x_1^{c^-} = 0$. Since $0 = I_2^u = I_1^u - x_1^u - x_1^r + d_2^u = \frac{4}{3}B - 0 - x_1^r + 0$, we have $x_1^r = \frac{4}{3}B$. From Claim 3.3.1, we have $\sum_{t=0}^{n+1} x_t^r = \frac{4}{3}B$; thus, $x_t^r = 0, t = 2, 3, \dots, n+2$. Since $x_t^{c+} \le x_t^r$, we have $x_t^{c+} = 0, t = 2, 3, \dots, n+2$. Since $0 = I_2^f = I_1^f + 0.75x_1^r - x_1^{c^+} + x_1^f + x_1^{c^-} - d_2^f = 0 + B - x_1^{c^+} + 0 + 0 - 0$, we have $x_1^{c^+} = B$. Using $0 = I_{t+1}^n = I_t^n + 0.25x_t^r - x_t^n$, we have $x_1^n = \frac{1}{3}B, x_t^n = 0, t = 2, 3, \dots, n+2$.

Claim 3.3.3 $\sum_{t=2}^{n+1} x_t^f = B$, $\sum_{t=2}^{n+1} x_t^{c-} = B$. If $x_t^{c-} > 0$, then $x_t^{c-} = d_{t+1}^f$, $2 \le t \le n+1$.

Proof: From Claim 3.3.1, we have $\sum_{t=0}^{n+1} x_t^f = B$, $\sum_{t=0}^{n+1} x_t^{c-} = B$. From Claim 3.3.2, we have $x_0^f = x_1^f = x_0^{c-} = x_1^{c-} = 0$. Thus, $\sum_{t=2}^{n+1} x_t^f = B$ and $\sum_{t=2}^{n+1} x_t^{c-} = B$.

Since $I_t^f = 0, t = 1, 2, ..., n + 2$, we do not have positive fit cash inventory at the end of these periods. Then, for each period $t \in \{3, 4, ..., n+2\}$, since $d_t^f > 0$, fit cash is withdrawn either from the Fed at the end of Period t - 1 or from custodial inventory during Period t to satisfy the fit-cash demand of that period. Thus, for $t \in \{3, 4, ..., n+2\}$, we incur either



the ordering cost or setup cost for withdrawal from custodial inventory (or both) during Period t. Thus, $O_t(X) + O_t^c(X) \ge 1$, for $3 \le t \le n+2$. Suppose, in X, there exists some \tilde{t} such that $x_{\tilde{t}}^{c-} > 0$ and $x_{\tilde{t}}^{c-} < d_{\tilde{t}+1}^f$. Then, we have $x_{\tilde{t}}^f = d_{\tilde{t}+1}^f - x_{\tilde{t}}^{c-} > 0$. Thus, during Period $\tilde{t} + 1$, both the setup costs s and s_c are incurred, i.e., $O_{\tilde{t}}(X) + O_{\tilde{t}}^c(X) = 2$. Thus, $C(X) \ge \frac{7}{6}hB + 1 + 2 + (n-1) > D$, which contradicts $C(X) \le D$. The result now follows.

Let $I = \{i | x_{i+1}^{c-} > 0\}, A_1 = \{a_i | i \in I\}$. As a consequence of Claim 3.3.3, we have $\sum_{t=2}^{n+1} x_t^{c-} = \sum_{i \in I} d_{i+2}^f = B$. Since $d_{i+2}^f = a_i, i = 1, 2, ..., n$, we have $\sum_{a_i \in A_1} a_i = B$. This concludes the proof of Theorem 3.3.1.

We now prove a structural property of RM.

Theorem 3.3.2 If $r < e\rho$, then there exists an optimal solution, say X, of RM such that $x_t^f = 0$ or $x_t^u = 0$ or $x_t^r = 0$ for t = 1, 2, ..., T.

Proof: Consider an arbitrary solution X of RM and let t be the period such that $x_t^f > 0$, $x_t^u > 0$, and $x_t^r > 0$. Let $\gamma = \min\{\frac{1}{\rho}x_t^f, x_t^u\}$ and consider another solution \bar{X} of RM, defined as follows. Let $\bar{x}_t^u = x_t^u - \gamma; \bar{x}_t^r = x_t^r + \gamma; \bar{x}_t^f = x_t^f - \rho\gamma; \bar{x}_t^n = x_t^n + (1-\rho)\gamma; \bar{x}_i^u = x_i^u, \bar{x}_t^r = x_t^r, \bar{x}_t^f = x_t^f, \bar{x}_i^{n} = x_i^n, i \in \{1, 2, \dots, T\} \setminus \{t\}; \bar{x}_t^{c+} = x_t^{c+}; \bar{x}_t^{c-} = x_t^{c-}, i = 1, 2, \dots, T.$ Thus, we have either $\bar{x}_t^u = 0$ or $\bar{x}_t^f = 0$. Since $x_t^u > 0$ and $\bar{x}_t^n > 0$, we have $O_{t+1}(\bar{X}) - O_{t+1}(X) = s - s = 0$. Also, $x_t^r > 0$ and $\bar{x}_t^r = x_t^r + \gamma > 0$. Hence, we have $F_{t+1}(\bar{X}) - F_{t+1}(X) = (s_r + r\bar{x}_t^r) - (s_r + rx_t^r) = r\gamma$. Next, note that $R_{t+1}(\bar{X}) - R_{t+1}(X) = e(\min\{\rho x_t^u - \rho\gamma, x_t^f - \rho\gamma\} - \min\{\rho x_t^u, x_t^f\}) = -e\rho\gamma$. Furthermore, $H(\bar{X}) - H(X) = 0$, $H^c(\bar{X}) - H^c(X) = 0$, and $O^c(\bar{X}) - O^c(X) = 0$. Therefore, $C(\bar{X}) - C(X) = O_{t+1}(\bar{X}) - O_{t+1}(X) + F_{t+1}(\bar{X}) - F_{t+1}(X) + R_{t+1}(\bar{X}) - R_{t+1}(X) = (r - e\rho)\gamma$. Hence, if $r < e\rho$, we have $C(\bar{X}) - C(X) < 0$, which implies \bar{X} is a better solution than X.



Thus, in the solution X of RM, if $r < e\rho$ and there exists some t with $x_t^f > 0$, $x_t^u > 0$, and $x_t^r > 0$, then it is possible to obtain a better solution (\bar{X}) in which at least one of x_t^f, x_t^u , and x_t^r , is zero for t = 1, 2, ..., T. To conclude the proof, note that the above argument can be applied repeatedly if there exist multiple periods that satisfy $x_t^f > 0$, $x_t^u > 0$, and $x_t^r > 0$.

Theorem 3.3.2 can be intuitively interpreted in the following manner: If the cost of fit-sorting one extra bundle of used-cash (i.e., r, the per bundle fit-sorting cost) is less than the marginal saving from using the resulting fit cash to reduce cross-shipping (i.e., $e\rho$, the product of the per bundle cross-shipping cost and the percentage of fit cash in the used cash), then a DI will not both deposit used-cash and withdraw fit-cash in the same period. For otherwise, fitsorting an extra bundle and, thereby, reducing the amount of fit-cash withdrawal results in a better solution. Using this argument iteratively, we can conclude that either the quantity of used-cash deposit or the quantity of fit-cash withdrawal becomes 0, which implies that cross-shipping is eliminated in that period.

Theorem 3.3.1 above, together with the significant recent advances in computational integer programming motivate us to formulate RM as a mixed-integer program.

3.3.3 Formulating RM as a Mixed-Integer Program (MIP)

We retain most of the notation from Sections 3.2.1 and 3.3.1, and only define the additional notation required for the formulation.

Additional Notation:

 z_t

 $z_t = 1$, if $x_t^f \le x_t^u, t = 0, 1, \dots, T$; $z_t = 0$, otherwise.



The Objective Function

Our objective is to minimize the sum of six types of costs (see Section 3.3.1), i.e.,

$$\min \left(O + O^c + H + H^c + F + R \right)$$

Constraints

We have five main classes of constraints.

1. Inventory Balance Constraints:

Recall that all the inventories are counted at the end of each period and before any decisions are taken. There are four types of inventories.

(i) Used Cash Inventory Balance:

$$I_{t+1}^u = I_t^u - x_t^r - x_t^u + d_{t+1}^u \qquad t = 0, 1, \dots, T-1$$

(ii) Unfit Cash Inventory Balance:

$$I_{t+1}^n = I_t^n + 0.25x_t^r - x_t^n \qquad t = 0, 1, \dots, T - 1$$

(iii) Fit Cash Inventory Balance:

$$I_{t+1}^f = I_t^f + 0.75x_t^r - x_t^{c+} + x_t^f + x_t^{c-} - d_{t+1}^f \qquad t = 0, 1, \dots, T-1$$

(iv) Custodial Inventory Balance:

$$I_{t+1}^c = I_t^c + x_t^{c+} - x_t^{c-} \qquad t = 0, 1, \dots, T-1$$



2. Cross-Shipping Constraints:

Recall that $R_{t+1} = e \min\{0.75x_t^u, x_t^f\}$. Since we are minimizing the objective function, this constraint can be replaced by the following two constraints:

$$R_{t+1} \ge 0.75e(x_t^u - z_t \sum_{t=1}^T d_t^u) \qquad t = 0, 1, \dots, T-1$$
$$R_{t+1} \ge e[x_t^f - (1 - z_t) \sum_{t=1}^T d_t^f] \qquad t = 0, 1, \dots, T-1$$

3. Constraints for Fit-Sorting:

At the end of each period, the sum of the quantity fit-sorted and the quantity of used-cash deposited cannot exceed the available used-cash inventory.

$$x_t^u + x_t^r \le I_t^u \qquad t = 0, 1, \dots, T$$

The quantity of fit cash placed in custodial inventory cannot exceed the quantity of fit-sorted cash.

$$x_t^{c+} \le 0.75 x_t^r$$
 $t = 0, 1, \dots, T-1$

4. Constraints for Computing Setup and Ordering Costs:

We have 3 types of setup and ordering costs:

(i) At the end of Period t, an ordering cost is charged if the DI requests the 3PLP to conduct any of the following three activities: (1) deposits used cash to the Fed,
(2) deposits unfit cash to the Fed, (3) withdraws fit cash from the Fed.

Thus,



$$x_t^u + x_t^n \le y_t^o \sum_{t=1}^T d_t^u \qquad t = 0, 1, \dots, T - 1$$
$$x_t^f \le y_t^o \sum_{t=1}^T d_t^f \qquad t = 0, 1, \dots, T - 1$$

(ii) At the end of Period t, a setup cost for custodial inventory is charged if the DI requests to withdraw from its custodial inventory. Thus,

$$x_t^{c-} \le 0.75 y_t^c \sum_{t=1}^T d_t^u \qquad t = 0, 1, \dots, T-1$$

(iii) At the end of Period t, a setup cost for fit-sorting is charged if the DI requests the 3PLP to conduct fit-sorting. Thus,

$$x_t^r \le y_t^r \sum_{t=1}^T d_t^u$$
 $t = 0, 1, \dots, T - 1$

5. Non-Negativity and 0-1 Restrictions:

$$\begin{aligned} x_t^u, x_t^f, x_t^r, x_t^{c+}, x_t^n, x_t^{c-} &\geq 0 & t = 0, 1, \dots, T-1 \\ I_t^f, I_t^c, I_t^n &\geq 0 & t = 0, 1, \dots, T-1 \\ y_t^o, y_t^c, y_t^r, z_t &\in \{0, 1\} & t = 0, 1, \dots, T-1 \end{aligned}$$

As will be seen in the next section, CPLEX (version 11.1.1) is able to solve realistic instances of RM in reasonable time. Our ability to solve instances of BM and RM to optimality allows us to evaluate the impact of local reuse of cash and derive a variety of insights on the forces that influence the volume of cross-shipping. This will be our goal in the next section.



3.4 Understanding the Value of Local Reuse and Influences on the Volume of Cross-Shipping

Recall that a DI may choose not to fit-sort its used-cash deposits and, instead, opt to pay recirculation fees for cross-shipping. To resolve the question of whether or not to fitsort, it becomes important for managers of DIs to quantify the impact of fit-sorting and participating in the custodial inventory program. Subsequently, if the DI decides to fit-sort, the optimal amount to fit-sort and the influence of the problem parameters on the volume of cross-shipping helps understand the various tradeoffs involved. We address these issues via a comprehensive numerical study.

An ideal metric, which we refer to as the Value of Local Reuse (VoR), to capture the monetary impact from implementing fit-sorting and custodial inventory is the percentage cost saving offered by RM over BM. That is,

$$VoR = \frac{(Objective Value of BM - Objective Value of RM) \times 100\%}{Objective Value of BM}$$

Similarly, the **Extent of Local Reuse** (EoR) is the percentage of fit cash demand satisfied by fit-sorting (i.e., by local recirculation) in RM. Since about 75% of the used-cash is fit-cash (Section 3.1), we have

$$EoR = \frac{(Total Quantity Fit-Sorted over the Planning Horizon) \times 75\%}{The Total Demand of Fit Cash over the Planning Horizon}$$

Another characteristic that is expected to influence the extent of cross-shipping for a DI is the imbalance between its average per-period deposit and demand. Accordingly, we define a DI's **Imbalance Ratio**, α , as follows:



$$\alpha = \frac{\text{Average Per-Period Used Cash Deposit}}{\text{Average Per-Period Fit-Cash Demand}}.$$

We first describe our test bed in Section 3.4.1 and then discuss the results and insights in Section 3.4.2.

3.4.1 The Test Bed

To ground our test bed in reality, we use information from the Fed's publicly-available documents (Federal Reserve 2006, 2009) and also from the relevant literature (Mehrotra et al. 2010, Dawande et al. 2010).

As mentioned in Section 3.1, the weekly demand of fit cash for medium-size DIs is typically between 1000 to 2000 bundles. Thus, we choose the weekly demand of fit cash (in bundles) randomly from U[1000, 2000]. As observed in Mehrotra et al. (2010), discussions with practitioners indicate that there is little week-to-week variation in the value of the ratio α (defined above) for a specific DI; however, across DIs, the value of α may vary and typically ranges between 1 and 3. Accordingly, we consider four values of α : 1, 1.5, 2, and 3. Thus, the weekly deposit of used cash is generated randomly from $U[1000\alpha, 2000\alpha]$. We consider three possible values for the length of the planning horizon, T: 5, 10, 15.

To evaluate the impact of the cross-shipping fee e on the optimal solution, we consider three values: \$5 (the current value), \$6, and \$7. The dominant component of a DI's perbundle holding cost, h, of fit cash is the cost of lost opportunity of lending via the Federal Funds Market. Between 1994 to 2008, the annual Federal Funds Rate has varied between


a minimum of 1.13% and a maximum of 6.24% (Federal Reserve 2010a). Currently (2009) and early 2010), this rate is below 1%, influenced largely by the economic downturn that started in late 2007. Assuming a rate of 1%, the holding cost per-bundle per-period (week) is $20 \times 1000 \times 1\% / 50 = 4$. Anticipating a moderate rise of this rate in the future, we consider two additional values: 3% and 5%. Thus, we consider three values of h: \$4, \$12, and \$20. Based on discussions with a secure-logistics provider, we choose three values of the fixed ordering cost, s, of transportation: \$1500, \$2000, \$2500. For a fair comparison of the two models (BM and RM) for assessing the value and extent of reuse, we avoid varying those parameters that are relevant only to RM, namely h_c, r, s_r , and s_c (Section 3.3.1). Recall that deposits to custodial inventory are credited to the DI's account with the Fed and can be lent (to other institutions) by the DI. Hence, such deposits incur no cost of lost lending opportunity. Consequently, the per-bundle per-period holding cost of custodial inventory, h_c , is only the physical storage cost, which we set at 1. The per-bundle fit-sorting fee, r, is set at \$2. Both the setup costs s_c and s_r are administrative fees that are expected to be modest; we fix them to \$200 and \$1000, respectively. For each combination of the horizon-length Tand the imbalance ratio α , we generate 5 instances of demand data. For each instance of demand data, we have $3 \times 3 \times 3 = 27$ combinations of the parameter settings above, for a total of $3 \times 4 \times 5 \times 27 = 1620$ instances.

In Section 3.4.2, when assessing the sensitivity of the cross-shipping volume, we fix e at \$5 and the values of T and α at their baseline values of 10 and 1.5, respectively, and vary only the values of s, h, and r. Thus, the total number of instances considered for this second experiment is $5 \times 3 \times 3 \times 4 = 180$.



	11 0				
Parameter	Values	Parameter	Values	Parameter	Value
T	5, 10, 15	h	\$4, \$12, \$20	s_r	\$1000
α	1, 1.5, 2, 3	S	\$1500, \$2000, \$2500	h_c	\$1
r	\$2, \$3, \$4, \$5	e	\$5, \$6, \$7	s_c	\$200

Table 3.1. Parameter Settings to Quantify the Value of Local Reuse and the Sensitivity of the Volume of Cross-Shipping.

Table 3.1 summarizes the representative values of the parameters used in our computational study.

3.4.2 Results

CPLEX (version 11.1.1) was able to optimally solve the MIPs corresponding to the instances of RM in a reasonable amount of time. The instances of BM were solved using the dynamic programming formulation of Section 3.2.

The Value and Extent of Local Reuse

For the instances in our test bed, VoR ranges between 14.9% and 36.4%. Over all the 1620 instances, both the mean and median are around 26.6%. EoR ranges between 41.3% and 100%, with mean and median values of 82.2% and 90.5%, respectively. We, therefore, conclude that the saving for a DI from implementing fit-sorting and custodial inventory as well as the extent of local recirculation are substantial. Indeed, our results confirm the ability of the Fed's policy to generate a welfare-like situation. From the viewpoint of DIs, substantial reuse is beneficial under the Fed's policy since this is the optimal choice. In turn, the 3PLPs benefit from the new fit-sorting business opportunity. Finally, the Fed too furthers its goal of reducing the social cost of providing currency to the public.





Figure 3.2. Impact of Imbalance and Length of Planning Horizon on Value and Extent of Local Reuse.

The imbalance ratio, α , is a good metric to highlight the relative differences between the behavior of VoR and EoR. We have VoR at its highest when $\alpha = 1.5$ (Figure 3.2(a)). To understand this, consider a scenario where the used-cash deposit in a week is consistently about 1/75% = 1.33 times the fit-cash demand in the next week. Then, by fit-sorting all the used-cash deposit each week, the DI can satisfy its fit-cash demand and completely eliminate the transportation – to and from the Fed – of fit cash and used cash. This simple observation is quite helpful. When the per-week supply of used cash is about the same as the demand for fit cash (i.e., $\alpha = 1$), the DI cannot generate enough fit cash from local reuse to meet its demand. Recall that about 75% of the quantity fit-sorted is fit-cash. Thus, for $\alpha = 1$, EoR has an upper bound of around 75% (Figure 3.2(b)), which implies that a maximum of 75%of the fit-cash demand is satisfied by fit-sorting. Thus, it becomes necessary for the DI to withdraw fit cash from the Fed and, consequently, incur transportation cost. On the other hand when α is too high (say, $\alpha \in \{2,3\}$), although the DI can satisfy all of its fit-cash demand by local reuse (i.e., EoR \approx 1), there is too much used cash at hand. This excess used cash must be deposited to the Fed which, again, implies an increase in transportation cost. Thus, the DI incurs additional costs at extreme values of α . When $\alpha = 1.5$, the deposit



94

of used cash is just of the correct magnitude to enable the DI to generate the required fit cash by local reuse. Thus, in this latter case, the shipment activity (if any) between the DI and the Fed are much less frequent than that in the former extreme cases. To summarize, *VoR is at its highest when the imbalance between the used-cash deposit and fit-cash demand is just enough for the DI to effectively satisfy all demand via local reuse.*

Another observation from Figure 3.2(a) is that VoR increases with the length of planning horizon. This is intuitive, since a longer planning horizon implies more opportunities to combine orders and enable better decisions. Figure 3.2(b) also shows that even with enough used cash (i.e., $\alpha > 1.5$), the DI still does not fully use local reuse to satisfy its fit-cash demand (EoR ≈ 0.80). The reason is that a DI may prefer to fit-sort less when the perbundle per-period holding cost of fit cash (h) is low. Fit-sorting enables the DI to store fit cash in custodial inventory, which has a much lower holding cost than the regular inventory. However, to exploit this, the DI may have to fit-sort in advance and keep fit cash in the custodial inventory for multiple periods. On the other hand, fit cash ordered from the Fed needs to be kept in the regular inventory but for a shorter time (e.g., one period). When h is low, the cost of holding fit cash for a longer time in custodial inventory can exceed that of holding it for a shorter time in regular inventory. Thus, withdrawing fit cash from the Fed to satisfy fit-cash demand becomes preferable.

Influences on the Volume of Cross-Shipping

To understand the combination of forces that influence the volume of cross-shipping, we first examine the impact of the fit-sorting cost on the optimal solution. When the per-bundle



cost of fit-sorting increases, if the DI intends to maintain the same level of reuse, then it is immediate that the total fit-sorting cost will increase. Thus, one option to reduce the total cost is to reduce the fit-sorting volume over the planning horizon. This reduces one or both of the fixed and variable costs of fit-sorting. However, reducing the fit-sorting volume implies the following two consequences: (1) with less fit cash from fit-sorting, the DI has to withdraw more fit cash from the Fed to meet its demand and (2) the DI has more inventory of used cash which needs to be eventually deposited to the Fed. In turn, these outcomes can increase the total cost in two possible ways: (i) if the DI decides to deposit used cash and withdraw fit cash near-simultaneously (i.e., within the same week), then the cross-shipping cost and (possibly) the ordering cost will increase, (ii) if the DI can avoid cross-shipping by either postponing the deposit of used cash or withdrawing fit cash in advance, the holding cost will increase. Thus, the DI faces a healthy tradeoff from reducing the fit-sorting volume.

The previous discussion suggests that the combined influence of per-bundle fit-sorting cost (r) and the per-bundle holding cost (h) may be the force that decides the volume of crossshipping. Comparing the optimal solutions of RM over different values of r and h leads to two interesting observations: (1) when the per-bundle holding cost is very low, the volume of cross-shipping, the total holding cost as well as the number of orders increase with fit-sorting cost, while the volume of fit-sorting and the volume in custodial inventory decrease with fitsorting cost, (2) when the per-bundle holding cost is very high, RM retains the same optimal solution even when the fit-sorting cost varies. We now provide an intuitive explanation for these two observations: When r has its minimum value, we have $r \ll \min\{h, e\}$. Thus, fully exploiting fit-sorting to satisfy fit-cash demand is most economical for the DI. When



r increases, the DI may consider the option of reducing the fit-sorting volume, which, as discussed above, will increase either the holding cost or the cross-shipping cost. When the per-bundle holding cost (h) is very high, it still significantly dominates the increased r. Thus, increasing inventory cost becomes uneconomical. The other option – increasing crossshipping – leads to an increase that dominates the savings resulting from the reduction in the fixed and variable costs of fit-sorting. Consequently, the DI chooses to maintain the maximum level of fit-sorting, resulting in the same optimal solution. However, when the per-bundle holding cost is low, the DI can tolerate an increase in inventory costs. Therefore, as r increases, the progressively larger impact of reducing the fit-sorting cost (by reducing the fit-sorting volume) begins to dominate the combined increase in the ordering, holding, and cross-shipping costs. As a result, the DI chooses to fit-sort less and, instead, increase the volume of cross-shipping. To summarize, the impact of changing the fit-sorting cost on the volume of cross-shipping is governed by the ratio of the holding cost and the fit-sorting cost. As long as the holding cost continues to significantly dominate fit-sorting cost, there is little or no change in the volume of cross-shipping as the fit-sorting cost increases. When the holding cost is comparable to fit-sorting cost, an increase in the fit-sorting cost leads to a consistent increase in the volume of cross-shipping.

Figure 3.3 confirms our interpretation. The percentage of instances with zero cross-shipping volume for each value of r and h is illustrated in Figure 3.3(a). At a high holding cost, we have a complete absence of cross-shipping as the per-bundle fit-sorting cost varies over its range. However, when the holding cost is low, the percentage of instances with zero cross-shipping cost decreases with the fit-sorting cost. To further explore the latter case, we illustrate the





Figure 3.3. Impact of Per-Bundle Fit-Sorting Cost (r) and Per-Bundle Holding Cost (h) on Cross-Shipping.

average volume of cross-shipping when the holding cost is low in Figure 3.3(b). As can be clearly seen, the marginal increase in the average volume of cross-shipping increases with an increase in the fit-sorting cost.

Our discussion thus far assumes that the weekly values of used-cash deposit and fit-cash demand during the planning horizon can be forecasted with reasonable accuracy. To adapt the decisions from our models to a real-time scenario, we now develop a rolling-horizon procedure that allows for a limited extent of uncertainty in these two forecasts.

3.5 A Rolling-Horizon Procedure for Real-Time Decisions

In this section, we assume that the actual value of the deposit (resp., demand) in a week varies between a pair of lower and upper bounds (that depend on the week). These bounds can be forecasted at the beginning of the planning horizon and updated throughout the horizon (if necessary). Also, in practice, managers prefer a scheme which can incorporate information (e.g., realized demand) as and when it becomes available. Thus, a rolling-horizon procedure which offers an iterative decision-making environment is an ideal technique to adapt our



98

models for real-time decisions. Under this procedure, we first solve a τ -period problem, for some constant $\tau \geq 1$, by using the initial inventories and the forecasted demand/deposit data for Periods 1 through τ . Although the decisions for the first τ periods are then available, we only implement the decisions for the first period. After the actual demand/deposit for the first period are realized, we observe the inventories at the end of Period 1 and solve another τ period problem by using the (possibly updated) demand/deposit data for Periods 2 through $\tau + 1$. The iterative process continues in this manner and can be applied over an infinite horizon or a finite, *T*-period horizon. The τ -period problem encountered in each iteration is referred to as the rolling-horizon problem.

In this section, we first describe the construction of the rolling-horizon problem (Section 3.5.1) and then provide an iterative procedure that adapts solutions of this problem for real-time decisions (Section 3.5.2). Finally, we show – if the widths of the interval forecasts over the planning horizon are modest – that our procedure provides near-optimal, real-time solutions to arbitrary realized instances (Section 3.5.3).

3.5.1 A τ -Period Rolling-Horizon Problem Based on Forecasted Data

We retain the same basic notation as in Section 3.3.1 and only define the additional notation needed.

For the first period of a τ -period rolling horizon problem, if the realized fit-cash demand exceeds the projected demand, decisions that are based on the projected demand but are used for the realized demand may cause an infeasibility (due to lack of the required fit cash). One



Additional Notation:

 $\begin{aligned} \kappa^u_t, \bar{d}^u_t, \nu^u_t & \text{The forecasted lower bound, average, and upper bound of deposit of used cash (in bundles) from the DI's customers during Period t, t = 1, 2, ..., T. \\ \kappa^f_t, \bar{d}^f_t, \nu^f_t & \text{The forecasted lower bound, average, and upper bound of demand of fit cash (in bundles) for the DI's customers during Period t, t = 1, 2, ..., T. \\ \eta^u_t & \text{The actual deposit of used cash (in bundles) from the DI's customers during Period t, t = 1, 2, ..., T. \\ \eta^f_t & \text{The actual demand of fit cash (in bundles) for the DI's customers during Period t, t = 1, 2, ..., T. \end{aligned}$

simple way to avoid such an infeasibility is to develop a solution that uses the upper bound of fit-cash demand for the first period. Since we need to implement only the decisions for the first period, we can safely use the forecasted average fit-cash demands for the remaining $\tau - 1$ periods. Also, we use the forecasted average used-cash deposits for the τ periods. Thus, given forecasted used-cash deposits of the first τ periods, $(\kappa_t^u, \bar{d}_t^u, \nu_t^u), 1 \le t \le \tau$, and forecasted fit-cash demands of the first τ periods, $(\kappa_t^f, \bar{d}_t^f, \nu_t^f), 1 \le t \le \tau$, we generate a τ -period rolling-horizon problem using the following data: $d_t^u = \bar{d}_t^u, 1 \le t \le \tau$; $d_1^f = \nu_1^f$; $d_t^f = \bar{d}_t^f, 2 \le t \le \tau$.

For BM, given the initial inventories I_0^u and I_0^f of used and fit cash, respectively, and the deposit/demand data described above, we use the forward recursion described in Section 3.2.4 to solve the τ -period rolling-horizon problem. Similarly, for RM, given the initial inventories of used cash, fit cash, unfit cash, and custodial inventory (i.e., $I_0^u, I_0^f, I_0^n, I_0^c$, respectively), and the deposit/demand data, we use the Mixed-Integer Program described in Section 3.2.4 to solve the rolling-horizon problem.



3.5.2 A Rolling-Horizon Procedure

We now formally describe the rolling-horizon procedure for BM. The procedure for RM is similar, except that we need to calculate two more types of inventories $(I_t^n \text{ and } I_t^c)$ and record four more types of decision variables $(x_t^r, x_t^{c+}, x_t^n, \text{ and } x_t^{c-})$; see Section 3.3.1. Therefore, for brevity, we avoid an explicit description of the procedure for RM.

Procedure ROLLING HORIZON

Step 1. Let i = 1.

Step 2. Forecast the used-cash deposits and fit-cash demands for the next τ periods, i.e., $(\kappa_t^u, \bar{d}_t^u, \nu_t^u), (\kappa_t^f, \bar{d}_t^f, \nu_t^f), i \le t \le i + \tau - 1.$

Step 3. Generate a τ -period problem, P_b , using $\hat{I}_0^u = I_{i-1}^u$, $\hat{I}_0^f = I_{i-1}^f$; $\hat{d}_j^u = \bar{d}_{i+j-1}^u$, $1 \le j \le \tau$; $\hat{d}_1^f = \nu_i^f$; $\hat{d}_j^f = \bar{d}_{i+j-1}^f$, $2 \le j \le \tau$.

Step 4. Solve P_b , implement the decisions for the first period, i.e., \hat{x}_0^u, \hat{x}_0^f , and record $x_{i-1}^u = \hat{x}_0^u, x_{i-1}^f = \hat{x}_0^f$.

Step 5. The actual used-cash deposit of Period i, η_i^u , and the actual fit-cash demand of Period i, η_i^f are realized. Record η_i^u and η_i^f . Observe the ending inventories of used-cash and fit-cash for Period i, $I_i^u = I_{i-1}^u - x_{i-1}^u + \eta_i^u$, $I_i^f = I_{i-1}^f + x_{i-1}^f - \eta_i^f$.

Step 6. If i = T, then exit; otherwise, i = i + 1, go to Step 2.

At the end of the Period T, we have the realized deposit/demand for the entire T-period planning horizon, i.e., $\eta_i^u, \eta_i^f, 1 \leq i \leq T$. For comparison, we define P_a as the instance of BM corresponding to the realized deposit/demand data (i.e., under the assumption that all realized data is known with certainty a priori). By construction, the solution $(x_i^u, x_i^f), 0 \leq$ $i \leq T - 1$ generated by Procedure ROLLING HORIZON is a feasible solution to Problem P_a .



3.5.3 The Performance of Procedure ROLLING HORIZON

Except for the length of the planning horizon and the demands/deposits, we set the other parameters of BM/RM to their baseline values of the test bed described in Section 3.4.1. We set T to 100 as an approximation of a long-term horizon. For t = 1, 2, ..., T, the average demands, \bar{d}_t^f , of fit cash (in bundles) during Period t are randomly generated as before from U[1000, 2000]. Corresponding to $\alpha = 1.5$, the average deposits, \bar{d}_t^u , of used cash (in bundles) are randomly generated from U[1500, 3000]. To indicate the varying accuracies of the lower/upper bounds of the intervals in which the actual demands/deposits lie, we consider symmetric intervals with widths (w; see Table 3.2) of 5%, 10% and 15%, around the average demands/deposits. That is, we consider three intervals for the actual fit-cash demand η_t^f : $[0.975\bar{d}_t^f, 1.025\bar{d}_t^f]$, $[0.95\bar{d}_t^f, 1.05\bar{d}_t^f]$ and $[0.925\bar{d}_t^f, 1.075\bar{d}_t^f]$. Similarly, we consider three intervals for the actual used-cash deposit η_t^u : $[0.975\bar{d}_t^u, 1.025\bar{d}_t^u]$ $[0.95\bar{d}_t^u, 1.05\bar{d}_t^u]$ and $[0.925\bar{d}_t^u, 1.075\bar{d}_t^u]$. We consider three values of the length, τ , of the rolling horizon: 2, 5, and 10. Next, we generate 10 instances of the average used-cash deposits and fit-cash demands throughout the T-period horizon. Thus, we have 10 vectors $(\bar{d}_t^u, \bar{d}_t^f), t = 1, 2, \dots, T$. For each of these instances, we generate three instances of the demands/deposits intervals, corresponding to the widths of 5%, 10% and 15%. Thus, we have 30 vectors $(\kappa_t^u, \bar{d}_t^u, \nu_t^u, \kappa_t^f, \bar{d}_t^f, \nu_t^f), t = 1, 2, \dots, T$. Finally, for each instance of the demands/deposits intervals, we generate 5 realized instances by randomly choosing the actual demands/deposits within the corresponding intervals, for a total of $30 \times 5 = 150$ instances of the T-period realized problem, P_a .

For each instance of P_a , let \hat{X}^* denote its optimal solution and let \hat{X} denote the solution



obtained by Procedure ROLLING HORIZON described above. The percentage increase in cost of \hat{X} over \hat{X}^* , computed as $\frac{(\text{Objective Value of } \hat{X} - \text{Objective Value of } \hat{X}^*)}{\text{Objective Value of } \hat{X}^*} \times 100\%$, is an appropriate metric to capture the performance of Procedure ROLLING HORIZON. Each entry in Table 3.2 indicates the average of this gap over the corresponding 50 instances of BM corresponding to each value of w.

Table 3.2. Impact of the Length of the Rolling Horizon (τ) and Variation Width (w) on the Performance of Procedure ROLLING HORIZON.

$\tau \downarrow w \rightarrow$	5%	10%	15%
2	1.00%	2.00%	2.95%
5	0.76%	1.77%	2.75%
10	0.75%	1.77%	2.75%

For RM, we note that CPLEX (version 11.1.1) is not able to solve instances with T = 100 to optimality within a reasonable amount of time (2 hours of CPU time). Therefore, to compare the solutions obtained by our procedure, we used the best solution available from CPLEX. Considering the size of our test bed, we imposed a time limit of 20 minutes within CPLEX for each 100-period instance of RM. In most cases, the best solution available at the end of the imposed time limit is within 10% of the optimal. We then compare this solution with the one obtained by Procedure ROLLING HORIZON. Again, the average percentage gap between the two solutions is similar to that shown in Table 3.2.

Given that the optimal solutions to P_a are "ideal" – in the sense that they assume future data is known with certainty a priori – we observe that Procedure ROLLING HORIZON offers near-optimal, real-time solutions efficiently even in the presence of a considerable amount of variation (signified by a width of 15%). We conclude that practitioners can effectively use the two models (BM and RM), together with this procedure to plan the decisions of transportation, fit-sorting, and custodial inventory.



Since DIs need to manage cash inventory in the presence of the Fed's implemented policy, Problems BM and RM are indeed the real problems that DIs need to address and have, therefore, been the focus of our analysis. In Dawande et al. (2010), the authors argue that the Fed's policy may not induce DIs to behave in a socially-optimal manner. In the next section, we compare a DI's inventory decisions under the Fed's mechanism with those under a socially-optimal mechanism. From a DI's individual perspective, a socially-optimal mechanism may not be necessarily better than the Fed's mechanism. Thus, in addition to help us compare – from a DI's viewpoint – the relative benefits of the two mechanisms, such an analysis may also help the Fed fine-tune the parameters of its policy in the future.

3.6 The Fed's Policy and a Socially-Optimal Mechanism: A Comparative Analysis

The Fed's mechanism focuses on cross-shipping with the intention of lessening its fit-sorting expense and, consequently, the societal cost of providing cash to the public. Dawande et al. (2010) argue that instead of this indirect manner of reducing social cost, the Fed can directly focus on reducing its fit-sorting volume, which, in turn, is defined by the non-fitsorted used cash deposited to the Fed by the DIs. The authors propose a (hypothetical) socially-optimal mechanism that differs from the Fed's current policy in two respects: (1) the calculation of the volume of cross-shipping and (2) a reward for additional fit-sorting. The definition of cross-shipping is the same under both mechanisms. Under the Fed's policy, the volume of cross-shipping is calculated as the lesser of the fit cash in the used-cash deposit and the fit-cash withdrawal in the same week. In contrast, under the socially-optimal mechanism,



the volume of cross-shipping is the amount of used-cash deposit, if there is also a fit-cash withdrawal in the same week. Also, under the socially-optimal mechanism, the DI receives a reward from the Fed if it deposits extra fit-sorted cash to the Fed after satisfying its own demand of fit cash.

We let RM^s denote the variant (of RM) that minimizes a DI's total cost incurred in managing the inventory of cash under the socially-optimal mechanism of Dawande et al (2010). The formulation of RM^s as an MIP is similar to that of RM in Section 3.3.3; we, therefore, avoid specifying the entire formulation and only summarize the main differences. In addition to the parameters of RM, the formulation of RM^s requires a new parameter w, defined as the ratio between the reward for depositing one bundle of fit-sorted cash and the fit-sorting cost (r/ρ) incurred for generating one bundle of fit-sorted cash. Furthermore, objective function of RM^s is to minimize the *net* cost for a DI, which is the difference between the total costs incurred (see Section 3.3.1) and the total reward received, over the planning horizon. Except for the constraints that define the volume of cross-shipping and the reward, all the other constraints in RM^s are the same as in RM. As in the case with RM, CPLEX was able to optimally solve the MIPs corresponding to the instances of RM^s in a reasonable time.

To illustrate the differences between the Fed's policy and the socially-optimal mechanism, we define the **Relative Benefit of Social Plan** (RBoSP) as the percentage cost saving offered by RM^s over RM. RBoSP captures the relative monetary impact on a DI from operating under the socially-optimal mechanism instead of the Fed's current policy. That is,

 $RBoSP = \frac{(Objective Value of RM - Objective Value of RM^s) \times 100\%}{Objective Value of RM}$



To understand the behavior of the RBoSP, it is instructive to examine the two mechanisms under two extreme scenarios: one in which both the ratio w and the imbalance ratio α are high and the other in which both these ratios are (relatively) low. To generate an appropriate test bed, we fix the length of the horizon (T) at 10. Also, the per bundle cross-shipping fee (e) and the per bundle per period holding cost (h) are fixed at \$5 and \$6, respectively. The per bundle fit-sorting cost (r) takes one of four values: 2, 4, 6, 8. Except for w and α , we keep the remaining parameters at their baseline values described in Section 3.4.1. Different combinations of the pair (w, α) are considered in the two scenarios. For each value of α , by using the approach described in Section 3.4.1, we generate 10 vectors of fit-cash demands and used-cash deposits over the horizon. Thus, for each value of α , we have 10 instances each of RM and RM^s. Figure 3.4(a) corresponds to the scenario when w = 2 and $\alpha \in \{2.0, 2.1, 2.2\}$ while Figure 3.4(b) represents the scenario when w = 1 and $\alpha \in \{0.8, 0.9, 1.0\}$. Each point in Figure 3.4 represents the average value of RBoSP over the 10 instances that correspond to a specific choice of α and r.



Figure 3.4. The Relative Benefit of Social Plan under Two Scenarios.

As seen in Figure 3.4(a), the RBoSP is non-negative and increases with the fit-sorting cost (r). Thus, in the first scenario, the DI incurs a lower cost under the socially-optimal policy as



compared to that under the Fed's policy, and the gap between these two costs increases with the fit-sorting cost. This behavior can be explained using two properties: availability of used cash to fit-sort an additional amount and the reward for undertaking additional fit-sorting. Since the imbalance ratio $\alpha \geq 2$, the DI has the opportunity to fit-sort more, after satisfying its own demand of fit-cash. Also, since w = 2, fit-sorting extra used-cash is profitable for the DI. Consequently, relative to the Fed's mechanism, the DI fit-sorts more and cross-ships less under the socially-optimal mechanism, which, in turn, implies that it earns more reward and pays less penalty. Recall that w is the ratio of the reward for depositing one bundle of fit-sorted cash to the fit-sorting cost r/ρ incurred for generating one bundle of fit-sorted cash. Since w is fixed, the per-bundle reward increases with r. Thus, the relative benefit of operating under the socially-optimal mechanism progressively increases with r, reflecting in an increase in RBoSP.

In contrast, the second scenario results in Figure 3.4(b), where the RBoSP is non-positive and further decreases with r. Thus, the DI is worse off under the socially-optimal mechanism. We discuss the reward first. Since $\alpha \leq 1$, the DI cannot satisfy all of its fit-cash demand by fit-sorting its used-cash deposit. Also, w = 1 implies that the reward from depositing fit-sorted cash to the Fed is only enough to cover the fit-sorting expense incurred to generate it. Thus, the DI cannot derive significant benefit from the reward component of the sociallyoptimal mechanism. Next, we examine the difference in the cross-shipping penalty between the two mechanisms. When r is low, under both the mechanisms, the DI chooses to fit-sort most of the used-cash to avoid cross-shipping. In turn, the absence of any cross-shipping implies that there is no or little difference between the objective values of RM and RM^s.



When r increases, the DI prefers to fit-sort less in RM. Consequently, the DI has more used-cash inventory and needs to withdraw more fit-cash from the Fed. Thus, in general, the volume of cross-shipping increases. Note, however, that for the same solution, the DI is charged more cross-shipping penalty under the socially-optimal mechanism than under the Fed's policy. Thus, when r increases, the objective value of RM^s increases faster than that of RM; consequently, RBoSP decreases with r.

To summarize, the DI can be either better or worse off under the socially-optimal mechanism than under the Fed's policy. There are two main determinants of the sign of RBoSP: the imbalance ratio (α) and the reward ratio (w). In general, under the socially-optimal mechanism, when α and w are both large, the DI benefits from the reward component and is able to reduce its net cost to a value lower than that under the Fed's mechanism. When α and w are both small, the DI incurs a heavier penalty as compared to the Fed's mechanism and, thus, incurs a relatively higher net cost. In both cases, the percentage gap between the two mechanisms enlarges with an increase in r.

3.7 Conclusions

The Federal Reserve's 2007 recirculation policy for physical cash is designed to incentivize DIs to locally reuse cash by fit-sorting their used-cash deposits. In this paper, we address the management of the inventories of different types of cash – used, fit, and unfit – over a finite planning horizon for a medium-size DI. The Fed's policy poses an important choice for a DI: whether or not to undertake the fit-sorting activity. Our first model (BM) captures the scenario in which a DI chooses not to fit-sort used cash. We identify two characteristics



of an optimal solution regarding the withdrawals of fit cash and deposits of used cash. By exploiting these structural properties, we develop a polynomial-time dynamic programming algorithm. Our second model (RM) captures the situation when a DI undertakes fit-sorting (and is therefore also able to exploit custodial inventory, which is another component of the Fed's policy). We establish the hardness of this model and develop an integer programming formulation. Through a comprehensive numerical study, we conclude that the saving for a DI from implementing fit-sorting and custodial inventory as well as the extent of local recirculation are substantial. Thus, the Fed's eventual goal of reducing the social cost of providing currency to the public can be furthered. Moreover, the saving reaches its highest value when the imbalance between the used-cash deposit and fit-cash demand is just enough to effectively satisfy all demand via local reuse. To adapt our models for real-time decisions, we develop a rolling-horizon procedure which operates in an iterative decision-making environment. We show that our procedure provides near-optimal, real-time solutions to arbitrary realized instances when the widths of the interval forecasts over the planning horizon are modest. We also show that a DI can be either better or worse off under the Fed's implemented policy as compared to a (hypothetical) socially-optimal mechanism.

Limitations and Directions for Future Work

The two models – BM and RM – studied in this paper assume that data is known with certainty. While this simplifying assumption seems to be reasonable based on our interactions with practitioners, uncertainty may arise from three sources: demand of fit cash, deposit of used cash, and the proportion of fit cash in used-cash deposit. The rolling-horizon procedure in Section 3.5 performs well under the assumption that the widths of the interval forecasts



over the planning horizon are modest. This can be seen as an effort to address – to a limited extent – the uncertainty from the first two sources. Nevertheless, the applicability of our models would benefit from a stochastic analysis in the presence of these uncertainties.

If cross-shipping cost is ignored, then the stochastic model corresponding to BM is equivalent to the problem of managing the inventories of two products experiencing stochastic demand and having a joint set-up cost. A limited set of results for this problem are available in the literature; see, e.g., Johnson (1967), Wheeler (1968), Kalin (1980), Liu and Esogbue (1999), Gallego and Sethi (2005). However, a characterization of the optimal policy under a general setting has, thus far, remained elusive. A promising but challenging future direction is to completely characterize the optimal policy for this problem. The main challenge here is that there is a fixed cost shared by two products. Thus, this direction is likely to warrant an advancement in the theory of multidimensional K-convexity. When cross-shipping cost is considered, the stochastic problems corresponding to both BM and RM involve complicated interactions between fit-cash withdrawals, used-cash deposits, and unfit-cash deposits. To our knowledge, such models have not been addressed in the literature. While it is unlikely that simple optimal policies exist for these problems, the task of obtaining near-optimal policies which can be efficiently implemented is a challenging and useful direction for future work.

It is instructive to establish a broader connection for the setting of our analysis in this paper. Conceptually, the need for local reuse in the cash supply chain is similar to that in the supply chain of false-failure returns (see, e.g., Ferguson et al. 2006). False-failure returns refer to those goods that are returned to the retailer (by customers) and, subsequently, to the



manufacturer (by the retailer) but have no functional or cosmetic defects. Like used cash, a significant fraction of which is actually reusable, the manufacturer may find it economical to incentivize retailers to locally identify and recirculate faulty returns. In turn, the retailer may hire the services of a third-party provider to accomplish this task. Such a setting provides an opportunity to expand the analysis to a genuine multi-product domain.



CHAPTER 4

PROCESS INNOVATION VIA AN INDUSTRIAL SYMBIOTIC SYSTEM: THE IMPACT OF COMPETITION ON THE WILLINGNESS TO IMPLEMENT

4.1 Synopsis

The principle of industrial ecology (Allenby, 1992; Jellinski et al., 1992; Ehrenfeld, 1995) proposes to reorganize the industrial system so that it evolves towards a mode of operation that is compatible with the biosphere and is sustainable over the long term. Industrial ecology suggests the idea of an industrial food chain in which companies can be linked in some form of network, in order to exploit unutilized resources or by-products and thereby increase resource utilization. The two main elements under this concept are (1) the need to optimize the use of materials and energy, and (2) to close material loops while minimizing emissions.

Industrial Symbiosis, a subfield of industrial ecology, can be defined as engaging "traditionally separate industries in a collective approach to competitive advantage involving physical exchange of material, energy, water, and by-products" (Chertow, 2000). The term *By-Product Synergy* (BPS) has also been used synonymously in the literature with industrial symbiosis. BPS can offer true business opportunities beyond cost reduction, if wastes are viewed as raw materials for other industries. As BPS networks develop, industry goals may shift from reducing waste generation towards producing near-zero waste and finally to producing 100% product, while emissions are lowered and energy use is minimized (Mangan and



Olivetti 2010). The economic activity created in a BPS network creates new businesses and jobs, where the premise is that turning waste output from one organization into a product stream for another organization can generate revenue while reducing emissions and the need for virgin materials.

To focus on the operational issues involved in the implementation of an industrial symbiotic system, it is beneficial to first discuss a specific example.

4.1.1 A Real-World Implementation of a Symbiotic System

Seshasayee Paper and Boards (SPB; http://www.spbltd.com) was set up as a public enterprise in 1961 in the southern state of Tamil Nadu, India (see Private Communications listed at the end of the chapter). The company's main products include Bristol boards, Manila boards, colored bank paper, colored poster paper, and writing paper. The paper industry is capital intensive and highly dependent on easy access to high-quality raw materials. In particular, the paper industry in India has traditionally faced considerable difficulty with the availability of raw materials. The continued efforts by the Government of India to minimize de-forestation have progressively increased the scarcity of wood. Consequently, paper manufacturers have been consistently forced to look for alternate sources of raw materials. One such alternative for the paper production process is *bagasse*, a fibrous mass remaining after the extraction process of juice from sugarcane. The use of bagasse in paper production is energy efficient and also has a lesser impact on the environment, relative to wood. Traditionally, bagasse has been used as an important fuel input in the sugar industry and can meet the requirement of fuel for the industry. Since alternative fuels such as coal or



furnace oil are relatively costlier, the sugar industry has been reluctant to sell the bagasse to the paper industry. Not surprisingly, following the steep increase in the prices of furnace oil and coal over the years, SPB was unable to make any arrangement to obtain a regular supply of bagasse from the sugar industry. The company also faces several environmental challenges. In addition to the emission of non-condensable gases, a paper mill also releases chlorinated compounds, dioxins, and furans. The waste water (effluent) carries high levels of Biochemical Oxygen Demand, Chemical Oxygen Demand, and other suspended solids. Furthermore, the problem of solid waste disposal is also a major concern in the face of local environmental pressures.



Figure 4.1. A Pictorial Illustration of the Paper-Sugar Symbiotic Industrial Complex.



SPB responded to these challenges by creating a revolutionary industrial complex. The company got involved in sugar production by locating a sugar mill (Ponni Sugars) close to its paper production facility, so that the entire output of bagasse from the production of sugar can be used as an input for the production of paper. This, however, led to another hurdle – availability of sugarcane that is a key ingredient for sugar production. This was a difficult problem as there was very little cultivation of sugarcane in the immediate neighborhood due to poor availability of water. The land around the nearby river Cauvery was elevated and dry, and there was a lack of facilities to pump water from the river (20 feet lower) to the surrounding areas. In 1991, the company entered into a tripartite agreement with the local farmers and its sugar mill, and decided to irrigate about 1500 acres of dry land with SPB's treated effluent for the cultivation of sugarcane. The sugar mill, in turn, purchases the sugarcane from these farmers and supplies its by-product bagasse to the paper mill of SPB. This "symbiotic triangle" is illustrated in Figure 4.1.

4.1.2 Differences from Traditional Process Innovation

There are several features of this implementation of the paper-sugar industrial complex that distinguish it from a traditional process innovation. We discuss a few below. Later, in Sections 4.2 and 4.3, our models exploit these properties.

1. Simultaneous Change in the Production Processes of Multiple Products: It is important to note the fundamental, and more environment-friendly, change in the use of bagasse in the symbiotic system. While bagasse has been used by sugar mills as a source of fuel, its use in paper production (in the symbiotic system) is as a core raw material. This



affects the production processes of both sugar and paper. On the one hand, the sugar mill needs to source alternate fuel since bagasse is no longer available. On the other hand, the production process of paper needs to be appropriately altered to use bagasse instead of wood. Also, in the larger symbiotic system, there are other implications: (1) sugar production benefits from the use of reliable and relatively cheap supply of sugarcane from the local farmers, (2) the effluent from paper production needs to be treated and delivered to the farmers, which is an additional expense for the paper mill, and (3) after implementation, the need to treat the waste from the productions of both paper and sugar reduces significantly relative to that before implementation. Clearly, these changes together imply advantages as well as disadvantages for both production processes.

- 2. A Common Fixed Cost: Since the process changes are symbiotic, the fixed investments required for the implementation naturally constitute one common cost for the entire implementation. This common fixed cost includes (i) acquiring the capability for the large-scale processing of bagasse in paper production, (ii) acquiring the capability to appropriately treat the effluent from the production processes, so that it can be used by sugarcane farmers, and (iii) building the transportation infrastructure for distributing the treated effluent to the sugarcane farmers.
- 3. Linked Production Costs and Production Quantities: Using the output (bagasse) of the sugar mill as raw material reduces the procurement cost of paper production. Therefore, the production quantities and marginal costs of the two products get interconnected. While the production quantities of paper and sugar do not necessarily



depend on each other (since, if required, the paper mill can obtain wood and the sugar mill can get additional sugarcane from other sources), their reciprocal influences on the cost economies in the symbiotic system are clear.

Finally, it should be emphasized that implementing the symbiotic system is a decision that implies accepting the (different) impacts on both the products together. In other words, it is not possible to change the production process of only one product while leaving the other unaffected. It is also clear from the discussion above that the cost-benefit tradeoff for the symbiotic system as a whole is not straightforward.

There are several other implementations of symbiotic systems that have been reported in the literature. We briefly mention three examples.

- Londonderry Eco-Industrial Park, New Hampshire (Block, 1998): The 100-acre park near the New Hampshire airport is one of the nation's prime examples of eco-industrial synergies. A plastics recycling company (a tenant at the park) purchases waste plastic from Stonyfield Farms Yogurt (a firm that is located close to the park). AES Corporation (Puerto Rico) commissioned and built a 720 MW combined natural gas power plant on site that uses treated waste water, which is pumped from the city of Manchester's sewage treatment facility, in its cooling towers.
- Industrial Symbiosis Park, Kalundborg, Denmark (Ehrenfeld and Gertler 1997): The park consists of a web of material and energy exchanges that occur amongst several diverse companies and the local community. There are five core partners – a power station, an oil refinery, a gypsum board facility, a pharmaceutical plant, and the city



of Kalundborg – who have all developed a series of bilateral exchanges. The partners share ground water, surface water, waste water, steam, electricity, and also exchange a variety of residues that become feedstock for other processes.

• Monfort Boys Town Integrated BioSystem, Suva, Fiji (Klee and Williams, 1999; Bequette, 1997; Kane, 1997; Klee, 1999): This symbiotic system mixes agriculture and industry. Here, the waste from a brewery (spent grain) is used as a substrate to grow mushrooms, which are sold in the marketplace. The mushrooms break down the waste, making it a high-value, edible pig feed. The pigs are sold in the market, and the waste generated from the pigs is processed through an anaerobic digester. The treated waste is then piped to local fishponds, where the nutrient-rich water spawns food for several tropical layers of fish. The waste also creates fertile soil for growing vegetables.

4.1.3 Our Goals and Contributions

Our discussion thus far raises a variety of important research issues concerning symbiotic systems. This paper focuses on the following:

• Characterization of the Impact of a Symbiotic System: The differences with traditional process innovation (Section 4.1.2) imply non-trivial cost-side (operational) tradeoffs of implementing a symbiotic system. The implementation provides a firm an option of offering "green" (i.e., environment-friendly) variants of its products. Accordingly, the demand-side impact (to be soon discussed in Section 4.2.3) depends on the nature of the consumer population. Two other factors affecting the demand-side influence are the presence of a competing firm and the nature of the products offered



by the competition. Motivated by the real-world system discussed in Section 4.1.1, we analyze a firm's production decisions for its two products – both in the presence and absence of a symbiotic system – under monopoly as well as under competition in Section 4.3.

- Understanding the "Willingness" to Implement a Symbiotic System: For a given setting (supply- and demand-side parameters; nature of competition, if any), the difference in a firm's total profits (from all the products connected by a symbiotic system) before and after the implementation of the system is a good measure to capture its willingness to implement the system. Section 4.4 exploits the corresponding optimal/equilibrium solutions to examine the behavior of this metric both under monopoly and competition, with respect to changes in several important parameters, including the variable production cost after implementation, consumers' appreciation of the green variant, and the proportion of green consumers.
- Increased Willingness under Competition: SPB's decision to implement the symbiotic system (Section 4.1.1) in 1991 came at an interesting time. Around the same time, the Government of India initiated the process of economic liberalization, which opened the country's markets to foreign firms in several sectors, including paper and pulp. This coincidence raises two interesting questions:
 - 1. Can the presence of competition increase the firm's willingness (relative to that under a monopoly) to implement the symbiotic system? We address this issue in Section 4.4.3 and identify some situations under which the answer is in the affirmative.



2. The firm can face competition either for its regular variant only or for both the regular and green variants. Can the willingness in the former situation be higher than that in the latter? Some conditions for an affirmative answer are identified in Section 4.4.3.

• Benefiting Both the Firm and the Consumers:

While the benefits of a symbiotic system for the society at large are clear, a stronger motivation for the firm to implement the system results when both the firm and its consumers benefit. In Section 4.4.4, we identify some conditions under which such a simultaneous improvement occurs. We also illustrate a stronger case where a firm's willingness to implement is negative in a monopoly, but positive under competition and consumer welfare too increases after the implementation.

Technically, the challenges in our analysis are twofold: Under monopoly, we derive a firm's optimal prices of two variants (regular and green) in the presence of three types of consumers (regular, flexible green, and dedicated green) who each have different valuations for these two variants. It is important to emphasize that although the flexible green consumers have higher valuation for the green variant, they may choose to purchase the regular variant in case the price of the green variant is too high relative to that of the regular variant. Under Cournot competition, we derive the equilibrium production quantities – and the corresponding prices – of two firms for both the variants, under variants vs. both variants).



4.1.4 Literature Review

Three broad streams of literature are relevant to our paper – Closed-Loop Supply Chain Management, Industrial Ecology and Symbiotic Systems, and Process Innovation.

• Closed-Loop Supply Chain Management: The primary focus of the closed-loop supply chain literature has been on taking back products from consumers and recovering added value by using all or part of the products. Here, product recovery is addressed at the end of products' use or end of life with an emphasis on preventing them from entering the waste stream by using materials recovery systems and then adding value. Fleischmann et al. (1997), Guide et al. (2000), Guide and Jayaraman (2001) and Guide et al. (2003), all offer comprehensive reviews of reverse logistics and closed-loop supply chain research, including remanufacturing, recycling, reselling, and disposal. Most of these studies address a combination of cost-efficient recovery processes, while meeting prescribed environmental standards. Researchers including Toktay et al. (2000), Savaskan et al. (2004), Ferrer and Swaminathan (2006), and Geyer et al. (2007), study the implications of remanufacturing on supply chain logistics, operations, and product design, while Majumder and Groenevelt (2001) and Ferguson and Toktay (2006) study remanufacturing in a competitive setting.

Recently, researchers have focused on the management of waste streams and the impact of regulation on both product and waste dispositions. For example, Subramanian et al. (2008) study the impact of extended producer responsibility for new product introductions in the electronics industry, while Plambeck and Wang (2009) investigate



product design of durable goods in a sustainable supply chain environment. Additionally, Atasu et al. (2009), Esenduran and Kermahlioglu-Ziya (2009) and Esenduran et al. (2009) all study the operational impact of product take-back legislation. Guide and Van Wassenhove (2009) recount the evolution of research in the closed-loop supply chain area and discuss its development as a recognized subfield in supply chain operations.

The existing work on closed-loop supply chains primarily addresses post-consumer waste, where the source of the waste is the consumer. In contrast, our paper considers the simultaneous utilization of pre-consumer wastes from multiple products. We examine a setting where the waste from the manufacture of one product is used as a raw material for a second product, and vice-versa. In our setting, the firm proactively modifies its production processes to accomplish this goal.

• Environmental Issues, Industrial Ecology, and Symbiotic Systems: Environmental issues form an integral part of the broad framework of sustainability. As managers become increasingly aware of the long-term business implications of sustainability, organizations have moved beyond the consideration of whether or not it pays to be green (King and Lennox 2002; Walley and Whitehead 1994). The focus has now shifted on how to address environmental challenges, while maintaining competitiveness (Kleindorfer et al. 2005; Corbett and Klassen 2006). Studies that aim at integrating sustainability into environmental and economic systems include Allenby (1992), Jelinski et al. (1992), Allen and Behmanish (1994), Ehrenfeld and Gertler (1997), Korhonen (2004), and Wang et al. (2010). Researchers including Daly (1991) and Ehrenfeld



(1995) suggest simple design principles, including closing material loops, dematerialization, toxic elimination, and pollution prevention, that can serve as emerging models for operationalizing industrial ecology. Ehrenfeld and Gertler (1997) highlight industrial symbiosis via the Kalundborg case in Denmark, while Schwarz and Steininger (1997) and Posch et al. (1998) discuss the industrial recycling network at Styria. In both these cases, the key drivers are the potential cost reduction that is accomplished from waste avoidance and savings in virgin raw material. Lee (2011) studies a firm's operating strategies when it implements a by-product synergy on two products that are generated by a single production process.

In this paper, our attempt has been to characterize both the cost-side and demand-side implications of implementing an industrial symbiotic system. In particular, we show that these considerations generate rich tradeoffs and provide valuable insights on the decision to implement the environment-friendly initiative.

• Process Innovation: One consequence of a symbiotic system is the potential reduction in the production cost of one or more products that are manufactured in the system. In this sense, our work is related to the literature on "cost-reducing process innovations." Arrow (1962) considers the case of a firm undertaking a cost-reducing investment that cannot be imitated by competitors. He shows that the gain from a cost-reducing innovation is higher for a firm in a perfectly competitive industry than that for a monopolist. Bonanno and Haworth (1998) consider a vertically-differentiated industry to examine the question of whether cost-reducing innovations are more likely to be observed in regimes of more intense or less intense competition. Rosenkranz



(2003) studies the optimal division between product and process innovation under competition, when firms have an incentive to invest R&D activities in both. Martinez-Ros and Labeaga (2009) conclude that firms with a capability to engage in process innovation can be expected to generate higher profits, relative to those possible for non-innovators. Skea (1995) describes a range of process innovation strategies that organizations can adopt – from installing an end-of-pipeline technology (such as building a water treatment facility) to waste and emission-reduction strategies (such as process improvements through supply-chain redesign).

In our paper, a symbiotic system can be regarded as an unconventional approach for achieving by-product synergy, where multiple, simultaneous process innovations are implemented on the products linked by the system. Our analysis exploits this fundamental difference from traditional process innovation to offer insights on an organization's willingness to implement the system.

We now proceed to describe the setting of our analysis.

4.2 Model Setting

To address the research issues raised in Section 4.1.3, we develop several models to capture the operations of a domestic firm (Firm 1) that produces two products, P and S, under various settings. These models represent the scenarios both before and after the implementation of an industrial symbiotic system, and will be described in detail in Section 4.3. As justified later in Section 4.2.3, under an assumption that typically holds in practice, the firm's operational



decisions for the two products can be made separately. Thus, for the purpose of analyzing the firm's optimal decisions (under monopoly) and equilibrium decisions (under competition), a single-product model suffices here. Subsequently, in Section 4.4, when evaluating the firm's strategic decision of whether or not to implement the symbiotic system, we consider the combined impact on the two products.

This section is organized as follows: in Section 4.2.1, we introduce the products and consumers studied in our models. Next, the structure of demand is described in Section 4.2.2. The cost-side and demand-side impacts of implementing the symbiotic system are formalized in Section 4.2.3.

4.2.1 Types of Product Variants and Consumers

There are two possible variants of the product: *regular* and *green*. If a firm does not implement the symbiotic system, its products can only be labeled as regular. Otherwise, if the system is implemented, then the firm has the option of labeling its products as green, which represents the use of an environment-friendly production process. Note that, in the latter case, the firm could still label its products as regular. We now introduce the primary notation used in our analysis. Additional notation is introduced later, as required.

Several surveys (e.g., Irland 1993, Forsyth et al. 1999) have indicated the existence of a significant proportion of environment-conscious consumers who are willing to pay a premium for variants produced by an ecologically friendly process. Wustenhagen et al. (2003) cite the example of Swiss utility companies that are able to offer "green" electricity by charging



Notation:

- α The upper bound of consumer's valuation of one unit of the regular variant.
- a The appreciation of the green variant, defined by the ratio of a green consumer's valuation for a unit of the green variant to that for a unit of the regular variant, $a \ge 1$.
- p_r The market price of the regular variant.
- p_g The market price of the green variant.
- v_r The consumer's valuation of the regular variant.
- v_q The consumer's valuation of the green variant.
- Q_r The total demand of the regular variant.
- Q_g The total demand of the green variant.

their customers premiums of about 400-700% over the regular product. Sedjo and Swallow (2002) model the premium for a green variant as a percentage over customers' valuation of the regular variant. We consider three types of consumers: *regular*, *flexible green* and *dedicated green*. On the one extreme, we have regular consumers who do not have any special appreciation for the "green" label. Accordingly, the valuation of these consumers for the green variant is the same as that for the regular variant. On the other extreme, dedicated green consumers are devoted to the green variant, if it is available. Thus, the valuation of dedicated green consumers for the regular variant becomes 0 if the green variant. Flexible green consumers have a relatively higher valuation for the green variant. Flexible green variant, relative to the regular variant. These consumers buy the regular variant, if it results in a higher utility. The precise valuations of these three types of consumers are defined in Table 4.1.

We assume that an individual consumer's valuation for a unit of regular variant is uniformly distributed between 0 and an upper bound α , with density 1¹. At any point $v, 0 \leq v \leq$

¹Note that there is no technical difficulty in assuming the density to be an arbitrary positive



Table 4.1. Valuations of the Three Types of Consumers for Regular and Green Variants, for an Arbitrary Value of $v, 0 \le v \le \alpha$.

Consumer's Type \downarrow Valuation \rightarrow	Regular Variant, Green Variant (v_r, v_g)		
Regular	(v, v)		
Flexible Green	(v, av)		
Dedicated Green	(v, -), if the green variant is unavailable		
Dedicated Green	(0, av), if the green variant is available		

 α , there stands one customer. With probability θ_1 (resp., θ_2 and θ_3), the consumer is a regular consumer (resp., a flexible green consumer and a dedicated green consumer). The distribution of consumers' valuation is shown in Figure 4.2.



Consumers valuation for the Regular variant (V)

Figure 4.2. Distribution of Consumers' Valuation for the Regular Variant.

4.2.2 Structure of Demand

To obtain the aggregate demands of the regular variant and the green variant, we need to derive the demand of each type (regular/green) of variant from each type (regular/flexible green/dedicated green) of consumer. We now discuss these demands under three scenarios.

number, say n, other than 1. In this case, however, the value of n plays a role in the calculation of the firm's profit.


1. Only the regular variant is available.

All the consumers who have a valuation higher than or equal to the market price will purchase the variant. Thus, we have

$$Q_r = \alpha - p_r.$$

2. Both regular and green variants are available.

There are four possible price settings: $0 < p_g \leq p_r$, $p_r \leq p_g \leq ap_r$, $ap_r \leq p_g \leq (a-1)\alpha + p_r$, and $(a-1)\alpha + p_r \leq p_g \leq a\alpha$. The choices of each type of consumer and the total market demand for each variant under these four price settings are summarized in the following table. For brevity, we avoid providing the derivations of the expressions in the table.

Table 4.2. The Market Demand when Both Regular and Green Variants are Available.

	\rightarrow Price Setting	$0 < p_a < p_r$	$p_r \leq p_g$	$ap_r \leq p_g \leq$	$(a-1)\alpha + p_r$	
	$\downarrow Demand$		$\leq ap_r$	$(a-1)\alpha + p_r$	$\leq p_g \leq a\alpha$	
Type(s) of	Regular Consumers	Green	Regular	Regular	Regular	
Variants	Flexible Green	Croon	Crear Crear		Doculor	
Chosen	Consumers	Green	Green	Regular	negular	
By	Dedicated Green	Croon	Croon	Croop	Croon	
Consumers	Consumers	Green	Green	Green	Green	
Aggregate	Total Demand of the	0	$(\alpha, \beta)\theta$	$(\alpha - p_r)\theta_1 +$	$(\alpha - p_r)$	
Ayyreguie	Regular Variant (Q_r)	0	$(\alpha - p_r)v_1$	$\frac{p_g - ap_r}{a - 1} \theta_2$	$(\theta_1 + \theta_2)$	
Demand	Total Demand of the	$(\alpha - p_g)\theta_1 +$	$\left(\alpha - \frac{p_g}{a}\right)$	$(\alpha - \frac{p_g}{a})\theta_3 +$	$(\alpha - \frac{p_g}{p_g})\theta_{\alpha}$	
Demana	Green Variant (Q_g)	$\left(\alpha - \frac{p_g}{a}\right)(1 - \theta_1)$	$(1- heta_1)$	$\left(\alpha - \frac{p_g - p_r}{a - 1}\right)\theta_2$	(a a) 03	

3. Only the green variant is available.

In this case, all the three types of consumers either buy the green variant or buy nothing. Thus,

$$Q_g = (\alpha - p_g)\theta_1 + (\alpha - \frac{p_g}{a})(1 - \theta_1).$$



4.2.3 Cost-Side and Demand-Side Impacts of Implementing a Symbiotic System

Like traditional process innovation, the implementation of the symbiotic system affects a firm's production cost. Furthermore, it is clear from the discussion above that implementing the system also influences the market demand of a firm's variants. To describe these two impacts, we need some additional notation. For brevity, we only define the notation for Product P. The corresponding notation for Product S will then be clear.

Notation:

π_1	The profit of Firm 1 before the implementation of the symbiotic system.
π_1^{\prime}	The profit of Firm 1 after the implementation of the symbiotic system.
K	The fixed cost of implementing the symbiotic system.
p_p	The market price of Product P.
$q_{1,p}$	Firm 1's production quantity of Product P.
$b_1 q_{1,p}^2$	The waste treatment cost before the implementation of the symbiotic system.
$C_{1,p}(q_{1,p})$	The total production cost of Firm 1 to produce $q_{1,p}$ units of Product P prior
	to the implementation of the symbiotic system: $C_{1,p}(q_{1,p}) = b_1 q_{1,p}^2 + c_{1,p} q_{1,p}$,
	where b_1 and $c_{1,p}$ are positive constants, $b_1 \ll c_{1,p}$.
$b_{1}^{'}q_{1,p}$	The waste treatment cost after the implementation of the symbiotic system.
$C'_{1,p}(q_{1,p})$	The total production cost of Firm 1 to produce $q_{1,p}$ units of Product P after
	the implementation of the symbiotic system: $C'_{1,p}(q_{1,p}) = c'_{1,p}q_{1,p}$, where $c'_{1,p}$ is
	a positive constant whose value will be defined later.
β_p	The amount of waste generated from the production of one unit of Product P.
γ_p	The amount of waste of Product P needed to produce one unit of Product S.
s_p	The cost saving for using one unit of the waste of Product P to produce
	Product S.
\bar{s}_p	The revenue from selling one unit of the additional waste of Product P in the

• The Cost-Side Impact

open market.

Before the implementation of a symbiotic system, Firm 1's total profit is:

$$\pi_1 = p_p q_{1,p} - b_1 q_{1,p}^2 - c_{1,p} q_{1,p} + p_s q_{1,s} - b_1 q_{1,s}^2 - c_{1,s} q_{1,s}.$$

The implementation of a symbiotic system affects the cost of production in three ways:



1. The additional fixed cost of implementing the symbiotic system

The firm incurs an additional fixed cost K.

2. The savings in waste treatment cost

We assume that the implementation simplifies the treatment process of the wastes of both the products, by reducing the need to extensively treat them. Thus, the waste treatment cost – which is represented by the quadratic term $(b_1q_{1,p}^2)$ in the original production cost function – now has a linear form $(b'_1q_{1,p})$.

3. The savings in procurement cost of raw material

We assume that β_p units of waste are generated from the production of one unit of Product P. One unit of Product S can be produced by using γ_p units of this waste. The firm saves an amount of s_p for using every unit of the waste of Product P to produce Product S. Thus, the firm saves $(s_p \min\{\beta_p q_{1,p}, \gamma_p q_{1,s}\})$ from using the waste of Product P to produce Product S. If any waste of Product P remains after satisfying the required production of Product S, then we assume that the firm can sell it in the open market at a per-unit price \bar{s}_p ; refer to Figure 4.3.

Then, Firm 1's total profit after the implementation of the system is:

$$\pi_{1}^{'} = p_{p}q_{1,p} - b_{1}^{'}q_{1,p} - c_{1,p}q_{1,p} + s_{p}\min\{\beta_{p}q_{1,p}, \gamma_{p}q_{1,s}\} + \bar{s}_{p}[\beta_{p}q_{1,p} - \gamma_{p}q_{1,s}]^{+} + p_{s}q_{1,s} - b_{1}^{'}q_{1,s} - c_{1,s}q_{1,s} + s_{s}\min\{\beta_{s}q_{1,s}, \gamma_{s}q_{1,p}\} + \bar{s}_{s}[\beta_{s}q_{1,s} - \gamma_{s}q_{1,p}]^{+} - K.$$

Our analysis in this paper assumes that the difference between (i) the cost saving achieved by using one unit of the waste of Product P (resp., S) to produce Product S





Figure 4.3. The Mutually Beneficial Use of Waste in a Symbiotic System. The Figure Shows the Production Cycle of Products P and S. The Notation is as Defined at the Start of Section 4.2.3.

(resp., P) and (ii) the revenue from selling one unit of the additional waste of Product P (resp., S), is negligible. We offer two arguments to justify this assumption in our context. First, in practical implementations, it is often the case that the waste generated from one product does not contribute a *dominant* supply of raw material required for the production of the other product. For instance, our extensive discussions with the senior administrators and operations managers at SPB (see Private Communications listed at the end of the paper) revealed that while a substantial amount of bagasse needed for paper comes from the sugar mill, it is far from sufficient to meet demand. Therefore, the company sources additional bagasse and wood from both domestic and international sources. Thus, all the waste of one product is used in the production of the other product, and vice-versa. Consequently, there is no additional waste of these products to be sold in the open market. This is a natural outcome in an "ideal"



symbiotic system, since the useful utilization of all wastes is a cornerstone of the concept. So, in this case, we do not need the assumption. Second, even if additional waste remains, its quantity is typically small enough. Thus, there is sufficient demand in the open market for the firm to realize a per-unit revenue similar to the savings achieved if the waste were to be used for its own production. Under this assumption, Firm 1's total profit after implementation can be rewritten as follows:

$$\pi_1' = p_p q_{1,p} - b_1' q_{1,p} - c_{1,p} q_{1,p} + s_p \beta_p q_{1,p} + p_s q_{1,s} - b_1' q_{1,s} - c_{1,s} q_{1,s} + s_s \beta_s q_{1,s} - K.$$

Denote $c'_{1,p} = b'_1 + c_{1,p} - s_p \beta_p$, $c'_{1,s} = b'_1 + c_{1,s} - s_s \beta_s$. Then, we have

$$\pi_{1}^{'} = p_{p}q_{1,p} - c_{1,p}^{'}q_{1,p} + p_{s}q_{1,s} - c_{1,s}^{'}q_{1,s} - K.$$

Note that depending on the value of $b'_1 - s_p \beta_p$, the coefficient $c'_{1,p}$ of the production cost after the implementation could be either higher or lower than $c_{1,p}$.

• The Demand-Side Impact

Prior to the implementation of the symbiotic system, the firm can only label Products P and S as regular. Implementation of the system enables the firm to introduce the "green" (environment-friendly) variants of the two products, in an attempt to capture the consumers who have a relatively higher valuation for these variants and take advantage of their higher willingness to pay. We emphasize that, if the firm so chooses, it can also label a green variant as regular. One of the key characteristics of the symbiotic system is that both Products P and S can be labeled as green *simultaneously*. This symbolizes the "mutually dependant" relationship of the two products. The reactions of the three types of consumers (regular,



flexible green, and dedicated green) to these two variants (see Section 4.2.2) then constitute the demand-side impact of the implementation.

In the next section, we analyze a firm's production decisions for the two products – both in the presence and absence of a symbiotic system – under monopoly as well as under competition.

4.3 Analyzing the Impact of the Symbiotic System under Monopoly and Competition

Section 4.3.1 considers the scenario in which the firm is the only supplier of both Products P and S in their respective markets. The justification of our chosen setting for analyzing competition and some related assumptions is provided in Section 4.3.2. Our discussion in Section 4.3.3 assumes that all green consumers are flexible. The case in which all green consumers are dedicated is discussed in Section 4.3.4. As argued in Section 4.2.3, the operational decisions about Product P and Product S can be made separately. Therefore, our discussion in this section is for a single product, say Product P. Accordingly, we simplify the notation by avoiding the subscript that represents the product (e.g., $c_{1,p}$ is simply denoted as c_1).

4.3.1 Monopoly Setting (Model M)

Consider the situation where Firm 1 is the only supplier of Product P in the market. We assume that the firm can decide the price of the product directly, and then produce to meet



the generated demand. We also assume $c_1 < \alpha$ and $c'_1 < \alpha$. In Section 4.3.1, we discuss the scenario when the firm does not implement the symbiotic system and the scenario when the firm implements the symbiotic system.

No Symbiotic System For Firm 1

If Firm 1 does not implement the symbiotic system, then it can only produce the regular variant. Since the firm is the only supplier, only the regular variant is available in the market. The optimal price, obtained by solving the firm's profit maximization problem, is given in Theorem 4.3.1. Since this is a standard result under a monopoly setting, we avoid providing the proof.

Theorem 4.3.1 In Model M, if Firm 1 does not implement the symbiotic system, then the optimal price of the regular variant is $p_r^* = \frac{(1+2b_1)\alpha + c_1}{2(1+b_1)}$.

Symbiotic System For Firm 1

In this case, the firm can produce both the regular and green variants. The optimal prices for the variants are summarized in the following result.

Theorem 4.3.2 In Model M, if Firm 1 implements the symbiotic system, the optimal price of regular variant is $p_r^* = \frac{\alpha + c_1'}{2}$ and the optimal price of green variant is $p_g^* = \frac{a\alpha + c_1'}{2}$.

Proof: We consider four scenarios.

Scenario M1: $0 < p_g \le p_r$

Since Firm 1 charges a lower price for the green variant, all consumers choose to buy either a green variant or nothing. The aggregate market demand is as follows:



$$Q_r = 0, \quad Q_g = (\alpha - p_g)\theta_1 + (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

Thus, Firm 1 faces the following profit-maximization problem:

$$\max_{p_r, p_g} \quad \pi_1 = (p_g - c_1')Q_g = (p_g - c_1')[(\alpha - p_g)\theta_1 + (\alpha - \frac{p_g}{a})(1 - \theta_1)]$$

The optimal price and optimal profit are as follows:

$$p_r^* \ge p_g^* = \frac{a\alpha + (a-1)\theta_1 c_1^{'} + c_1^{'}}{2[(a-1)\theta_1 + 1]}, \pi_1^{M1} = \frac{[a\alpha - (a-1)\theta_1 c_1^{'} - c_1^{'}]^2}{4a[(a-1)\theta_1 + 1]}.$$

Scenario M2: $p_r \leq p_g \leq ap_r$

Under this scenario, regular consumers purchase either the regular variant or nothing. Meanwhile, flexible and dedicated green consumers purchase either the green variant or nothing. The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)\theta_1, \quad Q_g = (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

Thus, Firm 1 faces the following profit-maximization problem.

$$\max_{p_r, p_g} \pi_1 = (p_r - c_1')Q_r + (p_g - c_1')Q_g$$
$$= (p_r - c_1')(\alpha - p_r)\theta_1 + (p_g - c_1')(\alpha - \frac{p_g}{a})(1 - \theta_1).$$

The optimal solution is in Table 4.3.

Since $p_r^* = \frac{\alpha + c_1'}{2}$, $p_g^* = \frac{a\alpha + c_1'}{2}$, $a \ge 1$, we have $p_r^* \le p_g^* \le ap_r^*$. Thus, Firm 1's maximum profit under Scenario M2 is $\pi_1^{M2} = \frac{\theta_1(\alpha - c_1')^2}{4} + \frac{(1-\theta_1)(a\alpha - c_1')^2}{4a}$.



	Regular Market	Green Market
Q^*	$rac{ heta_1(lpha-c_1')}{2}$	$\frac{(1-\theta_1)(a\alpha - c_1')}{2a}$
p^*	$\frac{\alpha + c_1'}{2}$	$\frac{\frac{a\alpha + c_1}{2}}{2}$
π_1^*	$\frac{\theta_1(\alpha - c_1')^2}{4}$	$\frac{(1-\theta_1)(a\alpha-c_1')^2}{4a}$

Table 4.3. The Optimal Results Under Scenario M2.

We compare the profits under scenarios M1 and M2.

$$\begin{aligned} \pi_1^{M2} - \pi_1^{M1} &= \frac{\theta_1 (\alpha - c_1')^2}{4} + \frac{(1 - \theta_1)(a\alpha - c_1')^2}{4a} - \frac{[a\alpha - (a - 1)\theta_1 c_1' - c_1']^2}{4a[(a - 1)\theta_1 + 1]} \\ &= \frac{\theta_1 (1 - \theta_1)(a - 1)^2 \alpha^2}{4[(a - 1)\theta_1 + 1]} > 0. \end{aligned}$$

Scenario M3: $(a-1)\alpha + p_r \le p_g \le a\alpha$

Under this scenario, regular and flexible green consumers purchase either the regular variant or nothing. Dedicated green consumers purchase either the green variant or nothing. The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)(\theta_1 + \theta_2), \quad Q_g = (\alpha - \frac{p_g}{a})\theta_3.$$

Thus, Firm 1 faces the following profit-maximization problem:

$$\max_{p_r, p_g} \pi_1 = (p_r - c_1')Q_r + (p_g - c_1')Q_g$$

= $(p_r - c_1')(\alpha - p_r)(\theta_1 + \theta_2) + (p_g - c_1')(\alpha - \frac{p_g}{a})\theta_3$
s.t. $(a - 1)\alpha + p_r \le p_g \le a\alpha.$

We first solve the unconstrained problem; the corresponding optimal solution is as follows: Clearly, Firm 1's maximum profit is bounded from above by that of the unconstrained problem. Thus,

$$\pi_1^{M3} \le \frac{(\theta_1 + \theta_2)(\alpha - c_1^{'})^2}{4} + \frac{\theta_3(a\alpha - c_1^{'})^2}{4a}.$$



136

	Regular Market	Green Market
Q^*	$rac{(heta_1+ heta_2)(lpha-c_1')}{2}$	$rac{ heta_3(alpha - c_1')}{2a}$
p^*	$\frac{\alpha + c_1'}{2}$	$\frac{\frac{a\alpha + c_1}{2}}{2}$
π_1^*	$\frac{(\theta_1+\theta_2)(\alpha-c_1')^2}{4}$	$\frac{\theta_3(a\alpha - c_1')^2}{4a}$

Table 4.4. Optimal Results for Firm 1's Unconstrained Problem, under Scenario M3.

We can then compare the profits under Scenarios M2 and M3.

$$\begin{aligned} \pi_1^{M2} - \pi_1^{M3} &\geq \frac{\theta_1(\alpha - c_1')^2}{4} + \frac{(1 - \theta_1)(a\alpha - c_1')^2}{4a} - \frac{(\theta_1 + \theta_2)(\alpha - c_1')^2}{4} - \frac{\theta_3(a\alpha - c_1')^2}{4a} \\ &= \theta_2[\frac{(a\alpha - c_1')^2}{4a} - \frac{(\alpha - c_1')^2}{4}] \\ &= \frac{\theta_2(a - 1)[a\alpha^2 - (c_1')^2]}{4a} > 0. \end{aligned}$$

Scenario M4: $ap_r \le p_g \le (a-1)\alpha + p_r$

Under this scenario, regular consumers purchase the regular variant or nothing and dedicated green consumers purchase the green variant or nothing. Among flexible green consumers, some buy the regular variant, some others buy the green variant, and the remaining buy nothing. The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)\theta_1 + \frac{p_g - ap_r}{a - 1}\theta_2, \quad Q_g = (\alpha - \frac{p_g}{a})\theta_3 + (\alpha - \frac{p_g - p_r}{a - 1})\theta_2.$$

The firm's total profit can be written as follows:

$$\begin{aligned} \max_{p_{r},p_{g}} & \pi_{1} &= Q_{r}p_{r} + Q_{g}p_{g} - c_{1}^{'}(Q_{r} + Q_{g}) \\ &= -(\theta_{1} + \frac{a\theta_{2}}{a-1})p_{r}^{2} + \frac{2\theta_{2}}{a-1}p_{g}p_{r} + [\alpha\theta_{1} + c_{1}^{'}(\theta_{1} + \theta_{2})]p_{r} \\ & -(\frac{\theta_{3}}{a} + \frac{\theta_{2}}{a-1})p_{g}^{2} + [\alpha(\theta_{2} + \theta_{3}) + c_{1}^{'}\frac{\theta_{3}}{a}]p_{g} - \alpha c_{1}^{'} \\ & s.t. \qquad p_{g} - ap_{r} \ge 0, \\ & (a-1)\alpha + p_{r} - p_{g} \ge 0. \end{aligned}$$



The Lagrangean and the Karush-Kuhn-Tucker optimality conditions are:

$$\begin{split} L(p_r,p_g) &= \pi_1 + \lambda_1(p_g - ap_r) + \lambda_2[(a-1)\alpha + p_r - p_g] \\ &\frac{\partial L}{\partial p_r} &= -2(\theta_1 + \frac{a\theta_2}{a-1})p_r + \frac{2\theta_2}{a-1}p_g + [\alpha\theta_1 + c_1^{'}(\theta_1 + \theta_2)] \\ &- a\lambda_1 + \lambda_2 = 0 \\ &\frac{\partial L}{\partial p_g} &= -2(\frac{\theta_3}{a} + \frac{\theta_2}{a-1})p_g + \frac{2\theta_2}{a-1}p_r + [\alpha(\theta_2 + \theta_3) + c_1^{'}\frac{\theta_3}{a}] \\ &+ \lambda_1 - \lambda_2 = 0 \\ &\lambda_1(p_g - ap_r) &= 0, \end{split}$$

$$\lambda_2[(a-1)\alpha + p_r - p_g] &= 0. \end{split}$$

• Scenario M4.1: $\lambda_1 = 0$ and $\lambda_2 = 0$

Solving the above system, we have

$$p_g^* = \frac{a\alpha + c_1^{'}}{2}, \qquad p_r^* = \frac{\alpha + c_1^{'}}{2}.$$

However, since $p_g^* - ap_r^* = \frac{(1-a)c_1'}{2} < 0$, the solution is invalid.

• Scenario M4.2: $p_g = ap_r$

We can rewrite the demand functions as follows:

$$Q_r = (\alpha - p_r)\theta_1, \quad Q_g = (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

Thus, the objective function has the same form as in Scenario M2. However, the optimization problem here has an additional constraint $p_g = ap_r$. Thus, the optimal value obtained in Scenario M2 is at least as good as that under Scenario M4.2.



• Scenario M4.3: $p_g = (a - 1)\alpha + p_r$

We can rewrite the demand functions as follows:

$$Q_r = (\alpha - p_r)(\theta_1 + \theta_2), \quad Q_g = (\alpha - \frac{p_g}{a})\theta_3.$$

Thus, the objective function has the same form as in Scenario M3. However, the optimization problem here has an additional constraint $p_g = (a - 1)\alpha + p_r$. Thus, the optimal value obtained in Scenario M3 is at least as good as that under Scenario M4.3.

Combining the analysis above, we can conclude that the decisions obtained under Scenario M2 are optimal. This completes the proof.

The argument in the proof of Theorem 4.3.2 shows that the optimal setting of the prices corresponds to the scenario when regular consumers buy the regular variant, while all the flexible and dedicated green consumers buy the green variant. We note the following two implications.

Corollary 4.3.3 Under a monopoly, if Firm 1 implements the symbiotic system, then it will avoid charging a price for the green variant that is high enough to drive (some) flexible green consumers to the regular variant. Under optimal prices for the regular and green variants, no flexible green consumer switches to the regular variant.

Consequently, under the optimal prices, the behavior of flexible green consumers and dedicated green consumers is the same.

Corollary 4.3.4 Under a monopoly, green consumers' type (either flexible or dedicated) does not affect Firm 1's optimal decisions.



Next, we discuss the situation when Firm 1 faces a competitor. We first justify our choice of the setting to analyze competition and then specify two assumptions on the production amounts in equilibrium.

4.3.2 The Challenges in the Model Setting under Competition

The first challenge is to choose the type of competition. There are two possible settings: price and quantity. If we assume that the two firms compete on price, then under the assumption of a homogenous product, the firm with a lower variable production cost sets an equilibrium price just below the other firm's variable production cost. Thus, the sales for the high-cost firm become 0, while the low-cost firm captures the entire market. This is hardly the case in practice. Therefore, we choose to use Cournot competition and let the firms compete on quantity. If a firm is the only one to provide a given variant, we assume that it can decide the price of that variant directly.

The next challenge is to determine the market price of the regular variant. In our setting, we have three types of consumers and two variants of the product. The flexible green consumers may purchase the regular variant, when the price of the green variant is high enough. Thus, the information about the prices of the two variants is needed to determine the demand from the flexible green consumers for the regular variant. However, under Cournot competition, the price of the regular variant is determined by the total demand of that variant. To break this deadlock, we make the following assumption: the market price of the regular variant is determined by the Cournot inverse demand function of the regular consumers.

The third challenge is to assign the flexible green consumers' demand of the regular variant



(if any) to the firms. There are two general approaches: (1) both firms share this demand (either equally or based on market share), and (2) one firm (e.g., the powerful one) exclusively supplies this demand. We choose the latter approach. Note that there is no technical difficulty to pursue the former approach. When Firm 1 can produce the green variant and Firm 2 cannot, each firm decides its production quantity of the regular variant. The market price of the regular variant is determined by the combined production quantity of both firms. Thus, each firm only partially influences the market price of the regular variant. Meanwhile, Firm 1 decides the price of the green variant directly. Thus, Firm 1 is relatively more influential in deciding the market prices of the two variants. We emphasize this advantage by allocating the demand of the regular variant from the flexible green consumers to Firm 1.

To focus our analysis on realistic situations, we make the following two assumptions:

 If a firm has the option to label its product as the green variant, then it will produce a positive amount of the green variant. Thus, the firm has two choices: (i) providing only the green variant, or (ii) providing both regular and green variants.

The theoretically possible scenario, in which a firm provides only the regular variant even if it can label its product as green, is rarely seen in practice. A firm will naturally be reluctant to forgo the opportunity to exploit the more-profitable green market. Under Cournot competition, since the price of a variant is determined by the total production quantities of two firms, neither firm can directly decide the prices of both variants. Thus, a firm's action of providing both regular and green variants is legal from the viewpoint of anti-price discrimination regulations (e.g., the Robinson-Patman Act).



2. The total amount of the product (i.e., the sum of the regular and green variants of the product, as applicable) produced by each firm is positive.

As mentioned earlier, we avoid considering the situation where a firm drives its competitor out of the market.

The Chosen Setting under Competition

We assume two firms, Firm 1 and Firm 2, in the market. Competition is realized via the following 2-stage game.

- Stage 1. Each firm indicates the types of variants it will produce. If a firm has the option of labeling its product as the green variant, then it has two choices: (1) providing only the green variant, or (2) providing both the regular and green variants. Each firm decides its choice and makes this decision public.
- Stage 2. If there are two firms who produce a given variant, then the firms decide their individual production quantities simultaneously. The market price of this variant is then determined by the corresponding Cournot demand function. If a firm is the only one to provide a given variant, then it decides the price of that variant directly.

For consistency, we continue to use the notation defined in Section 4.2. We need the following additional notation.

Notation:

- q_2 The production quantity of Firm 2. $C_2(q_2)$ The production cost of Firm 2 to produce q_2 units of product; $C_2(q_2) = c_2 q_2$.
- π_2 The profit of Firm 2.



Although equilibrium results can be obtained in the presence of both dedicated and flexible green consumers, (i.e., $\theta_2 > 0$, $\theta_3 > 0$), for the purpose of deriving managerial insights, we restrict our analysis to two special cases: (i) $\theta_2 = 0$ (all green consumers are dedicated), (ii) $\theta_3 = 0$ (all green consumers are flexible). The case when all green consumers are flexible is considered in Section 4.3.3. The case in which all green consumers are dedicated is analyzed in Section 4.3.4.

4.3.3 Presence of a Competitor When All Green Consumers Are Flexible

In this case, depending on the prices of the green and regular variants, the green consumers may choose to buy either one. As described in Section 4.2.2, consumers' demand for the regular and green variants has multiple possible forms. We focus our attention here on the more advantageous special case (for Firm 1) in which the competitor only produces the regular variant and Firm 1 decides to implement the symbiotic system.

Symbiotic System For Firm 1

Recall our assumption that both firms produce positive amounts of the product. Since Firm 2 can only produce the regular variant, a necessary condition for this to occur is that the price of the regular variant is lower than that of the green variant but higher than Firm 2's variable production cost ($c_2 < p_r^* < p_g^*$). If Firm 1's variable production cost after implementation is less than Firm 2's variable production cost ($c'_1 < c_2$), then the former is lower than the price of the regular variant as well ($c'_1 < p_r^*$). Thus, Firm 1 only producing the green variant is not an equilibrium, since Firm 1 can always improve its profit by providing a nonzero amount of



the regular variant, and gain additional profit. Thus, in equilibrium, Firm 1 provides both regular and green variants, while Firm 2 only provides the regular variant. The following result summarizes the equilibrium prices under three possible parameter settings.

Theorem 4.3.5 When a > 1 and $c'_1 < c_2 < \frac{1}{2} \left[\alpha + \frac{c'_1}{a(1-\theta_1)+\theta_1} \right]$, if Firm 1 implements the symbiotic system, then there are three possible equilibria (depending on parametric relationships). These are described below and pictorially illustrated in Figure 4.4.



Firm 2's Variable Production Cost (c_2)

Figure 4.4. Categorization of Equilibria Based on the Competitor's Variable Production Cost (c₂), when a > 1, $c'_1 < \frac{\theta_1 \alpha}{2(1+\theta_1)}$, and $c'_1 < c_2 < \frac{1}{2} \left[\alpha + \frac{c'_1}{a(1-\theta_1)+\theta_1} \right]$.

1. If $c'_1 < \frac{\theta_1 \alpha}{2(1+\theta_1)}$, then

• Type I: If
$$\left[\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'\right] < c_2 < \frac{1}{2}\left[\alpha + \frac{c_1'}{a(1-\theta_1)+\theta_1}\right]$$
, then we have
 $p_g^I = \frac{a\alpha + c_1'}{2}, \quad q_{1,r}^I = \frac{\theta_1(\alpha - 2c_1' + c_2)}{3}, \quad q_{2,r}^I = \frac{\theta_1(\alpha + c_1' - 2c_2)}{3}$

• Type II: If $(\frac{\alpha}{2} - \frac{1}{\theta_1}c'_1) \le c_2 \le [\frac{\alpha}{2} - (1 - \frac{3}{2a})c'_1]$, then we have $p_g^{II} = \frac{a(a + \theta_1 - a\theta_1)\alpha + ac'_1 + a\theta_1c_2}{2a(1 - \theta_1) + 3\theta_1}, q_{1,r}^{II} = \frac{\theta_1[\theta_1\alpha - 2c'_1 + (2a - 2a\theta_1 + \theta_1)c_2]}{2a(1 - \theta_1) + 3\theta_1},$ $q_{2,r}^{II} = \frac{\theta_1[(a - a\theta_1 + \theta_1)\alpha + c'_1 - 2(a - a\theta_1 + \theta_1)c_2]}{2a(1 - \theta_1) + 3\theta_1}.$

• Type III: If $c_1' < c_2 < (\frac{\alpha}{2} - \frac{1}{\theta_1}c_1')$, then we have

$$p_g^{III} = \frac{(2a + a\theta_1 - \theta_1)\alpha + 2c_1^{'} + 2\theta_1c_2}{2(2 + \theta_1)}$$



$$q_{1,r}^{III} = \frac{\theta_1[\theta_1\alpha - 2c_1^{'} + (2 - \theta_1)c_2]}{2 + \theta_1}, \quad q_{2,r}^{III} = \frac{\theta_1(\alpha + c_1^{'} - 2c_2)}{2 + \theta_1}$$

2. If $\frac{\theta_1 \alpha}{2(1+\theta_1)} \leq c'_1 < \frac{a\alpha}{4a-3}$, then the region representing Type III vanishes.

3. If $\frac{a\alpha}{4a-3} \leq c_1' < \frac{\alpha}{2-\frac{1}{a(1-\theta_1)+\theta_1}}$, then the regions representing Type II and III vanish.

Proof: In equilibrium, Firm 1 provides both regular and green variants. Firm 2 only provides the regular variant. Recall that among the multiple price settings described in Section 4.2.2, only $p_r < p_g \leq ap_r$ and $ap_r \leq p_g < (a-1)\alpha + p_r$ guarantee the existence of both regular and green variants in equilibrium. Thus, following our assumptions in Section 4.3.2, we consider only these two price settings in our analysis.

1. If $c'_1 < \frac{\theta_1 \alpha}{2(1+\theta_1)}$

We first consider the values of the lower and upper bounds of c_2 . Since a > 1, $a(1-\theta_1) + \theta_1$ decreases with an increase in θ_1 . Thus, for $0 \le \theta_1 \le 1$, we have $1 \le a(1-\theta_1) + \theta_1 \le a$. Therefore, we have $\frac{\alpha}{2} + \frac{c_1'}{2a} \le \frac{1}{2} [\alpha + \frac{c_1'}{a(1-\theta_1)+\theta_1}] \le \frac{\alpha}{2} + \frac{c_1'}{2}$. Thus, $c_2 < \frac{1}{2} [\alpha + \frac{c_1'}{a(1-\theta_1)+\theta_1}] \le (\frac{\alpha}{2} + \frac{c_1'}{2})$.

Since $a > 1, 0 \le \theta_1 \le 1$, and $c'_1 < \frac{\theta_1 \alpha}{2(1+\theta_1)}$, we have

$$\begin{aligned} &(\frac{\alpha}{2} + \frac{c_1'}{2a}) - [\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'] = (1 - \frac{1}{a})c_1' > 0, \\ &[\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'] - (\frac{\alpha}{2} - \frac{1}{\theta_1}c_1') = (\frac{3}{2a} + \frac{1 - \theta_1}{\theta_1})c_1' > 0, \\ &(\frac{\alpha}{2} - \frac{1}{\theta_1}c_1') - c_1' > \frac{\alpha}{2} - (1 + \frac{1}{\theta_1})\frac{\theta_1\alpha}{2(1 + \theta_1)} = 0. \end{aligned}$$

Thus, $c_1' < (\frac{\alpha}{2} - \frac{1}{\theta_1}c_1') < [\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'] < (\frac{\alpha}{2} + \frac{c_1'}{2a}) \le \frac{1}{2}[\alpha + \frac{c_1'}{a(1-\theta_1)+\theta_1}]$. Therefore,

all the three regions, which represent the corresponding three types of equilibria, have positive lengths.



We now consider three scenarios:

Scenario D1: $p_r < p_g < ap_r$

Under this scenario, regular consumers only purchase regular products and flexible green consumers only purchase green products. The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)\theta_1, \quad Q_g = (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

Thus, we have $p_r = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1}$. Firm 1's total profit can be written as follows:

$$\max_{q_{1r}, p_g} \quad \pi_1 = q_{1r}(p_r - c_1') + Q_g(p_g - c_1')$$

$$= q_{1r}(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1') + (\alpha - \frac{p_g}{a})(1 - \theta_1)(p_g - c_1').$$

The first-order conditions are

$$\frac{\partial \pi_1}{\partial q_{1r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1' - \frac{q_{1r}}{\theta_1} = 0, \qquad (4.1)$$

$$\frac{\partial \pi_1}{\partial p_g} = (\alpha - \frac{p_g}{a})(1 - \theta_1) - \frac{(1 - \theta_1)(p_g - c_1')}{a} = 0.$$
(4.2)

Next we find the Hessian for $\pi_1(q_{1r}, p_g)$

$$H(q_{1r}, p_g) = \begin{bmatrix} -\frac{2}{\theta_1} & 0\\ 0 & -\frac{2(1-\theta_1)}{a} \end{bmatrix}$$

Since $H_1(q_{1r}, p_g) = -\frac{2}{\theta_1} < 0$, $H_2(q_{1r}, p_g) = (-\frac{2}{\theta_1})(-\frac{2(1-\theta_1)}{a}) > 0$, the first-order conditions are both necessary and sufficient.

Firm 2's profit is $\max_{q_{2r}} \pi_2 = (p_r - c_2)q_{2r} = (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2)q_{2r}$. The first-order condition is

$$\frac{\partial \pi_2}{\partial q_{2r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2 - \frac{q_{2r}}{\theta_1} = 0$$
(4.3)



	Regular Market	Green Market		
q_1^*	$\frac{\theta_1(\alpha - 2c_1' + c_2)}{3}$	$\frac{(1-\theta_1)(a\alpha-c_1')}{2a}$		
q_{2}^{*}	$\frac{\theta_1(\alpha + c_1' - 2c_2)}{3}$	0		
p^*	$\frac{\alpha + c_1^{'} + c_2}{3}$	$\frac{a\alpha + c_1^{'}}{2}$		

Solving (4.1), (4.2), and (4.3), we obtain the following solution:

Verifying Validity: We now verify that the quantities derived above are all positive. Since $c'_1 < c_2 < \frac{\alpha + c'_1}{2}$, we have $\alpha - 2c'_1 + c_2 = (\alpha - c'_1) + (c_2 - c'_1) > 0$, and $\alpha + c'_1 - 2c_2 > 0$. Thus, $q^*_{1,r} > 0$ and $q^*_{2,r} > 0$. Also, since $c'_1 < \frac{\alpha + c'_1}{2}$, we have $c'_1 < \alpha < a\alpha$. Thus, $q^*_{1,g} > 0$. We also need to verify that the optimal prices satisfy the constraint $p_r < p_g < ap_r$. We have $p_g - p_r = \frac{(3a-2)\alpha + c'_1 - 2c_2}{6} \ge \frac{\alpha + c'_1 - 2c_2}{6} > 0$. We also have $ap_r - p_g = \frac{-a\alpha + (2a-3)c'_1 + 2ac_2}{6}$. Therefore, if $c_2 > \frac{\alpha}{2} - (1 - \frac{3}{2a})c'_1$, then we have $ap_r^* > p_g^*$. Thus, the above solution is valid.

Scenario D2: $p_g = ap_r$

The aggregate market demand is the same as that in scenario D1. We have $p_r = \alpha - \frac{q_{1r}+q_{2r}}{\theta_1}$ and $p_g = ap_r$. Firm 1's total profit can be written as follows:

$$\begin{aligned} \max_{q_{1r}} & \pi_1 &= q_{1r}(p_r - c_1') + Q_g(ap_g - c_1') \\ &= q_{1r}(p_r - c_1') + (\alpha - p_r)(1 - \theta_1)(ap_r - c_1') \\ &= q_{1r}(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1') + \frac{(q_{1r} + q_{2r})}{\theta_1}(1 - \theta_1)[a(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1}) - c_1']. \end{aligned}$$

The first-order condition is

$$\frac{\partial \pi_1}{\partial q_{1r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1' - \frac{q_{1r}}{\theta_1} + \frac{(1 - \theta_1)}{\theta_1} [a(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1}) - c_1'] \\
- \frac{a(q_{1r} + q_{2r})}{\theta_1^2} (1 - \theta_1) = 0.$$
(4.4)



Firm 2's profit is $\max_{q_{2r}} \pi_2 = (p_r - c_2)q_{2r} = (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2)q_{2r}$. The first-order condition is

$$\frac{\partial \pi_2}{\partial q_{2r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2 - \frac{q_{2r}}{\theta_1} = 0 \tag{4.5}$$

Solving (4.4) and (4.5), we obtain the following solution:

	Regular Market	Green Market
q_1^*	$\frac{\theta_1[\alpha\theta_1 - 2c_1^{'} + c_2(2a - 2a\theta_1 + \theta_1)]}{2a(1 - \theta_1) + 3\theta_1}$	$\frac{(1-\theta_1)[\alpha(a-a\theta_1+2\theta_1)-c_1'-c_2\theta_1]}{2a(1-\theta_1)+3\theta_1}$
q_2^*	$\frac{\theta_1[\alpha(a-a\theta_1+\theta_1)+c_1'-2c_2(a-a\theta_1+\theta_1)]}{2a(1-\theta_1)+3\theta_1}$	0
p^*	$\frac{\alpha(a+\theta_1-a\theta_1)+c_1'+c_2\theta_1}{2a(1-\theta_1)+3\theta_1}$	$\frac{a[\alpha(a+\theta_1-a\theta_1)+c_1'+c_2\theta_1]}{2a(1-\theta_1)+3\theta_1}$

Verifying Validity: We now verify that the quantities derived above are all positive. Since $c'_1 < c_2$, we have $\alpha \theta_1 - 2c'_1 + c_2(2a - 2a\theta_1 + \theta_1) = \theta_1(\alpha - c'_1) + (2 - \theta_1)(c_2 - c'_1) + 2(a - 1)(1 - \theta_1)c_2 > 0$. Thus, $q^*_{1,r} > 0$. Since $c_2 < \frac{1}{2}[\alpha + \frac{c'_1}{a(1-\theta_1)+\theta_1}]$, we have $\alpha(a - a\theta_1 + \theta_1) + c'_1 - 2c_2(a - a\theta_1 + \theta_1) > 0$. Thus, $q^*_{2,r} > 0$. Also, since $c'_1 < \alpha$ and $c_2 < \alpha$, we have $\alpha(a - a\theta_1 + 2\theta_1) - c'_1 - c_2\theta_1 > \alpha(a - a\theta_1 + 2\theta_1) - \alpha - \theta_1\alpha = \alpha(a - 1)(1 - \theta_1) \ge 0$. Thus, $q^*_{1,g} > 0$.

Scenario D3: $ap_r < p_g < (a-1)\alpha + p_r$

Under this scenario, regular consumers purchase the regular variant or nothing. Among flexible green consumers, some buy the regular variant, some others buy the green variant, and the remaining buy nothing. The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)\theta_1 + \frac{p_g - ap_r}{a - 1}(1 - \theta_1), \quad Q_g = (\alpha - \frac{p_g - p_r}{a - 1})(1 - \theta_1).$$

Recall our assumption (from Section 4.3.2) that the market price of the regular variant is determined by the Cournot inverse demand function of the regular consumers. Thus,



we have $p_r = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1}$. Firm 1's total profit can be written as follows:

$$\begin{aligned} \max_{q_{1r},p_g} \quad \pi_1 &= \quad [q_{1r} + \frac{p_g - ap_r}{a - 1}(1 - \theta_1)](p_r - c_1^{'}) + (\alpha - \frac{p_g - p_r}{a - 1})(1 - \theta_1)(p_g - c_1^{'}) \\ &= \quad [q_{1r} + \frac{p_g - a(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1})}{a - 1}(1 - \theta_1)](\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1^{'}) \\ &+ (\alpha - \frac{p_g - (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1})}{a - 1})(1 - \theta_1)(p_g - c_1^{'}). \end{aligned}$$

The firs-order conditions are

$$\frac{\partial \pi_1}{\partial q_{1r}} = \left[1 + \frac{a(1-\theta_1)}{(a-1)\theta_1}\right] \left(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1'\right) - \frac{1}{\theta_1} \left[q_{1r} + \frac{p_g - a(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1})}{a-1} (1-\theta_1)\right] - \frac{(1-\theta_1)(p_g - c_1')}{\theta_1(a-1)} = 0, \quad (4.6)$$

$$\frac{\partial \pi_1}{\partial p_g} = \frac{1-\theta_1}{a-1} \left(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1'\right) + \left(\alpha - \frac{p_g - (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1})}{a-1}\right) (1-\theta_1)$$

$$\frac{1}{p_g} = \frac{1}{a-1} \left(\alpha - \frac{1}{\theta_1} - c_1 \right) + \left(\alpha - \frac{1}{a-1} - c_1 \right) \left(1 - \theta_1 \right) - \frac{1 - \theta_1}{a-1} \left(p_g - c_1' \right) = 0.$$
(4.7)

Next, we consider the Hessian for $\pi_1(q_{1r}, p_g)$

$$H(q_{1r}, p_g) = \begin{bmatrix} -\frac{2(a-\theta_1)}{\theta_1^2(a-1)} & -\frac{2(1-\theta_1)}{\theta_1(a-1)} \\ -\frac{2(1-\theta_1)}{\theta_1(a-1)} & -\frac{2(1-\theta_1)}{a-1} \end{bmatrix}$$

Since $H_1(q_{1r}, p_g) = -\frac{2(a-\theta_1)}{\theta_1^2(a-1)} < 0$, $H_2(q_{1r}, p_g) = \frac{4(1-\theta_1)}{\theta_1^2(a-1)} > 0$, the first-order conditions are both necessary and sufficient.

Firm 2's profit is

$$\max_{q_{2r}} \pi_2 = (p_r - c_2)q_{2r}$$
$$= (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2)q_{2r}$$

The first-order condition is

$$\frac{\partial \pi_2}{\partial q_{2r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2 - \frac{q_{2r}}{\theta_1} = 0$$
(4.8)



	Regular Market	Green Market
q_{1}^{*}	$\frac{\theta_1[\alpha\theta_1 - 2c_1' + c_2(2 - \theta_1)]}{2 + \theta_1}$	$\frac{\alpha(1- heta_1)}{2}$
q_{2}^{*}	$\frac{\theta_1(\alpha + c_1' - 2c_2)}{2 + \theta_1}$	0
p^*	$\frac{\alpha + c_1^{'} + c_2 \theta_1}{2 + \theta_1}$	$\frac{\alpha(2a+a\theta_1-\theta_1)+2c_1'+2c_2\theta_1}{2(2+\theta_1)}$

Solving (4.6), (4.7), (4.8), we obtain the following solution:

Verifying Validity: Since $c_1' < c_2$ and $c_1' < \alpha$, we have $\alpha \theta_1 - 2c_1' + c_2(2 - \theta_1) = \theta_1(\alpha - c_1') + (2 - \theta_1)(c_2 - c_1') > 0$. Thus, $q_{1,r}^* > 0$. Since $c_2 < (\frac{\alpha}{2} + \frac{c_1'}{2})$, we have $q_{2,r}^* > 0$. Also, $q_{1,g}^* > 0$.

We also need to verify that the optimal prices satisfy the constraint $p_g \ge ap_r$. We have $p_g - ap_r = \frac{(a-1)(\theta_1 \alpha - 2c_1' - 2\theta_1 c_2)}{2(2+\theta_1)}$. Thus, if $c_2 < (\frac{\alpha}{2} - \frac{1}{\theta_1}c_1')$, then we have $p_g - ap_r > 0$.

We consider the two interior solutions in scenario D1 and scenario D3 (given that each satisfies the corresponding constraint) and one boundary solution in scenario D2. The equilibrium prices in these three scenarios are categorized and validated as follows:

- Type I: If $p_r < p_g < ap_r$, we have

$$p_r^* = \frac{\alpha + c_1^{'} + c_2}{3}, p_g^* = \frac{a\alpha + c_1^{'}}{2}$$

Under the imposed condition, $\left[\frac{\alpha}{2} - \left(1 - \frac{3}{2a}c_1'\right)\right] < c_2 < \frac{1}{2}\left[\alpha + \frac{c_1'}{a(1-\theta_1)+\theta_1}\right]$, we indeed have have $p_r^* < p_g^* < ap_r^*$.

- Type II: If $ap_r = p_g$, we have

$$p_r^* = \frac{(a+\theta_1 - a\theta_1)\alpha + c_1^{'} + \theta_1 c_2}{2a(1-\theta_1) + 3\theta_1}, p_g^* = \frac{a(a+\theta_1 - a\theta_1)\alpha + ac_1^{'} + a\theta_1 c_2}{2a(1-\theta_1) + 3\theta_1}$$



- Type III: If $ap_r < p_g < (a-1)\alpha + p_r$, we have

$$p_r^* = \frac{\alpha + c_1^{'} + \theta_1 c_2}{2 + \theta_1}, p_g^* = \frac{(2a + a\theta_1 - \theta_1)\alpha + 2c_1^{'} + 2\theta_1 c_2}{2(2 + \theta_1)}$$

Under the imposed condition, $c'_1 < c_2 < (\frac{\alpha}{2} - \frac{1}{\theta_1}c'_1)$, we have $ap_r^* < p_g^* < (a - 1)\alpha + p_r^*$.

Since $\left[\frac{\alpha}{2} - \left(1 - \frac{3}{2a}c'_{1}\right)\right] > \left[\frac{\alpha}{2} - \frac{1}{\theta_{1}}c'_{1}\right]$, Type I and Type III solutions cannot both be valid. When either of these two solutions is valid, it is straightforward to show that the valid solution is also better than the Type II solution. When neither is valid, the Type II solution is optimal.

2. If
$$\frac{\theta_1 \alpha}{2(1+\theta_1)} \le c_1' < \frac{a\alpha}{4a-3}$$

We have

$$\begin{aligned} &(\frac{\alpha}{2} + \frac{c_1'}{2a}) - [\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'] = (1 - \frac{1}{a})c_1' > 0, \\ &[\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'] - c_1' > \frac{\alpha}{2} - (2 - \frac{3}{2a})\frac{a\alpha}{4a - 3} = 0, \\ &(\frac{\alpha}{2} - \frac{1}{\theta_1}c_1') - c_1' \le \frac{\alpha}{2} - (1 + \frac{1}{\theta_1})\frac{\theta_1\alpha}{2(1 + \theta_1)} = 0. \end{aligned}$$

Thus, $(\frac{\alpha}{2} - \frac{1}{\theta_1}c'_1) \leq c'_1 < [\frac{\alpha}{2} - (1 - \frac{3}{2a}c'_1)] < (\frac{\alpha}{2} + \frac{c'_1}{2a})$. Therefore, only the two regions corresponding to Types I and II equilibria have positive lengths.

3. If
$$\frac{a\alpha}{4a-3} \le c'_1 < \frac{\alpha}{2 - \frac{1}{a(1-\theta_1)+\theta_1}}$$

We have $\left[\frac{\alpha}{2} - (1 - \frac{3}{2a})c_1'\right] \le c_1' < \left(\frac{\alpha}{2} + \frac{c_1'}{2a}\right)$. Only the region corresponding to the Type I equilibrium has positive length.

This completes the proof.



www.manaraa.com

Note that in both Type I and Type II equilibria, all the flexible green consumers buy either the green variant or nothing. In the Type III equilibrium, some flexible green consumers switch to the regular variant. As a consequence, we have the following corollary.

Corollary 4.3.6 When the competitor's cost disadvantage is marginal (c_2 is in Region III of Figure 4.4), Firm 1 does not have the power to retain all the green consumers. Some low-end flexible green consumers switch to the regular variant. When the competitor's cost disadvantage increases (c_2 is in either Region I or Region II), Firm 1 obtains this power.

4.3.4 Presence of a Competitor When All Green Consumers Are Dedicated

When all green consumers are dedicated, they purchase either the green variant or nothing. Thus, the demand of the green variant is unaffected by the price of the regular variant. Accordingly, only the price of the green variant determines its demand. Recall in Section 4.2.2, we list the aggregate market demand under various scenarios. Due to the absence of the flexible green consumers, the aggregate market demand now has only two different forms when both regular and green variants are available.

(i) $p_g \leq p_r$

Under this scenario, all consumers purchase either the green variant or nothing. The aggregate market demand is as follows:

$$Q_r = 0, \quad Q_g = (\alpha - p_g)\theta_1 + (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

(ii) $p_g > p_r$

Under this scenario, the regular consumers purchase either the regular variant or nothing. Meanwhile, the green consumers purchase either the green variant or nothing.



The aggregate market demand is as follows:

$$Q_r = (\alpha - p_r)\theta_1, \quad Q_g = (\alpha - \frac{p_g}{a})(1 - \theta_1).$$

Since our goal is to avoid exploring all theoretical possibilities and, instead, focus on practicallyrelevant scenarios in which the two firms enjoy healthy competition, we impose the following assumption: The two competing firms have similar production economies. In other words, both firms' variable production costs are of the same magnitude, both before and after Firm 1's implementation of the symbiotic system. Specifically, we assume that $2c_1 - \alpha < c_2 < \frac{\alpha+c_1}{2}$ and $2c'_1 - \alpha < c_2 < \frac{\alpha+c'_1}{2}$. This assumption is reasonable in practice, since a significant difference in the production costs of the two firms will typically result in the weaker firm being driven out of the market.

Next, we first describe Model CR, in which Firm 1 faces a competitor (Firm 2) that only produces the regular variant of the product. The case in which the competitor produces the green variant (Model CG) is analyzed later.

No Symbiotic System For Firm 1 Under Model CR

In the absence of the symbiotic system, only the regular variant is available in the market. Thus, both firms compete only in the regular market, where all consumers have the same distribution of valuation. The following result states the equilibrium solution. Since the proof is straightforward, we avoid providing it.

Theorem 4.3.7 In Model CR, if Firm 1 does not implement the symbiotic system, the equilibrium production quantities of the two firms are: $q_1^* = \frac{\alpha - 2c_1 + c_2}{4b_1 + 3}$ (Firm 1), $q_2^* = \frac{(2b_1+1)\alpha + c_1 - 2(b_1+1)c_2}{4b_1 + 3}$ (Firm 2).



Symbiotic System For Firm 1 Under Model CR

If Firm 1 implements the symbiotic system, then it has the option of labeling its product as the green variant. The two choices for Firm 1 are (1) Strategy g: provide only the green variant or (2) Strategy rg: provide both the regular and green variants. Let π_1^g (resp., π_1^{rg}) denote Firm 1's profit under Strategy g (resp., Strategy rg). There are two questions: in equilibrium, (i) which strategy will Firm 1 choose? and (ii) what are the production quantities of each variant for each firm? The following result answers these questions.

Theorem 4.3.8 In Model CR, if Firm 1 implements the symbiotic system, it will provide both regular and green variants. The equilibrium production quantities are: $q_{1,r}^* = \frac{\theta_1(\alpha - 2c_1' + c_2)}{3}$, $q_{2,r}^* = \frac{\theta_1(\alpha + c_1' - 2c_2)}{3}$, and $q_{1,g}^* = \frac{(1-\theta_1)(\alpha\alpha - c_1')}{2a}$.

Proof: Recall that the aggregate market demand has two different forms under two scenarios. Since Firm 2 only produces the regular variant, the condition for Firm 2 to produce positive amount in equilibrium is $p_g > p_r$. Thus, we only consider this scenario in the following analysis. We consider Firm 1's two choices separately.

• Strategy rg

Firm 1 provides both regular and green variants, while Firm 2 provides the regular variant.

When $p_r < p_g$, we have $Q_r = (\alpha - p_r)\theta_1, Q_g = (\alpha - \frac{p_g}{a})(1 - \theta_1).$

Firm 1's profit-maximization problem is as follows:

$$\max_{q_{1r}, p_g} \quad \pi_1 = q_{1r}(p_r - c_1') + Q_g(p_g - c_1')$$

$$= q_{1r}(\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1') + (\alpha - \frac{p_g}{a})(1 - \theta_1)(p_g - c_1').$$



Thus ,we have

$$\frac{\partial \pi_1}{\partial q_{1r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_1' - \frac{q_{1r}}{\theta_1} = 0$$
(4.9)

$$\frac{\partial \pi_1}{\partial p_g} = (\alpha - \frac{p_g}{a})(1 - \theta_1) - \frac{(1 - \theta_1)(p_g - c_1')}{a} = 0.$$
(4.10)

Firm 2 faces the following problem:

$$\max_{q_{2r}} \quad \pi_2 = (p_r - c_2)q_{2r} = (\alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2)q_{2r}.$$

We can obtain the first order condition:

$$\frac{\partial \pi_2}{\partial q_{2r}} = \alpha - \frac{q_{1r} + q_{2r}}{\theta_1} - c_2 - \frac{q_{2r}}{\theta_1} = 0$$
(4.11)

By solving equations 4.9, 4.10, and 4.11 simultaneously, we obtain the following equilibrium:

$$q_{1,r}^* = \frac{\theta_1(\alpha - 2c_1' + c_2)}{3}, q_{2,r}^* = \frac{\theta_1(\alpha + c_1' - 2c_2)}{3}, q_{1,g}^* = \frac{(1 - \theta_1)(a\alpha - c_1')}{2a},$$
$$p_r^* = \frac{\alpha + c_1' + c_2}{3}, p_g^* = \frac{a\alpha + c_1'}{2}.$$

To check the validity of this solution, we first examine the difference between the prices of the two variants. We have $p_g^* - p_r^* = \frac{(3a-2)\alpha + c_1' - 2c_2}{6} \ge \frac{\alpha + c_1' - 2c_2}{6}$. Since $c_2 < \frac{\alpha + c_1'}{2}$, we have $p_r^* < p_g^*$.

We then check whether all the production quantities are positive. Since $c_2 > 2c'_1 - \alpha$, we have $q_{1,r}^* > 0$. Since $c_2 < \frac{\alpha + c'_1}{2}$, we have $q_{2,r}^* > 0$. Since $c'_1 < \alpha$ and $a \ge 1$, we have $q_{1,g}^* > 0$. Thus, the solution above is valid. Firm 1's profit under Strategy rg is $\pi_1^{rg} = \frac{(1-\theta_1)(a\alpha - c'_1)^2}{4a} + \frac{\theta_1(\alpha - 2c'_1 + c_2)^2}{9}$.



• Strategy g

Then Firm 1 only provides the green variant, while Firm 2 provides the regular variant. According to Assumption 3, Firm 2 produces positive amount of product. Thus, the price of the regular variant should be less than that of the green variant. Therefore, Firm 1 can only obtain profit from the green consumers. Note that, Firm 1's maximum profit from the green consumers is $\frac{(1-\theta_1)(a\alpha-c_1')^2}{4a}$. Thus, Firm 1's profit under Strategy $g, \pi_1^g \leq \frac{(1-\theta_1)(a\alpha-c_1')^2}{4a} < \pi_1^{rg}$. The result follows.

Next, we consider the case in which a competitor has the ability to produce the green variant.

No Symbiotic System For Firm 1 Under Model CG

Under Model CG, we consider a competitor that also has the ability to produce the green variant. If Firm 1 does not implement the symbiotic system, then only Firm 2 has the option to label its product as the green variant. Firm 2 has two strategies, g and rg. The following result states the equilibrium.

Theorem 4.3.9 In Model CG, if Firm 1 does not implement the symbiotic system, then Firm 2 will provide both regular and green variants. The equilibrium production quantities are: $q_{1,r}^* = \frac{\theta_1(\alpha - 2c_1 + c_2)}{3 + 4b_1\theta_1}$, $q_{2,r}^* = \frac{\theta_1[(2b_1\theta_1 + 1)\alpha + c_1 - 2(b_1\theta_1 + 1)c_2]}{3 + 4b_1\theta_1}$, and $q_{2,g}^* = \frac{(1 - \theta_1)(a\alpha - c_2)}{2a}$.

Proof: Since Firm 2 can provide either only the green variant or both the regular and the green variants, there are two possible scenarios.

1. Firm 2 chooses Strategy rg:

Firm 1 provides the regular variant and Firm 2 provides both regular and green variants.



By using an approach similar to that in the proof of Theorem 4.3.8, we have the following equilibrium:

$$q_{1,r}^* = \frac{\theta_1(\alpha - 2c_1 + c_2)}{3 + 4b_1\theta_1}, q_{2,r}^* = \frac{\theta_1[(2b_1\theta_1 + 1)\alpha + c_1 - 2(b_1\theta_1 + 1)c_2]}{3 + 4b_1\theta_1},$$
$$q_{2,g}^* = \frac{(1 - \theta_1)(a\alpha - c_2)}{2a}, p_r^* = \frac{(2b_1\theta_1 + 1)\alpha + c_1 + (2b_1\theta_1 + 1)c_2}{3 + 4b_1\theta_1}, p_g^* = \frac{a\alpha + c_2}{2}$$

To check the validity of this solution, we first examine the difference between the prices of the two variants. We have $p_g^* - p_r^* = \frac{[a(3+4b_1\theta_1)-(2+4b_1\theta_1)]\alpha-2c_1+c_2}{2(3+4b_1\theta_1)}$. Since $c_2 > 2c_1 - \alpha$, we have $[a(3+4b_1\theta_1) - (2+4b_1\theta_1)]\alpha - 2c_1 + c_2 > (a-1)(3+4b_1\theta_1)\alpha \ge 0$. Thus, $p_g^* - p_r^* > 0$. Since $c_2 > 2c_1 - \alpha$, we have $q_{1,r}^* > 0$. Also, $c_2 < \frac{\alpha+c_1}{2}$ implies $[(2b_1\theta_1+1)\alpha+c_1-2(b_1\theta_1+1)c_2] > b_1\theta_1(\alpha-c_1) \ge 0$. Thus, we have $q_{2,r}^* > 0$. Since $c_2 < \alpha$ and $a \ge 1$, we have $q_{2,g}^* > 0$. Thus, the solution above is valid.

Firm 2's profit under Strategy rg is $\pi_2^{rg} = \frac{(1-\theta_1)(a\alpha-c_2)^2}{4a} + \frac{\theta_1[(2b_1\theta_1+1)\alpha+c_1-2(b_1\theta_1+1)c_2]^2}{(3+4b_1\theta_1)^2}.$

2. Firm 2 chooses Strategy g:

Then, Firm 1 provides the regular variant, while Firm 2 provides the green variant.

Since Firm 1 produces a positive amount (Assumption 3 above), the price of the regular variant is less than that of the green variant. Therefore, Firm 2 derives its entire profit from the green consumers; the maximum value of this profit is $\frac{(1-\theta_1)(a\alpha-c_2)^2}{4a}$. Thus, Firm 2's profit under Strategy g, $\pi_2^g \leq \frac{(1-\theta_1)(a\alpha-c_2)^2}{4a} < \pi_2^{rg}$.

The result follows.



Symbiotic System For Firm 1 Under Model CG

Under this setting, both firms are capable of offering both variants. Each firm has two strategies. We denote $\pi_i^{j,k}$ as Firm *i*'s profit if Firm 1 chooses Strategy *j* and Firm 2 chooses Strategy *k*; $i \in \{1,2\}, j,k \in \{g,rg\}$. The following result shows that both firms provide both regular and green variants in equilibrium and derives the corresponding production quantities. The payoffs of the two firms under the four scenarios are listed in Table 4.5. Table 4.5. The Two Firms' Payoffs when Both can Provide Regular and Green Variants.

		Firm 2's Strategy		
		Only Green		Regular & Green
Firm 1's	Only Green	$(\pi_1^{g,g},\pi_2^{g,g})$	\rightarrow	$(\pi_1^{g,rg},\pi_2^{g,rg})$
		\downarrow		\downarrow
Strategy	Regular & Green	$(\pi_1^{rg,g},\pi_2^{rg,g})$	\rightarrow	$(\pi_1^{rg,rg},\pi_2^{rg,rg})$

Theorem 4.3.10 In Model CG, if Firm 1 implements the symbiotic system, then in equilibrium, both firms provide both the regular and the green variants. The equilibrium production quantities are:

$$q_1^{r*} = \frac{\theta_1(\alpha - 2c_1^{'} + c_2)}{3}, \quad q_1^{g*} = \frac{(1 - \theta_1)(a\alpha - 2c_1^{'} + c_2)}{3a},$$
$$q_2^{r*} = \frac{\theta_1(\alpha + c_1^{'} - 2c_2)}{3}, \quad q_2^{g*} = \frac{(1 - \theta_1)(a\alpha + c_1^{'} - 2c_2)}{3a}.$$

Proof: Since each firm has two strategies, there are four possible combinations. We derive the payoffs of both firms under each combination in the following analysis.

1. Combination (g, g): If both firms only provide the green variant

All consumers choose to buy either a green variant or nothing. The aggregate market demand is as follows:

$$q_{1,g} + q_{2,g} = Q_g = (\alpha - p_g)\theta_1 + (\alpha - \frac{p_g}{a})(1 - \theta_1).$$



Thus, $p_g = \frac{\alpha - q_{1,g} - q_{2,g}}{\theta_1 + \frac{1 - \theta_1}{a}}$. The two firms face the following profit-maximization problems:

$$\max_{q_{1,g}} \quad \pi_1 = (p_g - c_1')q_{1,g},$$
$$\max_{q_{2,g}} \quad \pi_2 = (p_g - c_2)q_{2,g}.$$

By using the method similar to that used in the proof of Theorem 4.3.8, we obtain the equilibrium results as follows:

$$p_g^* = \frac{1}{3} [c_1^{'} + c_2 + \frac{a\alpha}{1 + (a - 1)\theta_1}],$$

$$q_{1,g}^* = \frac{a\alpha - 2c_1^{'} + c_2 - (a - 1)(2c_1^{'} - c_2)\theta_1}{3a}, q_{2,g}^* = \frac{a\alpha + c_1^{'} - 2c_2 + (a - 1)(c_1^{'} - 2c_2)\theta_1}{3a},$$

$$\pi_1^* = \frac{[a\alpha - 2c_1^{'} + c_2 - (a - 1)(2c_1^{'} - c_2)\theta_1]^2}{9a[1 + (a - 1)\theta_1},$$

$$\pi_2^* = \frac{[a\alpha + c_1^{'} - 2c_2 + (a - 1)(c_1^{'} - 2c_2)\theta_1]^2}{9a[1 + (a - 1)\theta_1}.$$

We check whether the production quantities derived are positive.

- If $(2c'_1 c_2) \ge 0$, then $[a\alpha 2c'_1 + c_2 (a 1)(2c'_1 c_2)\theta_1]$ reaches its minimum when θ_1 reaches its upper bound 1. Thus, $a\alpha - 2c'_1 + c_2 - (a - 1)(2c'_1 - c_2)\theta_1 \ge a\alpha - 2c'_1 + c_2 - (a - 1)(2c'_1 - c_2) = a(\alpha - 2c'_1 + c_2)$. Since $c_2 > 2c'_1 - \alpha$, we have $a(\alpha - 2c'_1 + c_2) > 0$. Thus, $q^*_{1,g} > 0$.
- If $(2c'_1 c_2) < 0$, then $[a\alpha 2c'_1 + c_2 (a 1)(2c'_1 c_2)\theta_1]$ reaches its minimum when θ_1 reaches its lower bound 0. Thus, $a\alpha - 2c'_1 + c_2 - (a - 1)(2c'_1 - c_2)\theta_1 \ge a\alpha - 2c'_1 + c_2 \ge (\alpha - 2c'_1 + c_2) > 0$. Thus, $q^*_{1,g} > 0$.

Similarly, we can show $q_{2,g}^* > 0$. Thus, the solution above is valid. These two firms' profits are as follows.



$$\pi_{1}^{g,g} = \frac{[a\alpha - 2c_{1}^{'} + c_{2} - (a-1)(2c_{1}^{'} - c_{2})\theta_{1}]^{2}}{9a[1 + (a-1)\theta_{1}]},$$

$$\pi_{2}^{g,g} = \frac{[a\alpha + c_{1}^{'} - 2c_{2} + (a-1)(c_{1}^{'} - 2c_{2})\theta_{1}]^{2}}{9a[1 + (a-1)\theta_{1}]}.$$

2. Combination (rg, rg): If both firms provide both regular and green variants

Under this scenario, both firms compete in both the regular and the green markets. We obtain the equilibrium results as follows:

$$p_{r}^{*} = \frac{\alpha + c_{1}^{'} + c_{2}}{3}, \\ p_{g}^{*} = \frac{a\alpha + c_{1}^{'} + c_{2}}{3}, \\ q_{1,r}^{*} = \frac{\theta_{1}(\alpha - 2c_{1}^{'} + c_{2})}{3}, \\ q_{2,r}^{*} = \frac{\theta_{1}(\alpha + c_{1}^{'} - 2c_{2})}{3}, \\ q_{1,g}^{*} = \frac{(1 - \theta_{1})(a\alpha - 2c_{1}^{'} + c_{2})}{3a}, \\ q_{2,g}^{*} = \frac{(1 - \theta_{1})(a\alpha + c_{1}^{'} - 2c_{2})}{3a}.$$

We have $p_g^* - p_r^* = \frac{(a-1)\alpha}{3} > 0$. We can show all four production quantities are positive. These two firms' profits are as follows.

$$\pi_1^{rg,rg} = \frac{\theta_1(\alpha - 2c_1' + c_2)^2}{9} + \frac{(1 - \theta_1)(a\alpha - 2c_1' + c_2)^2}{9a},$$
$$\pi_2^{rg,rg} = \frac{\theta_1(\alpha + c_1' - 2c_2)^2}{9} + \frac{(1 - \theta_1)(a\alpha + c_1' - 2c_2)^2}{9a}.$$

If we compare the profits under this setting with those under (g,g), we have

$$\pi_1^{rg,rg} - \pi_1^{g,g} = \frac{\theta_1(1-\theta_1)(a-1)^2\alpha^2}{9[1+(a-1)\theta_1]} \ge 0,$$

$$\pi_2^{rg,rg} - \pi_2^{g,g} = \frac{\theta_1(1-\theta_1)(a-1)^2\alpha^2}{9[1+(a-1)\theta_1]} \ge 0.$$

3. Combination (g, rg): If Firm 1 only provides the green variant, Firm 2 provides both regular and green variants



We now compare these two firms' profits under this setting and those under the setting (rg, rg). First, both firms obtain the same amount of profits from the green market under both settings. Second, Firm 1 produces the regular variant under the setting (rg, rg) but not under the setting (g, rg). Thus, in the regular market, Firm 1 obtains positive profit under the setting (rg, rg) but 0 under the setting (g, rg). Third, in the regular market, Firm 2 competes with Firm 1 under the setting (rg, rg) but is the exclusive supplier under the setting (g, rg). Thus, Firm 2 obtains more profit in the regular market under the setting (g, rg). Thus, Firm 2 obtains more profit in the negalar market under the setting (rg, rg) than under the setting (g, rg). Therefore, we have the following results:

$$\pi_1^{rg,rg} > \pi_1^{g,rg}, \quad \pi_2^{g,rg} > \pi_2^{rg,rg}.$$

4. Combination (rg, g): If Firm 1 provides both regular and green variants, Firm 2 only provides the green variant

This scenario is similar to Combination (g, rg). If we compare these two firms' profits under this setting and under the setting (rg, rg), we have the following results:

$$\pi_1^{rg,g} > \pi_1^{rg,rg}, \quad \pi_2^{rg,rg} > \pi_2^{rg,g}.$$

Now we derive the Nash equilibrium. Since we have $\pi_1^{rg,g} > \pi_1^{rg,rg} > \pi_1^{g,g}$ and $\pi_1^{rg,rg} > \pi_1^{g,rg}$, Firm 1's dominating strategy is rg no matter which strategy Firm 2 chooses. Similarly, we have $\pi_2^{rg,rg} > \pi_2^{rg,g}$ and $\pi_2^{g,rg} > \pi_2^{rg,rg} > \pi_2^{g,g}$. Thus, Firm 2's dominating strategy is rg as well. Therefore, (rg, rg) is the only Nash equilibrium. Both Firms provide both the regular and the green variants. This completes the proof.

In the next section, we use the obtained results to derive some useful managerial insights.



4.4 Understanding the Willingness to Implement the Symbiotic System

In the previous section, we derived Firm 1's optimal (under monopoly) and equilibrium (under competition) decisions, under various settings. These results enable us to assess the firm's profits both before and after the implementation of a symbiotic system. We interpret the difference between these two profits as the firm's "willingness" to implement the system. Note that this difference may not always be positive. Accordingly, our interest is in understanding the forces that influence willingness and the impact on their relative strengths with respect to changes in operational parameters, consumer characteristics, and competition. Section 4.4.1 (resp., Section 4.4.2) discusses the situation under Model M (resp., Model CR). We also briefly comment on the main differences under Model CG. Next, in Section 4.4.3, we compare the firm's willingness under monopoly with that under competition. Our focus is on analyzing the impact of the nature of competition (competitors offering only the regular variants vs. both variants) and on identifying scenarios where competition improves the firm's willingness to implement. Finally, Section 4.4.4 discusses the impact of the implementation of a symbiotic system on consumer welfare. Here, our aim is to identify situations in which the firm's willingness to implement is positive and consumers (as a whole) benefit from the implementation as well.

For simplicity, when considering competition, we assume that green consumers are dedicated. The case when green consumers are flexible can also be analyzed, but is technically more cumbersome. To calculate the firm's profits before and after implementation, we need to consider the decisions for both Product P and Product S. Therefore, we introduce the corresponding subscripts (p and s) to the notation defined for the single-product models in



earlier sections. Let π_1^* (resp., $\pi_1^{'*}$) denote Firm 1's maximum total profit before (resp., after) implementing the symbiotic system.

4.4.1 The Willingness in a Monopoly

Recall from Section 4.3.1 the optimal decisions under monopoly, obtained under the assumptions $\max\{c_{1,p}, c'_{1,p}\} < \alpha_p$ and $\max\{c_{1,s}, c'_{1,s}\} < \alpha_s$. Before implementing the symbiotic system, Firm 1's total profit from products P and S is:

$$\pi_1^* = \frac{(\alpha_p - c_{1,p})^2}{4(1+b_1)} + \frac{(\alpha_s - c_{1,s})^2}{4(1+b_1)}$$

The total profit after implementation is:

$$\pi_{1}^{'*} = \frac{\theta_{1,p}(\alpha_{p} - c_{1,p}^{'})^{2}}{4} + \frac{(1 - \theta_{1,p})(a_{p}\alpha_{p} - c_{1,p}^{'})^{2}}{4a_{p}} + \frac{\theta_{1,s}(\alpha_{s} - c_{1,s}^{'})^{2}}{4} + \frac{(1 - \theta_{1,s})(a_{s}\alpha_{s} - c_{1,s}^{'})^{2}}{4a_{s}} - K.$$

To capture the firm's willingness to implement, let $\Delta_M = \pi_1^{\prime *} - \pi_1^*$. To better understand the impact of parametric changes, it is convenient to partition the expression for Δ_M into four terms:

$$\Delta_{M} = \underbrace{\frac{(1-\theta_{1,p})(a_{p}-1)(a_{p}\alpha_{p}^{2}-c_{1,p}^{'2})}{4a_{p}} + \frac{(1-\theta_{1,s})(a_{s}-1)(a_{s}\alpha_{s}^{2}-c_{1,s}^{'2})}{4a_{s}}}_{T_{1}^{M}: \text{ The gain from exploiting the green market segment}} \\ + \underbrace{\frac{(c_{1,p}-c_{1,p}^{'})(2\alpha_{p}-c_{1,p}-c_{1,p}^{'})}{4}}_{T_{2}^{M}: \text{ The gain/loss from the changes in the variable production costs}} \\ + \underbrace{\frac{b_{1}(\alpha_{p}-c_{1,p})^{2}}{4(1+b_{1})} + \frac{b_{1}(\alpha_{s}-c_{1,s})^{2}}{4(1+b_{1})}}_{T_{3}^{M}: \text{ The gain from better usage of the waste}} - \underbrace{K_{1}^{K} + \underbrace{K_{2}^{K}}_{T_{4}^{M}: \text{ The fixed cost}}}_{T_{4}^{M}: \text{ The fixed cost}} + \underbrace{k_{2}^{K} + \underbrace{k_{2}^{K}}_{T_{3}^{M}: \text{ The gain from better usage of the waste}}_{T_{4}^{M}: \text{ The fixed cost}} + \underbrace{k_{2}^{K} + \underbrace{k_{2}^{K} + \underbrace{k_{2}^{K}}_{T_{4}^{K}: \text{ The gain from better usage of the waste}}_{T_{4}^{M}: \text{ The fixed cost}} + \underbrace{k_{2}^{K} + \underbrace{$$

These four terms can be further categorized into two types: Demand-side and Cost-side.


• Demand-side Influence:

$$T_1^M = \frac{(1-\theta_{1,p})(a_p-1)(a_p\alpha_p^2 - c_{1,p}')}{4a_p} + \frac{(1-\theta_{1,s})(a_s-1)(a_s\alpha_s^2 - c_{1,s}')}{4a_s}$$

This represents the additional revenue for Firm 1 from providing the green variants of the two products (instead of the regular variants) to the green consumers. Since the green consumers have higher valuation for the green variants (relative to their regular counterparts), this term is always non-negative. If either (i) the green consumers do not exist (i.e., $\theta_{1,p} = 1$ and $\theta_{1,s} = 1$), or (ii) the green consumers value the green variants the same as the regular variants (i.e., $a_p = 1$; $a_s = 1$), then we have $T_1^M = 0$.

• Cost-side Influence:

1.
$$T_2^M = \frac{(c_{1,p} - c'_{1,p})(2\alpha_p - c_{1,p} - c'_{1,p})}{4} + \frac{(c_{1,s} - c'_{1,s})(2\alpha_s - c_{1,s} - c'_{1,s})}{4}$$

This represents the gain/loss from changes in the variable production costs. If the production costs of both products reduce after the implementation of the symbiotic system (i.e., $c'_{1,p} \leq c_{1,p}$ and $c'_{1,s} \leq c_{1,s}$), then this term represents a benefit for Firm 1. Otherwise, if both production costs increase, then the term reflects a loss. In general, T_2^M can be either a benefit or a loss. If the variable production costs of both products remain the same after implementation (i.e., $c_{1,p} = c'_{1,p}$ and $c_{1,s} = c'_{1,s}$), then we have $T_2^M = 0$.

2. $T_3^M = \frac{b_1(\alpha_p - c_{1,p})^2}{4(1+b_1)} + \frac{b_1(\alpha_s - c_{1,s})^2}{4(1+b_1)}$

This represents the saving for Firm 1 in the waste treatment cost. Recall that one benefit of implementation is the simplification in the treatment process of the wastes of both the products. This term is always non-negative. If the waste



treatment cost had a linear form before the implementation of the symbiotic system (i.e., $b_1 = 0$), then we have $T_3^M = 0$.

3.
$$T_4^M = K$$

This is the fixed cost incurred by Firm 1 in implementing the symbiotic system.

Our interest is in gaining insights on the conditions under which $\Delta_M \ge 0$, since this would indicate the firm's willingness to implement the system. The classification above implies the following remark.

Remark 1 In a monopoly, if the combined benefit of (i) exploiting the green market segment (T_1^M) , (ii) better usage of the waste (T_3^M) , and (iii) lowering the variable production cost $(T_2^M, \text{ if } T_2^M > 0)$ is more than the sum of the (1) higher variable production costs $(-T_2^M, \text{ if } T_2^M < 0)$ and (2) fixed cost (T_4^M) , then the firm's willingness (Δ_M) to implement the symbiotic system is nonnegative.



Figure 4.5. Willingness in a Monopoly: Influence on the Decision to Implement of the Change in the (i) Variable Production Cost after Implementation $(c'_{1,p})$, and (ii) Proportion of Regular Consumers of Product P $(\theta_{1,p})$.



Figure 4.5(a) illustrates Firm 1's combined benefit and combined cost from implementation with a change in the variable production cost after implementation $(c'_{1,p})$, while keeping all other parameters fixed. To allow us to focus on one product (say Product P), we assume $c_{1,s} = c'_{1,s}$ for simplicity. When $c'_{1,p}$ is less than the value of the variable production cost $c_{1,p}$ before implementation, we have $T_2^M \ge 0$. Thus, when $c'_{1,p} \le c_{1,p}$, the combined benefit consists of three terms: T_1^M , T_2^M , and T_3^M , while the combined cost is T_4^M . When $c'_{1,p} > c_{1,p}$, the term T_2^M is negative. Thus, the combined benefit now consists of only two terms: T_1^M and T_3^M , while the combined cost has two terms: $-T_2^M$ and T_4^M . As shown in the figure, the curve representing the combined benefit (resp., combined cost) of implementing the system decreases (resp., increases) with an increase in $c'_{1,p}$. The willingness measure Δ_M changes sign at the threshold production cost $\bar{c}_{1,p}$. As long as $c'_{1,p}$ is smaller than this threshold, the combined benefit exceeds the combined cost.

In Figure 4.5(b), we illustrate the impact of the proportion of regular consumers of Product P $(\theta_{1,p})$ on the firm's decision to implement the system. Note that (i) $c'_{1,p}$ is between 0 and α_p and (ii) the value of Δ_M decreases with an increase in $c'_{1,p}$.

First, consider the (worst-case) situation when $c'_{1,p}$ is at its upper bound α_p . In this case, the gain from exploiting the green market is positive (i.e., $T_1^M > 0$), while the firm incurs a loss from changes in the variable product costs (i.e., $T_2^M < 0$). Also, recall that the firm gains from the better usage of the wastes of the two products (i.e., $T_3^M \ge 0$). In Region I of Figure 4.5(b), where $\theta_{1,p}$ is less than a threshold $\underline{\theta}_{1,p}$, the combined benefit of T_1^M and T_3^M dominates the combined loss of $-T_2^M$ and T_4^M . Consequently, we have $\Delta_M > 0$. Since Δ_M increases with a decrease in $c'_{1,p}$, we continue to have $\Delta_M > 0$ as $c'_{1,p}$ reduces from α_p to 0.



Next, consider the (best-case) situation when $c'_{1,p}$ is at its lower bound 0. Here, the firm benefits from changes in the variable product costs (i.e., $T_2^M > 0$). Thus, the combined benefit consists of three terms T_1^M , T_2^M , and T_3^M . However, the contribution of T_1^M decreases with an increase in $\theta_{1,p}$. Therefore, in Region III of Figure 4.5(b), where $\theta_{1,p}$ is greater than a threshold $\overline{\theta}_{1,p}$, the combined cost T_4^M dominates the combined benefit of T_1^M , T_2^M , and T_3^M . Consequently, we have $\Delta_M < 0$. Again, as Δ_M decreases with an increase in $c'_{1,p}$, we continue to have $\Delta_M < 0$ as $c'_{1,p}$ increases from 0 to α_p .

In Region II, when the value of $\theta_{1,p}$ is between the two thresholds $\underline{\theta}_{1,p}$ and $\overline{\theta}_{1,p}$, there is a healthier tradeoff between the combined cost and the combined benefit. For any value of $\theta_{1,p}$ in this region, there exists a threshold of $c'_{1,p}$ (represented by the curve) below which the decision for implementation is in the affirmative.

In the next section, we consider the case under Model CR.

4.4.2 The Willingness under Competition for the Regular Variants

Recall the optimal decisions under competition (Section 4.3.4), obtained under the assumptions $\max\{2c_{1,p} - \alpha_p, 2c'_{1,p} - \alpha_p, 0\} < c_{2,p} < \min\{\frac{\alpha_p + c_{1,p}}{2}, \frac{\alpha_p + c'_{1,p}}{2}\}$ and $\max\{2c_{1,s} - \alpha_s, 2c'_{1,s} - \alpha_s, 0\} < c_{2,s} < \min\{\frac{\alpha_s + c_{1,s}}{2}, \frac{\alpha_s + c'_{1,s}}{2}\}$. Firm 1's total profit from the two products before implementation is:

$$\pi_1^* = \frac{(b_1+1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1+3)^2} + \frac{(b_1+1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1+3)^2}.$$

After implementation, the total profit is:



$$\pi_{1}^{'*} = \frac{\theta_{1,p}(\alpha_{p} - 2c_{1,p}^{'} + c_{2,p})^{2}}{9} + \frac{(1 - \theta_{1,p})(a_{p}\alpha_{p} - c_{1,p}^{'})^{2}}{4a_{p}} + \frac{\theta_{1,s}(\alpha_{s} - 2c_{1,s}^{'} + c_{2,s})^{2}}{9} + \frac{(1 - \theta_{1,s})(a_{s}\alpha_{s} - c_{1,s}^{'})^{2}}{4a_{s}} - K.$$

Let $\Delta_{CR} = \pi_1^{\prime *} - \pi_1^*$. We partition Δ_{CR} into five terms that can be conveniently interpreted:

$$\begin{split} \Delta_{CR} &= \pi_1^{'*} - \pi_1^* \\ &= (1 - \theta_{1,p}) [\frac{(a_p \alpha_p - 2c_{1,p}' + c_{2,p})^2}{9a_p} - \frac{(\alpha_p - 2c_{1,p}' + c_{2,p})^2}{9}] \\ &+ (1 - \theta_{1,p}) [\frac{(a_p \alpha_p - 2c_{1,s}' + c_{2,s})^2}{9a_s} - \frac{(\alpha_s - 2c_{1,s}' + c_{2,s})^2}{9}] \\ &+ (1 - \theta_{1,p}) [\frac{(a_p \alpha_p - c_{1,p}')^2}{4a_p} - \frac{(a_p \alpha_p - 2c_{1,p}' + c_{2,p})^2}{9a_p}] \\ &+ (1 - \theta_{1,p}) [\frac{(a_s \alpha_s - c_{1,s}')^2}{4a_s} - \frac{(a_s \alpha_s - 2c_{1,s}' + c_{2,s})^2}{9a_s}] \\ &T_2^R: \text{ The benefit of being the exclusive supplier of the green market segment} \\ &+ \frac{4(c_{1,p} - c_{1,p}')(\alpha_p - c_{1,p} - c_{1,p}' + c_{2,p})}{9} + \frac{4(c_{1,s} - c_{1,s}')(\alpha_s - c_{1,s} - c_{1,s}' + c_{2,s})}{9} \\ &T_3^R: \text{ The benefit of being the exclusive supplier of the green market segment} \\ &+ \frac{4(c_{1,p} - c_{1,p}')(\alpha_p - c_{1,p} - c_{1,p}' + c_{2,p})}{9} + \frac{4(c_{1,s} - c_{1,s}')(\alpha_s - c_{1,s} - c_{1,s}' + c_{2,s})}{9} \\ &T_3^R: \text{ The gain/loss from the changes in the variable production costs} \\ &+ \frac{b_1(16b_1 + 15)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9(4b_1 + 3)^2} + \frac{b_1(16b_1 + 15)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9(4b_1 + 3)^2} \\ &T_4^R: \text{ The gain from better usage of the waste} \end{split}$$

 T_1^R , T_3^R , T_4^R and T_5^R have similar corresponding terms $(T_1^M, T_2^M, T_3^M \text{ and } T_4^M, \text{ respectively})$ in the expression of Δ_M in Section 4.4.1. The only term unique to Δ_{CR} is T_2^R . We, therefore, discuss this expression here.

$$T_{2}^{R} = (1 - \theta_{1,p}) \left[\frac{(a_{p}\alpha_{p} - c_{1,p}^{'})^{2}}{4a_{p}} - \frac{(a_{p}\alpha_{p} - 2c_{1,p}^{'} + c_{2,p})^{2}}{9a_{p}} \right] + (1 - \theta_{1,s}) \left[\frac{(a_{s}\alpha_{s} - c_{1,s}^{'})^{2}}{4a_{s}} - \frac{(a_{s}\alpha_{s} - 2c_{1,s}^{'} + c_{2,s})^{2}}{9a_{s}} \right]$$



This term represents the additional revenue of Firm 1 for being the exclusive supplier to the green consumers. It can be shown that $T_2^R \ge 0$, with equality holding only when green consumers do not exist (i.e., $\theta_{1,p} = 1$ and $\theta_{1,s} = 1$). Under competition for the regular variants, the demand-side impact is represented by T_1^R and T_2^R while the cost-side influence is reflected in T_3^R , T_4^R , and T_5^R . This implies our next remark. The subsequent discussion offers insights on the willingness to implement.

Remark 2 Under competition for the regular variants of the two products (Model CR), if the combined benefit of (i) exploiting the green consumers (T_1^R) , (ii) being the exclusive supplier of the green market segment (T_2^R) , (iii) better usage of the waste (T_4^R) , and (iv) lowering the variable production cost (T_3^R) , if $T_3^R > 0$ exceeds the sum of the (1) higher variable production cost $(-T_3^R)$, if $T_3^R < 0$ and (2) fixed cost (T_5^R) , then the firm's willingness (Δ_{CR}) to implement the symbiotic system is nonnegative.



The Appreciation of the Green Variant of Product P (a_p)

Figure 4.6. Willingness under Competition for Regular Variants: Influence on the Decision to Implement of the Change in the Appreciation of the Green Variant of Product P (a_p) .

Figure 4.6 illustrates the impact of the appreciation a_p of the green variant of Product P on the firm's decision to implement the system. Again, note that (i) $c'_{1,p}$ is between $c'_{lb} =$



 $\max\{2c_{2,p} - \alpha_p, 0\}$ and $c'_{ub} = \frac{\alpha_p + c_{2,p}}{2}$ and (ii) the value of Δ_{CR} decreases with an increase in $c_{1,p}^{'}$. When $c_{1,p}^{'}$ is at its lower bound $c_{lb}^{'}$, the firm benefits from changes in the variable product costs (i.e., $T_3^R > 0$). Thus, the combined benefit consists of four terms T_1^R , T_2^R , T_3^R , and T_4^R . However, the contributions of T_1^R and T_2^R are marginal when a_p is close to its lower bound 1. Therefore, in Region I of Figure 4.6, where a_p is less than a threshold \underline{a}_p , the cost T_5^R dominates the combined benefit of T_1^R , T_2^R , T_3^R , and T_4^R . Consequently, we have $\Delta_{CR} < 0$. Since Δ_{CR} decreases with an increase in $c'_{1,p}$, we continue to have $\Delta_{CR} < 0$ as $c_{1,p}^{'}$ increases from $c_{lb}^{'}$ to $c_{ub}^{'}$. Now consider the situation when $c_{1,p}^{'}$ is at its upper bound $c_{ub}^{'}$. The firm incurs a loss from changes in the variable product costs (i.e., $T_3^R < 0$). However, the contributions of T_1^R and T_2^R increase with an increase of a_p . In Region III of Figure 4.6, where a_p is more than a threshold \overline{a}_p , the combined benefit of T_1^R , T_2^R , and T_4^R dominates the combined loss of $-T_3^R$ and T_5^R . Consequently, we have $\Delta_{CR} > 0$ and (since $\Delta_{CR} > 0$ increases as $c'_{1,p}$ reduces) this continues to hold as $c'_{1,p}$ reduces from c'_{ub} to c'_{lb} . As in Figure 4.6, in the intermediate region (Region II), where the value of a_p is between the two thresholds \underline{a}_p and \overline{a}_p , there exists a threshold of $c'_{1,p}$ (represented by the curve) below which the willingness to implement the symbiotic system is positive.

When competitors can produce both regular and green variants (Model CG), the behavior of the the firm's willingness to implement the symbiotic system can be analyzed in a similar manner. We, therefore, avoid providing a detailed description here and briefly list the main change in the expression of the willingness Δ_{CG} under Model CG over Δ_{CR} .

In the willingness under Model CG, there exists an additional term reflecting the access to the green market. Before implementation of the system, under Model CR, both Firm 1 and



Firm 2 provide the regular variant. Since the green variant is unavailable, Firm 1 provides the regular variant to the entire market which includes the dedicated green consumers. Under Model CG, Firm 2 produces the green variant. Since the dedicated green consumers only purchase the green variant, Firm 1 cannot access the green market before the implementation. After implementation, under Model CR, Firm 1 exclusively supplies the green variant to the green consumers. Under Model CG, Firm 1 gains access to the green market after implementation.

In the next section, our aim is to investigate how a change in the nature of competition affects the firm's willingness to implement the symbiotic system.

4.4.3 A Comparative Look at the Shift in Willingness under Competition

We first summarize Firm 1's willingness to implement the symbiotic system under Models M, CR, and CG, in the following table.

Table 4.6. Willingness to Implement the Symbiotic System Under Monopoly, Competition for the Regular Variant, and Competition for Both Regular and Green Variants.

Δ_M	$\frac{\frac{\theta_{1,p}(\alpha_p - c'_{1,p})^2}{4} + \frac{(1 - \theta_{1,p})(a_p \alpha_p - c'_{1,p})^2}{4a_p} - \frac{(\alpha_p - c_{1,p})^2}{4(1 + b_1)}}{+ \frac{\theta_{1,s}(\alpha_s - c'_{1,s})^2}{4} + \frac{(1 - \theta_{1,s})(a_s \alpha_s - c'_{1,s})^2}{4a_s} - \frac{(\alpha_s - c_{1,s})^2}{4(1 + b_1)} - K$
Δ_{CR}	$\frac{\theta_{1,p}(\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c'_{1,p})^2}{4a_p} - \frac{(b_1 + 1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1 + 3)^2} + \frac{\theta_{1,s}(\alpha_s - 2c'_{1,s} + c_{2,s})^2}{9} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - c'_{1,s})^2}{4a_s} - \frac{(b_1 + 1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1 + 3)^2} - K$
Δ_{CG}	$\frac{\theta_{1,p}(\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9a_p} - \frac{\theta_{1,p}(b_1\theta_{1,p} + 1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1\theta_{1,p} + 3)^2} + \frac{\theta_{1,s}(\alpha_s - 2c'_{1,s} + c_{2,s})^2}{9a_s} - \frac{\theta_{1,s}(b_1\theta_{1,s} + 1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1\theta_{1,s} + 3)^2} - K$

In general, with the introduction of competition, Firm 1 produces less quantities of the two products and earns less profit. Competition also hurts Firm 1's market share. For example,



if the competitor produces the green variant, then Firm 1 shares the green market after the implementation of the symbiotic system. Since one of the benefits of the implementation is to exploit the green market, competition reduces this benefit. Thus, an intuitive projection would be that Firm 1 would be less willing to implement the system under competition than under monopoly. A natural question arises: *Can the firm's willingness to implement the symbiotic system increase under competition, relative to that under a monopoly?* In Theorems 4.4.1, 4.4.3, and 4.4.5 and their corresponding corollaries, our effort is to (1) identify and interpret scenarios where competition results in an increase in willingness and (2) understand the relative impact of the nature of the competition.

Theorem 4.4.1 If $c'_{1,p} = c_{1,p}$, $c'_{1,s} = c_{1,s}$, $0 \le \theta_{1,p} \le \frac{1}{b_1+1}$, and $0 \le \theta_{1,s} \le \frac{1}{b_1+1}$, then we have $\Delta_{CR} \ge \Delta_M$.

Proof: If $c'_{1,p} = c_{1,p}$ and $c'_{1,s} = c_{1,s}$, then we have

$$\begin{split} &\Delta_M - \Delta_{CR} \\ = \frac{(\alpha_p - c_{1,p})^2}{4} (\theta_{1,p} - \frac{1}{b_1 + 1}) - \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} [\theta_{1,p} - \frac{9(b_1 + 1)}{(4b_1 + 3)^2}] \\ &+ \frac{(\alpha_s - c_{1,s})^2}{4} (\theta_{1,s} - \frac{1}{b_1 + 1}) - \frac{(\alpha_s - 2c_{1,s}' + c_{2,s})^2}{9} [\theta_{1,s} - \frac{9(b_1 + 1)}{(4b_1 + 3)^2}] \\ &= [\frac{(\alpha_p - c_{1,p})^2}{4} - \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9}] (\theta_{1,p} - \frac{1}{b_1 + 1}) \\ &+ [\frac{(\alpha_s - c_{1,s})^2}{4} - \frac{(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9}] (\theta_{1,s} - \frac{1}{b_1 + 1}) \\ &- \frac{b_1(7b_1 + 6)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9(b_1 + 1)(4b_1 + 3)^2} - \frac{b_1(7b_1 + 6)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9(b_1 + 1)(4b_1 + 3)^2} \end{split}$$

Since $\frac{(\alpha_p - c_{1,p})}{2} - \frac{(\alpha_p - 2c_{1,p} + c_{2,p})}{3} = \frac{(\alpha_p + c_{1,p} - 2c_{2,p})}{6} > 0$, we have

$$\frac{(\alpha_p - c_{1,p})^2}{4} - \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} > 0.$$



Similarly, we have

$$\frac{(\alpha_s - c_{1,s})^2}{4} - \frac{(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} > 0.$$

Also, $\frac{b_1(7b_1+6)(\alpha_p-2c_{1,p}+c_{2,p})^2}{9(b_1+1)(4b_1+3)^2} \ge 0$, and $\frac{b_1(7b_1+6)(\alpha_s-2c_{1,s}+c_{2,s})^2}{9(b_1+1)(4b_1+3)^2} \ge 0$. Thus, if $\theta_{1,p} - \frac{1}{b_1+1} \le 0$ and $\theta_{1,s} - \frac{1}{b_1+1} \le 0$, then $\Delta_M \le \Delta_{CR}$.

Corollary 4.4.2 If Firm 1's variable production costs of both products remain the same after the implementation of the symbiotic system, and the proportions of green consumers in both products' markets are more than a threshold, then competing with firms who only produce regular variants encourages the firm to implement the symbiotic system.

Theorem 4.4.3 If $c'_{1,p} = c_{1,p}$, $c'_{1,s} = c_{1,s}$, and $b_1 = 0$, then

- (i) If $a_p = 1$ and $a_s = 1$, we have $\Delta_{CG} \ge \Delta_M$.
- (*ii*) If $c_{2,p} \leq c_{1,p}$, $c_{2,s} \leq c_{1,s}$, $a_p \geq \frac{9}{5}$, and $a_s \geq \frac{9}{5}$, we have $\Delta_{CG} \leq \Delta_M$.

Proof: If $c'_{1,p} = c_{1,p}, c'_{1,s} = c_{1,s}$, and $b_1 = 0$, then

$$= \frac{(1 - \theta_{1,p})(a_p\alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} + \frac{(1 - \theta_{1,p})(\alpha_p - c_{1,p})^2}{4} - \frac{(1 - \theta_{1,p})(a_p\alpha_p - c_{1,p})^2}{4a_p} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - 2c_{1,s} + c_{2,s})^2}{9a_s} + \frac{(1 - \theta_{1,s})(\alpha_s - c_{1,s})^2}{4} - \frac{(1 - \theta_{1,s})(a_s\alpha_s - c_{1,s})^2}{4a_s} + \frac{(1 -$$

(i) If $a_p = 1$ and $a_s = 1$, then

$$\Delta_{CG} - \Delta_M = \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,s})(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} \ge 0.$$



ii) If $c_{2,p} \leq c_{1,p}$ and $c_{2,s} \leq c_{1,s}$, then

$$\begin{split} &\Delta_{CG} - \Delta_M \\ \leq \quad \frac{(1 - \theta_{1,p})(a_p \alpha_p - c_{1,p})^2}{9a_p} + \frac{(1 - \theta_{1,p})(\alpha_p - c_{1,p})^2}{4} - \frac{(1 - \theta_{1,p})(a_p \alpha_p - c_{1,p})^2}{4a_p} \\ &+ \frac{(1 - \theta_{1,s})(a_s \alpha_s - c_{1,s})^2}{9a_s} + \frac{(1 - \theta_{1,s})(\alpha_s - c_{1,s})^2}{4} - \frac{(1 - \theta_{1,s})(a_s \alpha_s - c_{1,s})^2}{4a_s} \\ = \quad \frac{(1 - \theta_{1,p})}{36a_p} [(-5a_p^2 + 9a_p)\alpha_p^2 - 8a_p \alpha_p c_{1,p} + (9a_p - 5)c_{1,p}^2] \\ &+ \frac{(1 - \theta_{1,s})}{36a_s} [(-5a_s^2 + 9a_s)\alpha_s^2 - 8a_s \alpha_s c_{1,s} + (9a_s - 5)c_{1,s}^2] \end{split}$$

If $a_p \geq \frac{9}{5}$, then $(-5a_p^2 + 9a_p) \leq 0$. Since $\alpha_p > c_{1,p}$, we have

$$\frac{(1-\theta_{1,p})}{36a_p} [(-5a_p^2+9a_p)\alpha_p^2 - 8a_p\alpha_p c_{1,p} + (9a_p-5)c_{1,p}^2]$$

$$\leq \frac{(1-\theta_{1,p})}{36a_p} [(-5a_p^2+9a_p)c_{1,p}^2 - 8a_p c_{1,p}^2 + (9a_p-5)c_{1,p}^2]$$

$$= \frac{-5(1-\theta_{1,p})(a_p-1)^2 c_{1,p}^2}{36a_p} \leq 0.$$

Similarly, we have $\frac{(1-\theta_{1,s})}{36a_s}[(-5a_s^2+9a_s)\alpha_s^2-8a_s\alpha_s c_{1,s}+(9a_s-5)c_{1,s}^2] \leq 0$. Therefore, we have $\Delta_{CG} \leq \Delta_M$.

Corollary 4.4.4 If the implementation of the symbiotic system does not change the variable production costs of both products, and the savings on the waste treatment cost are negligible, then

 (1) Even if the dedicated green consumers' appreciation for the green variants is marginal, the firm is more willing (than in a monopoly) to implement the symbiotic system when competing with firms who produce green variants.



(2) Even if the dedicated green consumers' appreciation for the green variants is higher than a threshold, competing with firms who produce green variants discourages the firm to implement the symbiotic system if the competitors have cost advantages over Firm 1 for both products.

Theorem 4.4.5 If $b_1 = 0$, $c_{2,p} \le c_{1,p} = c'_{1,p}$, and $c_{2,s} \le c_{1,s} = c'_{1,s}$, then $\Delta_{CG} < \Delta_{CR}$.

Proof: If $b_1 = 0$, $c'_{1,p} = c_{1,p}$, and $c'_{1,s} = c_{1,s}$, then we have

$$\begin{aligned} \Delta_{CG} &- \Delta_{CR} \\ = & (1 - \theta_{1,p}) \left[\frac{(a_p \alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} - \frac{(a_p \alpha_p - c_{1,p})^2}{4a_p} + \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \right] \\ &+ (1 - \theta_{1,s}) \left[\frac{(a_s \alpha_s - 2c_{1,s} + c_{2,s})^2}{9a_s} - \frac{(a_s \alpha_s - c_{1,s})^2}{4a_s} + \frac{(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} \right] \end{aligned}$$

Since $\frac{\partial \frac{(a_p \alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p}}{\partial a_p} = \frac{(a_p \alpha_p - 2c_{1,p} + c_{2,p})(a_p \alpha_p + 2c_{1,p} - c_{2,p})}{9a_p^2} > 0$, the quantity $\frac{(a_p \alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p}$ increases with an increase in a_p . Thus, for $a_p \ge 1$, we have $\frac{(a_p \alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} \ge \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9}$.

Therefore,

$$\frac{(a_p\alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} - \frac{(a_p\alpha_p - c_{1,p})^2}{4a_p} + \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ \leq \frac{2(a_p\alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} - \frac{(a_p\alpha_p - c_{1,p})^2}{4a_p}.$$

Since $c_{2,p} \leq c_{1,p}$, we have

$$\frac{2(a_p\alpha_p - 2c_{1,p} + c_{2,p})^2}{9a_p} - \frac{(a_p\alpha_p - c_{1,p})^2}{4a_p} \leq \frac{2(a_p\alpha_p - c_{1,p})^2}{9a_p} - \frac{(a_p\alpha_p - c_{1,p})^2}{4a_p}$$
$$= \frac{-(a_p\alpha_p - c_{1,p})^2}{36a_p} < 0.$$

Similarly, $\frac{(a_s\alpha_s - 2c_{1,s} + c_{2,s})^2}{9a_s} - \frac{(a_s\alpha_s - c_{1,s})^2}{4a_s} + \frac{(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} < 0$. Thus, we have $\Delta_{CG} < \Delta_{CR}$.



Corollary 4.4.6 If we suppress (i) the savings in the waste treatment cost and (ii) the savings in the variable production costs of both products, and assume that competitors have cost advantages over Firm 1 for both products, then Firm 1 is more willing to implement the symbiotic system when competitors can only produce regular variants as compared to the situation when competitors produce green variants.

4.4.4 Simultaneously Benefiting the Firm and Its Consumers

It is important to note that our comparison of willingness in Section 4.4.3 across the different settings was purely a relative one. In other words, we did not impose that the value of willingness be non-negative. Clearly, one desirable outcome in support of implementation would be that the willingness be positive. Furthermore, the motivation for implementing the system is higher if consumers (as a whole) benefit from it. In this section, our goal is to identify situations under which the above two conditions are simultaneously satisfied. We need the following additional notation.

Notation:

- $\begin{array}{ll} W^b_i & \mbox{Consumer welfare before the implementation of symbiotic system under Model} \, i, \\ & i \in \{CR, CG\}. \end{array}$
- $$\begin{split} W^a_i & \text{Consumer welfare after the implementation of symbiotic system under Model} \ i, \\ i \in \{CR, CG\}. \end{split}$$
- $p_{i,j}^b$ Market price of variant *i* of Product *j* before the implementation of symbiotic system, $i \in \{r, g\}$, *r* (resp., *g*) represents regular (resp., green); $j \in \{p, s\}$.
- $\begin{array}{ll} p_{i,j}^a & \text{Market price of variant } i \text{ of Product } j \text{ after the implementation of symbiotic} \\ & \text{system, } i \in \{r,g\}, \ r \ (\text{resp., } g) \text{ represents regular (resp., green); } j \in \{p,s\}. \end{array}$

We first consider competition only for the regular variants (Model CR). The equilibrium prices and production quantities for both the variants of a product (say, Product P) are listed in Table 4.7.



Table 4.7. Equilibrium Results under Competition for Regular Variants, both Before and After the Implementation of the Symbiotic System. Here, q_i^* is the Total Production Quantity of Firm i; i = 1, 2.

Model	Before Implementation	After Implementation	
CR	Regular Variant	Regular Variant	Green Variant
p^*	$\frac{(2b_1+1)\alpha + c_1 + (2b_1+1)c_2}{4b_1+3}$	$\frac{\alpha + c_1' + c_2}{3}$	$\frac{a\alpha + c_1'}{2}$
$q^* = q_1^* + q_2^*$	$\frac{(2b_1+2)\alpha - c_1 - (2b_1+1)c_2}{4b_1+3}$	$\frac{\theta_1(2\alpha - c_1' - c_2)}{3}$	$\frac{(1-\theta_1)(a\alpha - c_1')}{2a}$

Recall from Section 4.2 that consumers' valuation of the regular variant is uniformly distributed between 0 and α with density 1. Therefore, we have

$$W_{CR}^{b} = \int_{p_{r,p}^{b}}^{\alpha_{p}} [1(v_{p} - p_{r,p}^{b})] dv_{p} + \int_{p_{r,s}^{b}}^{\alpha_{s}} [1(v_{s} - p_{r,s}^{b})] dv_{s} = \frac{(\alpha_{p} - p_{r,p}^{b})^{2}}{2} + \frac{(\alpha_{s} - p_{r,s}^{b})^{2}}{2} \\ = \frac{[(2b_{1} + 2)\alpha_{p} - c_{1,p} - (2b_{1} + 1)c_{2,p}]^{2}}{2(4b_{1} + 3)^{2}} + \frac{[(2b_{1} + 2)\alpha_{s} - c_{1,s} - (2b_{1} + 1)c_{2,s}]^{2}}{2(4b_{1} + 3)^{2}}.$$

Similarly, we have

$$\begin{split} W_{CR}^{a} &= \int_{p_{r,p}^{a}}^{\alpha_{p}} [\theta_{1,p}(v_{p} - p_{r,p}^{a})] \, \mathrm{d}v_{p} + \int_{\frac{p_{g,p}^{a}}{a_{p}}}^{\alpha_{p}} [(1 - \theta_{1,p})(a_{p}v_{p} - p_{g,p}^{a})] \, \mathrm{d}v_{p} \\ &+ \int_{p_{r,s}^{a}}^{\alpha_{s}} [\theta_{1,s}(v_{s} - p_{r,s}^{a})] \, \mathrm{d}v_{s} + \int_{\frac{p_{g,s}^{a}}{a_{s}}}^{\alpha_{s}} [(1 - \theta_{1,s})(a_{s}v_{s} - p_{g,s}^{a})] \, \mathrm{d}v_{s} \\ &= \frac{\theta_{1,p}(\alpha_{p} - p_{r,p}^{a})^{2}}{2} + \frac{(1 - \theta_{1,p})(a_{p}\alpha_{p} - p_{g,p}^{a})^{2}}{2a_{p}} \\ &+ \frac{\theta_{1,s}(\alpha_{s} - p_{r,s}^{a})^{2}}{2} + \frac{(1 - \theta_{1,s})(a_{s}\alpha_{s} - p_{g,s}^{a})^{2}}{2a_{s}} \\ &= \frac{\theta_{1,p}(2\alpha_{p} - c_{1,p}^{'} - c_{2,p})^{2}}{18} + \frac{(1 - \theta_{1,p})(a_{p}\alpha_{p} - c_{1,p}^{'})^{2}}{8a_{p}} \\ &+ \frac{\theta_{1,s}(2\alpha_{s} - c_{1,s}^{'} - c_{2,s})^{2}}{18} + \frac{(1 - \theta_{1,s})(a_{s}\alpha_{s} - c_{1,s}^{'})^{2}}{8a_{s}}. \end{split}$$

The following result identifies conditions under which consumer welfare improves after the implementation of the system.



Theorem 4.4.7 If $a_p \ge 2$, $a_s \ge 2$, $c'_{1,p} \le c_{1,p}$, and $c'_{1,s} \le c_{1,s}$, then we have $W^a_{CR} > W^b_{CR}$.

Proof: Since
$$\frac{\partial \frac{(1-\theta_{1,p})(a_p\alpha_p-c_{1,p})^2}{8a_p}}{\partial a_p} = \frac{(1-\theta_{1,p})(a_p\alpha_p-c_{1,p}')(a_p\alpha_p+c_{1,p}')}{8a_p^2} > 0$$
, the quantity $\frac{(1-\theta_{1,p})(a_p\alpha_p-c_{1,p}')^2}{8a_p}$

increases with an increase in a_p . Thus, for $a_p \ge 2$, we have

$$\frac{(1-\theta_{1,p})(a_p\alpha_p - c'_{1,p})^2}{8a_p} \geq \frac{(1-\theta_{1,p})(2\alpha_p - c'_{1,p})^2}{16} \\ > \frac{(1-\theta_{1,p})(2\alpha_p - c'_{1,p} - c_{2,p})^2}{18}.$$

Therefore,

$$\frac{\theta_{1,p}(2\alpha_p - c_{1,p}^{'} - c_{2,p})^2}{18} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c_{1,p}^{'})^2}{8a_p}$$

$$> \frac{\theta_{1,p}(2\alpha_p - c_{1,p}^{'} - c_{2,p})^2}{18} + \frac{(1 - \theta_{1,p})(2\alpha_p - c_{1,p}^{'} - c_{2,p})^2}{18} = \frac{(2\alpha_p - c_{1,p}^{'} - c_{2,p})^2}{18}.$$

Next, we show $\frac{[(2b_1+2)\alpha_p - c_{1,p} - (2b_1+1)c_{2,p}]^2}{2(4b_1+3)^2}$ reaches its maximum at $b_1 = 0$.

$$\begin{aligned} &\frac{[2\alpha_p - c_{1,p} - 2c_{2,p}]^2}{18} - \frac{[(2b_1 + 2)\alpha_p - c_{1,p} - (2b_1 + 1)c_{2,p}]^2}{2(4b_1 + 3)^2} \\ &= \frac{1}{18(4b_1 + 3)^2} \{ [(4b_1 + 3)(2\alpha_p - c_{1,p} - c_{2,p})]^2 - [3(2b_1 + 2)\alpha_p - 3c_{1,p} - 3(2b_1 + 1)c_{2,p}]^2 \} \\ &= \frac{2b_1(\alpha_p - 2c_{1,p} + c_{2,p})}{18(4b_1 + 3)^2} [(14b_1 + 12)\alpha_p - (4b_1 + 6)c_{1,p} - (10b_1 + 6)c_{2,p}] \ge 0. \end{aligned}$$

Since $c'_{1,p} \leq c_{1,p}$, we have

$$\frac{\left[(2b_1+2)\alpha_p - c_{1,p} - (2b_1+1)c_{2,p}\right]^2}{2(4b_1+3)^2} \le \frac{(2\alpha_p - c_{1,p} - 2c_{2,p})^2}{18} \le \frac{(2\alpha_p - c_{1,p}' - 2c_{2,p})^2}{18}.$$

Therefore,

$$\frac{\theta_{1,p}(2\alpha_p - c_{1,p}^{'} - c_{2,p})^2}{18} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c_{1,p}^{'})^2}{8a_p} > \frac{[(2b_1 + 2)\alpha_p - c_{1,p} - (2b_1 + 1)c_{2,p}]^2}{2(4b_1 + 3)^2}$$

Similarly,

$$\frac{\theta_{1,s}(2\alpha_{s}-c_{1,s}^{'}-c_{2,s})^{2}}{18} + \frac{(1-\theta_{1,s})(a_{s}\alpha_{s}-c_{1,s}^{'})^{2}}{8a_{s}} > \frac{[(2b_{1}+2)\alpha_{s}-c_{1,s}-(2b_{1}+1)c_{2,s}]^{2}}{2(4b_{1}+3)^{2}}$$

Thus, we have $W_{CR}^a > W_{CR}^b$.



Corollary 4.4.8 When Firm 1 competes with firms who only produce regular variants, if the green consumers' appreciation for the green variant is relatively high, and the implementation of the symbiotic system does not increase the variable production costs of both products, then consumer welfare increases after the implementation.

The result in Theorem 4.4.7 can be used to identify a special case in which both Firm 1 and consumers benefit from the implementation of the symbiotic system.

Theorem 4.4.9 If $a_p \ge 2$, $a_s \ge 2$, $c'_{1,p} \le c_{1,p}$, $c'_{1,s} \le c_{1,s}$, and $K \le \frac{(1-\theta_{1,p})}{8}[(2\alpha_p - c_{1,p})^2 - (\alpha_p - 2c_{1,p} + c_{2,p})^2] + \frac{(1-\theta_{1,s})}{8}[(2\alpha_s - c_{1,s})^2 - (\alpha_s - 2c_{1,s} + c_{2,s})^2]$, then we have $W^a_{CR} > W^b_{CR}$ and $\Delta_{CR} > 0$.

Proof: Since $a_p \geq 2$, $a_s \geq 2$, $c'_{1,p} \leq c_{1,p}$, and $c'_{1,s} \leq c_{1,s}$, we have $W^a_{CR} > W^b_{CR}$ from Theorem 4.4.7.

By using an approach similar to that in the proof of Theorem 12, we can show the following two properties:

- (i) $\frac{(b_1+1)(\alpha_p-2c_{1,p}+c_{2,p})^2}{(4b_1+3)^2}$ reaches its maximum when $b_1 = 0$,
- (ii) $\frac{(1-\theta_{1,p})(a_p\alpha_p-c'_{1,p})^2}{4a_p}$ increases with an increase in a_p .

Thus, for $a_p \ge 2$, we have

$$\frac{(1-\theta_{1,p})(a_p\alpha_p - c_{1,p}')^2}{4a_p} \geq \frac{(1-\theta_{1,p})(2\alpha_p - c_{1,p})^2}{8}$$
$$\frac{(b_1+1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1+3)^2} \leq \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9}.$$



Thus,

$$\begin{aligned} \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c_{1,p}')^2}{4a_p} - \frac{(b_1 + 1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1 + 3)^2} \\ \ge & \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(2\alpha_p - c_{1,p})^2}{8} - \frac{(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ = & \frac{(1 - \theta_{1,p})(2\alpha_p - c_{1,p})^2}{8} - \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ \ge & \frac{(1 - \theta_{1,p})(2\alpha_p - c_{1,p})^2}{8} - \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{8} \\ = & \frac{(1 - \theta_{1,p})}{8} [(2\alpha_p - c_{1,p})^2 - (\alpha_p - 2c_{1,p} + c_{2,p})^2] \\ = & \frac{(1 - \theta_{1,p})}{8} [(\alpha_p + c_{1,p} - c_{2,p})(3\alpha_p - 3c_{1,p} + c_{2,p})] > 0. \end{aligned}$$

Similarly, we have

$$\frac{\theta_{1,s}(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - c_{1,s}')^2}{4a_s} - \frac{(b_1 + 1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1 + 3)^2}$$

$$\geq \frac{(1 - \theta_{1,s})}{8} [(2\alpha_s - c_{1,s})^2 - (\alpha_s - 2c_{1,s} + c_{2,s})^2].$$

Thus, if $K \leq \frac{(1-\theta_{1,p})}{8} [(2\alpha_p - c_{1,p})^2 - (\alpha_p - 2c_{1,p} + c_{2,p})^2] + \frac{(1-\theta_{1,s})}{8} [(2\alpha_s - c_{1,s})^2 - (\alpha_s - 2c_{1,s} + c_{2,s})^2],$ then we have $\Delta_{CR} > 0$. The result follows.

Corollary 4.4.10 If (i) green consumers' appreciation of the green variant is relatively high, (ii) the implementation of the symbiotic system reduces the variable production costs of both products, and (iii) the fixed cost of implementation of the symbiotic system is modest, then the willingness of Firm 1 to implement the symbiotic system under competition for the regular variants is positive. Also, consumer welfare increases after the implementation.

To extend the line of thought in Theorem 4.4.9, note that a further motivation for implementation occurs when the willingness of Firm 1 in a monopoly is negative. The following result illustrates one such situation.



Theorem 4.4.11 If $a_p = 2$, $a_s = 2$, $c_{2,p} \leq c_{1,p} = c'_{1,p}$, $c_{2,s} \leq c_{1,s} = c'_{1,s}$, $b_1 = 0$, and $K = \frac{(1-\theta_{1,p})}{8} [(2\alpha_p - c_{1,p})^2 - (\alpha_p - 2c_{1,p} + c_{2,p})^2] + \frac{(1-\theta_{1,s})}{8} [(2\alpha_s - c_{1,s})^2 - (\alpha_s - 2c_{1,s} + c_{2,s})^2],$ then we have $W^a_{CR} > W^b_{CR}$, $\Delta_M < 0$, and $\Delta_{CR} > 0$.

Proof: From Theorem 4.4.9, we have $W_{CR}^a > W_{CR}^b$ and $\Delta_{CR} > 0$. Since $a_p = 2$, $a_s = 2$, $c'_{1,p} = c_{1,p}, c'_{1,s} = c_{1,s}, b_1 = 0$, we have

$$\begin{split} \Delta_M &= \frac{\theta_{1,p}(\alpha_p - c_{1,p}')^2}{4} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c_{1,p}')^2}{4a_p} - \frac{(\alpha_p - c_{1,p})^2}{4(1 + b_1)} \\ &+ \frac{\theta_{1,s}(\alpha_s - c_{1,s}')^2}{4} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - c_{1,s}')^2}{4a_s} - \frac{(\alpha_s - c_{1,s})^2}{4(1 + b_1)} - K \\ &= \frac{\theta_{1,p}(\alpha_p - c_{1,p})^2}{4} + \frac{(1 - \theta_{1,p})(2\alpha_p - c_{1,p})^2}{8} - \frac{(\alpha_p - c_{1,p})^2}{4} \\ &+ \frac{\theta_{1,s}(\alpha_s - c_{1,s})^2}{4} + \frac{(1 - \theta_{1,s})(2\alpha_s - c_{1,s})^2}{8} - \frac{(\alpha_s - c_{1,s})^2}{4} - K \\ &= \frac{(1 - \theta_{1,p})[(2\alpha_p - c_{1,p})^2 - 2(\alpha_p - c_{1,p})^2]}{8} + \frac{(1 - \theta_{1,s})[(2\alpha_s - c_{1,s})^2 - 2(\alpha_s - 2c_{1,s})^2]}{8} - K. \end{split}$$

Using
$$K = \frac{(1-\theta_{1,p})}{8} [(2\alpha_p - c_{1,p})^2 - (\alpha_p - 2c_{1,p} + c_{2,p})^2] + \frac{(1-\theta_{1,s})}{8} [(2\alpha_s - c_{1,s})^2 - (\alpha_s - 2c_{1,s} + c_{2,s})^2]$$
, we get $\Delta_M = \frac{(1-\theta_{1,p})}{8} [(\alpha_p - 2c_{1,p} + c_{2,p})^2 - 2(\alpha_p - c_{1,p})^2] + \frac{(1-\theta_{1,s})}{8} [(\alpha_s - 2c_{1,s} + c_{2,s})^2 - 2(\alpha_s - 2c_{1,s})^2]$.

Also, $c_{2,p} \leq c_{1,p}, c_{2,s} \leq c_{1,s}$. Therefore,

$$\begin{split} \Delta_M &= \frac{(1-\theta_{1,p})}{8} [(\alpha_p - 2c_{1,p} + c_{2,p})^2 - 2(\alpha_p - c_{1,p})^2] \\ &+ \frac{(1-\theta_{1,s})}{8} [(\alpha_s - 2c_{1,s} + c_{2,s})^2 - 2(\alpha_s - 2c_{1,s})^2] \\ &< \frac{(1-\theta_{1,p})}{8} [(\alpha_p - 2c_{1,p} + c_{2,p})^2 - (\alpha_p - c_{1,p})^2] \\ &+ \frac{(1-\theta_{1,s})}{8} [(\alpha_s - 2c_{1,s} + c_{2,s})^2 - (\alpha_s - 2c_{1,s})^2] \\ &= \frac{(1-\theta_{1,p})(c_{2,p} - c_{1,p})}{8} [(\alpha_p - 2c_{1,p} + c_{2,p}) + (\alpha_p - c_{1,p})] \\ &+ \frac{(1-\theta_{1,s})(c_{2,s} - c_{1,s})}{8} [(\alpha_s - 2c_{1,s} + c_{2,s}) + (\alpha_s - 2c_{1,s})] \le 0. \end{split}$$

The result follows.



Corollary 4.4.12 If (1) green consumers appreciate the green variant about two times the regular variant, (2) implementation of the symbiotic system does not change the variable production costs of both products, (3) competitors have a cost advantage over Firm 1, (4) savings in the waste treatment cost are negligible, and (5) the fixed cost of implementation of the symbiotic system is modest, then (i) willingness of Firm 1 to implement is negative in a monopoly, but positive under competition for the regular variants and (ii) consumer welfare increases after the implementation.



Figure 4.7. A Pictorial Representation of the Conditions in Theorems 4.4.7, 4.4.9, and 4.4.11.

The nested relationships among the results in Theorems 4.4.7, 4.4.9, and 4.4.11, are illustrated in Figure 4.7. The conditions under which $W_{CR}^a > W_{CR}^b$ are represented by the region $A_1 \cup A_2 \cup A_3 \cup A_4$; one such condition forms the premise of Theorem 4.4.7. Similarly, the conditions under which $\Delta_{CR} > 0$ (resp., $\Delta_M < 0$) are represented by the region $A_2 \cup A_3 \cup A_5 \cup A_6$ (resp., $A_3 \cup A_4 \cup A_6 \cup A_7$). Thus, the condition assumed in Theorem 4.4.9 belongs to $A_2 \cup A_3$, while that in Theorem 4.4.11 belongs to Region A_3 .



Next, consider competition for both regular and green variants (Model CG). The equilibrium prices and production quantities for both the variants of a product (say, Product P) are listed in Table 4.8.

Table 4.8. Equilibrium Results under Competition for both Regular and Green Variants, both before and after the Implementation of the Symbiotic System.

Model	Before Implement	ntation	After Implementation	
CG	Regular Variant	Green Variant	Regular Variant	Green Variant
p^*	$\frac{(2b_1\theta_1+1)\alpha + c_1 + (2b_1\theta_1+1)c_2}{3+4b_1\theta_1}$	$\frac{a\alpha+c_2}{2}$	$\frac{\alpha + c_1' + c_2}{3}$	$\frac{a\alpha + c_1' + c_2}{3}$
$q_1^* + q_2^*$	$\frac{\theta_1[(2b_1\theta_1+2)\alpha - c_1 - (2b_1\theta_1+1)c_2]}{3+4b_1\theta_1}$	$\frac{(1-\theta_1)(a\alpha-c_2)}{2a}$	$\frac{\theta_1(2\alpha - c_1' - c_2)}{3}$	$\frac{(1-\theta_1)(2a\alpha - c_1' - c_2)}{3a}$

We first derive consumer surplus before and after the implementation of a symbiotic system.

$$\begin{split} W^b_{CG} &= \frac{\theta_{1,p}(\alpha_p - p^b_{r,p})^2}{2} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - p^b_{g,p})^2}{2a_p} \\ &+ \frac{\theta_{1,s}(\alpha_s - p^b_{r,s})^2}{2} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - p^b_{g,s})^2}{2a_s}, \\ W^a_{CG} &= \frac{\theta_{1,p}(\alpha_p - p^a_{r,p})^2}{2} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - p^a_{g,p})^2}{2a_p} \\ &+ \frac{\theta_{1,s}(\alpha_s - p^a_{r,s})^2}{2} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - p^a_{g,s})^2}{2a_s}. \end{split}$$

As with Model CR, our effort here is to identify conditions that benefit both Firm 1 and consumers. Theorem 4.4.13 is used to establish Theorem 4.4.15, which identifies such conditions.

Theorem 4.4.13 If $c'_{1,p} \leq c_{1,p}$ and $c'_{1,s} \leq c_{1,s}$, then we have $W^{a}_{CG} > W^{b}_{CG}$.

Proof: The consumer welfare before and after the implementation are:

$$W_{CR}^{b} = \frac{\theta_{1,p}(\alpha_{p} - p_{r,p}^{b})^{2}}{2} + \frac{(1 - \theta_{1,p})(a_{p}\alpha_{p} - p_{g,p}^{b})^{2}}{2a_{p}} + \frac{\theta_{1,s}(\alpha_{s} - p_{r,s}^{b})^{2}}{2} + \frac{(1 - \theta_{1,s})(a_{s}\alpha_{s} - p_{g,s}^{b})^{2}}{2a_{s}}$$



$$\begin{split} W^{a}_{CR} &= \frac{\theta_{1,p} (\alpha_{p} - p^{a}_{r,p})^{2}}{2} + \frac{(1 - \theta_{1,p}) (a_{p} \alpha_{p} - p^{a}_{g,p})^{2}}{2a_{p}} \\ &+ \frac{\theta_{1,s} (\alpha_{s} - p^{a}_{r,s})^{2}}{2} + \frac{(1 - \theta_{1,s}) (a_{s} \alpha_{s} - p^{a}_{g,s})^{2}}{2a_{s}} \end{split}$$

Thus, if we can show $p_{r,p}^b \ge p_{r,p}^a$, $p_{g,p}^b \ge p_{g,p}^a$, $p_{r,s}^b \ge p_{r,s}^a$, and $p_{g,s}^b \ge p_{g,s}^a$, then we have $W_{CG}^a > W_{CG}^b$. We first compare the prices of the regular variants of Product P before and after the implementation.

$$p_{r,p}^{b} - p_{r,p}^{a} = \frac{(2b_{1}\theta_{1,p} + 1)\alpha_{p} + c_{1,p} + (2b_{1}\theta_{1,p} + 1)c_{2,p}}{3 + 4b_{1}\theta_{1,p}} - \frac{\alpha_{p} + c_{1,p}^{'} + c_{2,p}}{3}$$
$$= \frac{2b_{1}\theta_{1,p}\alpha_{p} + 3c_{1,p} - (3 + 4b_{1}\theta_{1,p})c_{1,p}^{'} + 2b_{1}\theta_{1,p}c_{2,p}}{3(3 + 4b_{1}\theta_{1,p})}.$$

Since $c'_{1,p} \leq c_{1,p}$, we have

$$p_{r,p}^{b} - p_{r,p}^{a} \geq \frac{2b_{1}\theta_{1,p}\alpha_{p} + 3c_{1,p} - (3 + 4b_{1}\theta_{1,p})c_{1,p} + 2b_{1}\theta_{1,p}c_{2,p}}{3(3 + 4b_{1}\theta_{1,p})} \\ = \frac{2b_{1}\theta_{1,p}(\alpha_{p} - 2c_{1,p} + c_{2,p})}{3(3 + 4b_{1}\theta_{1,p})} \geq 0.$$

Next, we compare the prices of the green variants of Product P before and after the implementation. $p_{g,p}^b - p_{g,p}^a = \frac{a_p \alpha_p + c_{2,p}}{2} - \frac{a_p \alpha_p + c_{1,p}' + c_{2,p}}{3} = \frac{a_p \alpha_p - 2c_{1,p}' + c_{2,p}}{6} \ge 0.$

Similarly, for Product S, we have $p_{r,s}^b \ge p_{r,s}^a$ and $p_{g,s}^b \ge p_{g,s}^a$. Thus, $W_{CG}^a > W_{CG}^b$.

Corollary 4.4.14 When competitors produce both regular and green variants, if the implementation of the symbiotic system reduces the variable production costs of both products, then consumer welfare increases after the implementation.

Theorem 4.4.15 If $K \leq \frac{(1-\theta_{1,p})(a_p\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9a_p} + \frac{(1-\theta_{1,s})(a_s\alpha_s - 2c'_{1,s} + c_{2,s})^2}{9a_s}$, $c'_{1,p} \leq c_{1,p}$, and $c'_{1,s} \leq c_{1,s}$, then we have $W^a_{CG} > W^b_{CG}$ and $\Delta_{CG} > 0$.



Proof: From Theorem 4.4.13, we have $W_{CG}^a > W_{CG}^b$. Using an approach similar to that in the proof of Theorem 4.4.9, we can establish the following two properties:

(i)
$$\frac{\theta_{1,p}(b_1\theta_{1,p}+1)(\alpha_p-2c_{1,p}+c_{2,p})^2}{(4b_1\theta_{1,p}+3)^2}$$
 reaches its maximum at $b_1 = 0$,

(ii)
$$\frac{(1-\theta_{1,p})(a_p\alpha_p-2c'_{1,p}+c_{2,p})^2}{9a_p}$$
 increases with an increase in a_p .

Then, for $a_p \ge 1$, $c'_{1,p} \le c_{1,p}$, we have

$$\begin{aligned} \frac{\theta_{1,p}(\alpha_p - 2c_{1,p}' + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - 2c_{1,p}' + c_{2,p})^2}{9a_p} \\ - \frac{\theta_{1,p}(b_1\theta_{1,p} + 1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1\theta_{1,p} + 3)^2} \\ \ge \quad \frac{\theta_{1,p}(\alpha_p - 2c_{1,p}' + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p}' + c_{2,p})^2}{9} - \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ \ge \quad \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} - \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ \ge \quad \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} - \frac{\theta_{1,p}(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} \\ = \quad \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} > 0. \end{aligned}$$

Similarly, we have

$$\frac{\theta_{1,s}(\alpha_s - 2c_{1,s}^{'} + c_{2,s})^2}{9} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - 2c_{1,s}^{'} + c_{2,s})^2}{9a_s} - \frac{\theta_{1,s}(b_1\theta_{1,s} + 1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1\theta_{1,s} + 3)^2} \ge \frac{(1 - \theta_{1,s})(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9}.$$

Thus,

$$\begin{split} \Delta_{CG} &= \frac{\theta_{1,p}(\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - 2c'_{1,p} + c_{2,p})^2}{9a_p} \\ &- \frac{\theta_{1,p}(b_1\theta_{1,p} + 1)(\alpha_p - 2c_{1,p} + c_{2,p})^2}{(4b_1\theta_{1,p} + 3)^2} + \frac{\theta_{1,s}(\alpha_s - 2c'_{1,s} + c_{2,s})^2}{9} \\ &+ \frac{(1 - \theta_{1,s})(a_s\alpha_s - 2c'_{1,s} + c_{2,s})^2}{9a_s} - \frac{\theta_{1,s}(b_1\theta_{1,s} + 1)(\alpha_s - 2c_{1,s} + c_{2,s})^2}{(4b_1\theta_{1,s} + 3)^2} - K \\ \geq \frac{(1 - \theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1 - \theta_{1,s})(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} \\ &- K. \end{split}$$

المنارات المستشارات

Therefore, if $K \leq \frac{(1-\theta_{1,p})(\alpha_p-2c_{1,p}+c_{2,p})^2}{9} + \frac{(1-\theta_{1,s})(\alpha_s-2c_{1,s}+c_{2,s})^2}{9}$, then we have $\Delta_{CG} > 0$. The result follows.

Corollary 4.4.16 If (i) the implementation of the symbiotic system reduces the variable production costs of both products and (ii) the fixed cost of implementation of the symbiotic system is modest, then the willingness of Firm 1 to implement the symbiotic system under competition for both regular and green variants is positive. Also, consumer welfare increases after the implementation.

The following result extends Theorem 4.4.15 by further requiring that the willingness of implementation in a monopoly be negative.

Theorem 4.4.17 If $a_p = 1$, $a_s = 1$, $c'_{1,p} = c_{1,p}$, $c'_{1,s} = c_{1,s}$, $b_1 = 0$, and $\frac{(1-\theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9}$ $+ \frac{(1-\theta_{1,s})(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9} \ge K > 0$, then we have $W^a_{CG} > W^b_{CG}$ and $\Delta_{CG} > 0 > \Delta_M$.

Proof: Since $c'_{1,p} \leq c_{1,p}$, $c'_{1,s} \leq c_{1,s}$, and $K \leq \frac{(1-\theta_{1,p})(\alpha_p - 2c_{1,p} + c_{2,p})^2}{9} + \frac{(1-\theta_{1,s})(\alpha_s - 2c_{1,s} + c_{2,s})^2}{9}$, then from Theorem 4.4.15, we have $W^a_{CG} > W^b_{CG}$ and $\Delta_{CG} > 0$. If $a_p = 1$, $a_s = 1$, $c'_{1,p} = c_{1,p}$, $c'_{1,s} = c_{1,s}$, $b_1 = 0$, we have

$$\begin{split} \Delta_M &= \frac{\theta_{1,p}(\alpha_p - c'_{1,p})^2}{4} + \frac{(1 - \theta_{1,p})(a_p\alpha_p - c'_{1,p})^2}{4a_p} - \frac{(\alpha_p - c_{1,p})^2}{4(1 + b_1)} \\ &+ \frac{\theta_{1,s}(\alpha_s - c'_{1,s})^2}{4} + \frac{(1 - \theta_{1,s})(a_s\alpha_s - c'_{1,s})^2}{4a_s} - \frac{(\alpha_s - c_{1,s})^2}{4(1 + b_1)} - K \\ &= \frac{\theta_{1,p}(\alpha_p - c_{1,p})^2}{4} + \frac{(1 - \theta_{1,p})(\alpha_p - c_{1,p})^2}{4} - \frac{(\alpha_p - c_{1,p})^2}{4} \\ &+ \frac{\theta_{1,s}(\alpha_s - c_{1,s})^2}{4} + \frac{(1 - \theta_{1,s})(\alpha_s - c_{1,s})^2}{4} - \frac{(\alpha_s - c_{1,s})^2}{4} - K \\ &= -K. \end{split}$$

Thus, for any K > 0, we have $\Delta_M < 0$. The result follows.

🖄 للاستشارات

186

Corollary 4.4.18 If (i) the green consumers' appreciation for the green variant is negligible, (ii) the implementation of the symbiotic system does not change the variable production costs of both products, and (iii) the savings in the waste treatment costs are negligible, then Firm 1 is more willing to implement the symbiotic system under the competition with firms who produce both regular and green variants. The consumer welfare also increases after the implementation.

To summarize, Theorems 4.4.7–4.4.17 and Corollaries 4.4.8–4.4.18 together, suggest that competition for either the regular variant or both regular and green variants can indeed shift a firm's willingness from negative to positive, and improve consumer welfare as well. This perhaps explains the timing of SPB's decision to implement the paper-sugar complex (see Section 4.1.1).

4.5 Directions for Future Work

Going back to the real-world implementation described in Section 4.1.1, the unique symbiotic model followed by SPB has transformed the livelihood of the local farming community and has provided a reliable supply of raw material for both SPB and Ponni Sugars. The rain shadow region around the paper-sugar complex has witnessed a green revolution, through the irrigation of more than 1500 acres of dry and fallow land with treated effluent. For the government, the positive societal impact of this implementation provides an ideal case study to identify and incentivize similar symbiotic systems. Some examples of possible incentives structures that can be studied include (1) providing a one-time subsidy, (2) providing longterm tax relief, (3) subsidizing the supply of electricity and water, and (4) improving societal



consciousness of symbiotic initiatives. To cite an instance, the recycling facilities established through the Eco-Town Program in Kawasakia, Japan, received an average investment subsidy of 48% from the government (Van Berkel et al. 2009).

In the models analyzed in this paper, we considered two products that are symbiotically connected. A much-larger example of Burnside Industrial Park, which involves about 1300 businesses, is described in Noronha (1999). The businesses within the industrial park are involved in (1) scavenger roles (reuse, remanufacture, refurbish, repair and recover), (2) decomposer roles (recycling), (3) producing/selling environment-friendly products, and (4) providing environmental management services. These companies deal in a diverse range of materials and have established complex relationships within the park. This example motivates the notion of *Design for Symbiosis*: problems of designing the layout of a symbiotic system, coordinating the material exchanges between the firms, and scheduling the logistics activities.

For the two-product symbiotic system analyzed in this paper, we assumed that the production decisions of both the products are taken by a common firm interested in maximizing its total profit. Under this centralized setting, even if the implementation of the system hurts the profit from one product, the firm may be willing to implement if total profit improves. Once the system is implemented, the complete exchange of wastes between the two production processes is guaranteed. However, if the products are manufactured by two independent firms, then participation in a symbiotic relationship becomes voluntary. Thus, in a decentralized setting, the problem of pricing the wastes becomes relevant. On the one hand, if the price of one type of waste is too low, then the firm that generates the waste may be less willing to supply it to the other firm (that uses this waste as a raw material). On the other



hand, if this price is excessive, then the firm that can potentially purchase the waste may switch to an alternative source of raw material. Thus, the pricing of the wastes affects not only the revenues of the firms involved but also their operations.

Private Communications (2008 – 2011)

Gopalaratnam, N., Chairman and Managing Director, SPB Ltd.

Kannan, S., President, SPB Ltd.

Chandramouli, R., President and Chief Operating Officer, Ponni Sugars.

Arivalagan, A., Assistant General Manager (Technical Services), SPB Ltd.

Srinivasan, S., Vice President (Process), SPB Ltd.

Ponnuswamy, K., General Manager (Control Systems), SPB Ltd.

Shanmugan, K., Director, World Class Manufacturing Program & General Manager (Environment), SPB Ltd.

Balasubramanian, C., Vice President (Business Development), SPB Ltd.



REFERENCES

- Allen, D., N. Behmanish. 1994. Wastes as Raw Materials. B.R. Allenby and D. Richards, eds. *The Greening of Industrial Ecosystems*. National Academy Press, Washington, DC.
- Allenby, B. R. 1992. Achieving Sustainable Development through Industrial Ecology. International Environmental Affairs 4(1) 56-68.
- Arrow, K. J. 1962. Economic Welfare and the Allocation of Resources for Invention. R. R. Nelson, eds. *The Rate and Direction of Inventive Activity*. Princeton University Press, Princeton, NJ, 609-625.
- Atasu, A., L. N. Van Wassenhove, M. Sarvary. 2009. Efficient Take-Back Legislation. Production and Operations Management 18 243258.
- Balintfy, J. 1964. On a basic class of multi-item inventory problems. *Management Science* **10**(2) 287-297.
- Bequette, F. 1997. A Waste-Free Farm. UNESCO Courier September 44-46.
- Blacketer, B., B. Evetts. 2004. Forging your bank's link in the new currency supply chain. Accessed: March 2010. http://www.carreker.fiserv.com/main/media/art_wtpapers/2004/cash_supply_wp _may2004.pdf.
- Block, D. 1998. Ecoindustrial Parks Are Ready to Take Off. In Business Sept/Oct 18-20.
- Bonanno, G., B. Haworth. 1998. Intensity of Competition and the Choice Between Product and Process Innovation. *International Journal of Industrial Organization* **16** 495510.
- Borgatti, S. P. 2006. Identifying Sets of Key Players in a Social Network, Computational and Mathematical Organization Theory, 12, 21-34.
- Boyd, D. M., N. B. Ellison. 2007. Social Network Sites: Definition, History, and Scholarship, Journal of Computer-Mediated Communication, 13(1), 210-230.



- Brandes, U., T. Erlebach. 2005. Network Analysis Methodological Foundations, Springer Verlag.
- Carrington, P. J., J. Scott, S. Wasserman. 2005. Models and Methods in Social Network Analysis, Cambridge University Press.
- Cattani, G., S. Ferriani. 2008. A Core/Periphery Perspective on Individual Creative Performance: Social Networks and Cinematic Achievements in the Hollywood Film Industry, *Organization Science*, 19(6), 824-844.
- Chertow, M. R. 2000. Industrial Symbiosis: Literature and Taxonomy. Annual Review of Energy and Environment 25 313-337.
- Chertow, M. R. 2007. "Uncovering" Industrial Symbiosis. *Journal of Industrial Ecology* **11**(1) 11-30.
- Corbett, C. J., R. D. Klassen. 2006. Extending the Horizons: Environmental Excellence as Key to Improving Operations. *Manufacturing & Service Operations Management* 8(1) 5-22.
- Cross, R., A. Parker. 2004. The Hidden Power of Social Networks: Understanding How Work Really Gets Done in Organizations, Harvard Business School Press.
- Daly, H. E. 1991. Steady State Economics. 2nd edition, Island Press, Washington DC.
- Dawande, M., M. Mehrotra, V. Mookerjee, C. Sriskandarajah. 2010. An Analysis of Coordination Mechanisms for the U.S. Cash Supply Chain. *Management Science* **56**(3) 553-570.
- DeMatteis, J., A. Mendoza. 1968. An Economic Lot Sizing Technique. IBM Systems Journal 7 30-46.
- Drexl, A., A. Kimms. 1997. Lot Sizing and Scheduling Survey and Extensions. *European Journal of Operational Research* **99**(2) 221-235.
- Ehrenfeld, J. R. 1995. Industrial Ecology: A Strategic Framework for Product Policy and Other Sustainable Practices. *In Green goods*, edited by E. Ryden and J. Strahl. Stockholm.
- Ehrenfeld, J. R., N. Geretler. 1997. Industrial Ecology in Practice: The Evolution of Interdependence at Kalundborg. *Journal of Industrial Ecology* 1(1) 67-79.



Erickson, B. H. 1981. Secret Societies and Social Structure, Social Forces, 60(1), 188-210.

- Esenduran, G., E. Kemahlioglu-Ziya. 2009. Complying with Take-Back Legislation: A Cost Comparison and Benefit Analysis of Compliance Schemes. Working Paper.
- Esenduran, G., E. Kemahlioglu-Ziya, and J. Swaminathan. 2009. The Impact of Take-Back Legislation on Remanufacturing. Working Paper.
- Everett, M. G., S. P. Borgatti. 1999. The Centrality of Groups and Classes, Journal of Mathematical Sociology, 23(3), 181-201.
- Federal Reserve. 2003. Currency Recirculation Policy, docket no. OP-1164. Accessed: March 2010. http://www.federalreserve.gov/boarddocs/press/Other/2003/20031008/attachment.pdf.
- Federal Reserve. 2006. Federal Reserve Currency Recirculation Policy. Accessed: March 2010. http://www.federalreserve.gov/newsevents/press/other/other20060317a1.pdf.
- Federal Reserve. 2009. FedCash Services 2009 Fee Schedule. Accessed: March 2010. http://www.frbservices.org/servicefees/fedcash_services_2009.html.
- Federal Reserve. 2010a. Federal Funds Effective Rate. Accessed: March 2010. http://www.federalreserve.gov/releases/h15/data/Annual/H15_FF_O.txt.
- Federal Reserve. 2010b. Currency Processing. Accessed: March 2010. http://www.frbservices.org/operations/currency/currency.html.
- Federal Reserve. 2010c. Operating Circular 2 References. Accessed: March 2010. http://www.frbservices.org/operations/currency/oc2_reference.html
- Ferguson, M. E., D. Guide, G. C. Souza. 2006. Supply Chain Coordination for False Failure Returns. Manufacturing & Service Operations Management 8(4) 376-393.
- Ferguson, M. E., L. B. Toktay. 2006. The Effect of Competition on Recovery Strategies. Production and Operations Management 15(3) 351-368.
- Ferrer, G., J. M. Swaminathan. 2006. Managing New and Remanufactured Products. Management Science 52(1) 15-26.



- Fleischmann, M. 2003. Reverse Logistics Network Structures and Design, in V. D. R. Guide and L. N. Van Wassenhove, eds. Business Aspects of Closed-Loop Supply Chains: Exploring the Issues. Carnegie Bosch Institute, Pittsburgh, PA. 117-148.
- Fleischmann, M., J. Bloemhof-Ruwaard, R. Dekker, E. van der Laan, J. A. van Nunen, L. N. van Wassenhove. 1997. Quantitative Models for Reverse Logistics: A Review. *European Journal of Operational Research* 103 1-17.
- Forsyth, K., D. Haley, R. Kozak. 1999. Will Consumers Pay More for Certified Wood Products? Journal of Forestry 9(2) 19-22.
- Fowler, J. H. 2006. Legislative Cosponsorship Networks in the US House and Senate, Social Networks, 28, 454-465.
- Freeman, L. C. 1979. Centrality in Social Networks: Conceptual Clarification, Social Networks, 1(3), 215-239.
- Gallego, G., S. P. Sethi. 2005. \mathcal{K} -convexity in \Re^n . Journal of Optimization Theory and Applications 127(1) 71-88.
- Garey, M. R., D. S. Johnson. 1979. Computers and Intractability, A Guide to the Theory of NP-Completeness, Freeman, San Francisco, CA.
- Geismar, H. N., M. Dawande, D. Rajamani, C. Sriskandarajah. 2007. Managing a Bank's Currency Inventory under New Federal Reserve Guidelines. *Manufacturing & Service Op*erations Management. 9(2) 147-167.
- Geyer, R., L. N. Van Wassenhove, A. Atasu. 2007. The Economics of Remanufacturing under Limited Component Durability and Finite Product Life Cycles. *Management Science* 53(1) 88-100.
- Gloor, P. A., D. Oster, J. Putzke, K. Fischback, D. Schoder, K. Ara, T. J. Kim, R. Laubacher, A. Mohan, D. O. Olguin, A. Pentland, B. N. Waber. 2007. Studying Microscopic Peer-to-Peer Communication Patterns, *Americas Conference on Information Systems*, 2007.

Greene, K. 2008. 10 Emerging Technologies 2008, Technology Review, 2008.

Guide, D. 2000. Production Planning and Control for Remanufacturing: Industry Practice and Research Needs. *Journal of Operations Management* **18**(4) 467-483.



- Guide, D., V. Jayaraman, R. Srivastava, W. C. Benton. 2000. Supply Chain Management for Recoverable Manufacturing Systems. *INTERFACES* **30**(3) 125-142.
- Guide, D., V. Jayaraman. 2000. Product Acquisition Management: Current Industry Practice and a Proposed Framework. International Journal of Production Research 38(16) 3779-3800.
- Guide, D., V. Jayaraman, J. Linton. 2003. Building Contingency Planning for Closed-Loop Supply Chains with Product Recovery. *Journal of Operations Management* **21** 259-279.
- Guide, D., L. N. Van Wassenhove. 2001. Managing Product Returns for Remanufacturing. Production and Operations Management **10**(2) 142-155.
- Guide, D., L. N. Van Wassenhove. 2009. OR FORUM–The Evolution of Closed-Loop Supply Chain Research. *Operations Research* 57(1) 10-18.
- Hanaki, N., A. Peterhansl, P. S. Dodds, D. J. Watts. 2007. Cooperation in Evolving Social Networks, *Management Science*, 53(7), 1036-1050.
- Hawe, P., L. Ghali. 2008. Use of Social Network Analysis to Map the Social Relationships of Staff and Teachers at School, *Health Education Research*, **23**, 62-69.
- Hesseldahl, A. 2008. There's Gold in 'Reality Mining', Business Week, 23, March 24, 2008.
- Hill, S., F. Provost, C. Volinsky. 2006. Network-Based Marketing: Identifying Likely Adopters via Consumer Networks, *Statistical Science*, **21**(2), 256-276.
- Hoffman, A. J., J. B. Kruskal. 1956. Integral Boundary Points of Convex Polyhedra, Linear Inequalities and Related Systems, Annals of Mathematics Studies, 38, 223-246.
- Irland, L. 1993. Wood Products Face Green Marketing Era. Wood Technology 120(2) 34-36.
- Jelinski, L. W., T. E. Graedel, A. Laudise, D. W. McCall, C. K. Patel. 1992. Industrial Ecology: Concepts and Approaches. *Proceedings of the National Academy of Sciences* 89 793-797.
- Johnson, E. L. 1967. Optimality and Computation of (σ, S) Policies in the Multi-Item Infinite-Horizon Inventory Problem. *Management Science* **13** 475-491.



- Joneja, D. 1990. The Joint Replenishment Problem: New Heuristics and Worst Case Performance Bounds. *Operations Research* **38**(4) 711-723.
- Kalin, D. 1980. On the Optimality of (σ, S) Policies. *Mathematics of Operations Research* **5** 293-307.
- Kane, H. 1997. Eco-Farming in Fiji. World Watch 10(4) 29-34.
- King, A., M. Lenox. 2002. Exploring the Locus of Profitable Pollution Reduction. Management Science 48(2) 289-299.
- Klee, R. J. 1999. Zero Waste in Paradise. BioCycle 40(2) 66-67.
- Klee, R. J., R. Williams. 1999. Emerging International Eco-Industrial Projects. Casebook: Asia, The Pacific and Africa. Rep. US EPA Office of Policy, Planning and Evaluation. Washington, DC.
- Kleindorfer, P. R., K. Singhal, L. N. van Wassenhove. 2005. Sustainable Operations Management. *Production and Operations Management* 14(4) 482-492.
- Klovdahl, A. S., E. A. Graviss, A. Yaganehdoost, M. W. Ross, A. Wanger, G. J. Adams, J. M. Musser. 2001. Networks and Tuberculosis: an Undetected Community Outbreak Involving Public Places, *Social Science and Medicine*, **52**, 681-694.
- Kolaczyk, E. D., D. B. Chua, M. Barthelemy. 2009. Group Betweenness and Co-Betweenness: Inter-Related Notions of Coalition Centrality, *Social Networks*, **31**, 190-203.
- Korhonen, J. 2010. Industrial Ecology in the Strategic Sustainable Development Model: Strategic Applications of Industrial Ecology. *Journal of Cleaner Production* **12** 809-823.
- Kuik, R., M. Salomon, L. N. Van Wassenhove. 1994. Batching Decisions: Structure and Models. European Journal of Operational Research 75(2) 243-263.
- Lee, D. 2012. Turning Waste into By-Product. Manufacturing & Service Operations Management 14(1) 115-127.
- Liu, B., A. O. Esogbue. 1999. Decision Criteria and Optimal Inventory Processes, Kluwer, Boston, Massachusetts.



- Majumder, P., H. Groenevelt. 2001. Competition in Remanufacturing. *Production and Operations Management* **10**(2) 125-141.
- Mangan, A., E. Olivetti. (2010) By-Product Synergy Networks: Driving Innovation through Waste Reduction and Carbon Mitigation. J. Harmsen, J. B. Powell, eds. Sustainable Development in the Process Industries: Cases and Impact. John Wiley & Sons, Inc., Hoboken, NJ, USA.
- Martinez-Ros, E., J. M Labeaga. 2009. Product and Process Innovation: Persistence and Complementarities. *European Management Review* 6 6475.
- Mehrotra, M., M. Dawande, C. Sriskandarajah. 2010. A Depository Institution's Optimal Currency Supply Network under the Fed's New Guidelines: Operating Policies, Logistics, and Impact. Production and Operations Management 19(6) 709-724.
- Morselli, C., C. Giguere. 2006. Legitimate Strengths in Criminal Networks, Crime, Law and Social Change, 45, 185-200.
- Nemhauser, G. L., L. A. Wolsey. 1988. Integer Programming and Combinatorial Optimization, John Wiley & Sons, Inc., New York.
- Noronha, J. 1999. Scavengers and Decomposers in an Industrial Park System: A Case Study of Burnside Industrial Park. Master's thesis. Dalhousie University, Halifax, Nova Scotia.
- Owen-Smith, J., M. Riccaboni, F. Pammolli, W. W. Powell. 2002. A Comparison of U.S. and European University-Industry Relations in the Life Sciences, *Management Science*, 48(1), 24-43.
- Pentland, A. 2004. 'Reality Mining' the Organization, Technology Review, March, 2004.
- Perer, A., C. Wilson. 2007. The Steroids Social Network: An Interactive Feature on the Mitchell Report, *Slate Magazine*, available at http://www.slate.com/id/2180392/.
- Plambeck, E. L., Q. Wang. 2009. Effects of E-Waste Regulation on New Product Introduction. *Management Science* 55(3) 333-347.
- Posch, A., E. Schwarz, G. Steiner, H. Strebel, S. Vorbach. 1998. *The Recycling Network Styria and its Potential.*



- Puzis, R., Y. Elovici, S. Dolev. 2007. Fast Algorithm for Successive Computation of Group Betweenness Centrality, *Physical Review E*, 76(5), 056709.
- Rajamani, D., H. N. Geismar, C. Sriskandarajah. 2006. A Framework to Analyze Cash Supply Chains. Production and Operations Management 15(4) 544-552.
- Robinson, P., A. Narayanan, F. Sahin. 2009. Coordinated Deterministic Dynamic Demand Lot-Sizing Problem: A Review of Models and Algorithms. *Omega* **37**(1) 3-15.
- Rosenkranz, S. 2003. Simultaneous Choice of Process and Product Innovation when Consumers Have a Preference for Product Variety. *Journal of Economic Behavior & Organization* **50**(2) 183-201.
- Salomon, M., L. G. Kroon, R. Kuik, L. N. Van Wassenhove. 1991. Some Extensions of the Discrete Lot Sizing and Scheduling Problem. *Management Science* 37(7) 801-812.
- Savaskan, R. C., S. Bhattacharya, L. N. van Wassenhove. 2004. Closed-Loop Supply Chain Models with Product Remanufacturing. *Management Science* 50(2) 239-252.
- Scarf, H. 1960. The Optimality of (s, S) Policies in the Dynamic Inventory Problem, Mathematical Methods in the Social Sciences, Edited by J. Arrow, S. Karlin, and P. Suppes, Stanford University Press, Stanford, California, 196-202.
- Schilling, M. A., C. Phelps. 2007. Interfirm Collaboration Networks: The Impact of Large-Scale Network Structure on Firm Innovation, *Management Science*, 53(7), 1113-1126.
- Schwarz, E., K. Steininger. 1997. Implementing Nature's Lessons: The Industrial Recycling Network Enhancing Regional Development. Journal of Cleaner Production 5 47-56.
- Scott, J. 2000. Social Network Analysis: A Handbook, Second Edition, Sage Publications, Thousand Oaks, CA.
- Sedjo, R., S. Swallow. 2002. Voluntary Eco-Labeling and the Price Premium. Land Economics 78(2) 272-284.
- Skea, J. 1995. Environmental Technology. H. Folmer, H.L. Gabel, H. Opschoor, eds. Principles of Environmental and Resource Economics.



- Subramanian, R., S. Gupta, B. Talbot. 2008. Product Design and Supply Chain Coordination under Extended Producer Responsibility. *Production and Operations Management* 18(3) 259-277.
- Thompson, C. 2008. Real-World Social Networks vs. Facebook 'Friends', *Wired Magazine*, **16**(8), 2008.
- Toktay, L. B., L. M. Wein, S. A. Zenios. 2000. Inventory Management of Remanufacturable Products. *Management Science* **46**(11) 1412-1426.
- Van Berkel, R., T. Fujita, S. Hashimoto, M. Fuji. 2009. Quantitative Assessment of Urban and Industrial Symbiosis in Kawasaki (Japan). *Environmental Science and Technology* 43(5) 12711281.
- Wagner, H. M., T. M. Whitin. 1958. Dynamic Version of the Economic Lot Size Model. Management Science 5 89-96.
- Walley, N., B. Whitehead. 1994. Its Not Easy Being Green. *Harvard Business Review* **72**(3) 46-52.
- Wang, Z., L. Shi, D. Hu, D. Sun. 2010. Pursuing Sustainable Industrial Development through the Ecoindustrial Parks. Annals of the New York Academy of Sciences 1195 E145-E153.
- Wasserman, S., K. Faust. 1994. Social Network Analysis: Methods and Applications, Cambridge University Press.
- Wein, L. 2009. Homeland Security: From Mathematical Models to Policy Implementation: The 2008 Philip McCord Morse Lecture, *Operations Research*, **57**(4), 801-811.
- Wellman, B., W. Chen, W. Dong. 2001. Networking Guanxi, Social Networks in China: Institutions, Culture, and the Changing Nature of Guanxi, Cambridge University Press.
- Wheeler, A. 1968. Multiproduct Inventory Models with Set-up. Technical report. Stanford University, Stanford.
- Wustenhagen R., J. Markard, B. Truffer. 2003. Diffusion of Green Power Products in Switzerland. Energy Policy 31(7) 621-632.



VITA

Yunxia Zhu was born in Fuzhou, Fujian, China, to Mr. Xiaohong Zhu and Ms. Haiqin He. He received his Bachelor of Science in Operations Research from Fudan University, China, in 2001. After graduation, he worked for an international trade company in Shanghai, China for a few years. In Fall 2004, he joined the Weatherhead School of Management, Case Western Reserve University. In 2006, he received his Master of Science degree in Supply Chain Management from Case Western Reserve University. In Spring 2007, he joined School of Management, The University of Texas at Dallas (UTD) to pursue his Ph.D in Operations Management. In 2010, he received his MS degree from UTD. At UTD, he taught the undergraduate core course in Operations Management in Spring 2011, 2012, and Summer 2011, and received the Outstanding Student Teacher Award for the 2010-2011 academic year from the School of Management. His papers have been accepted for publication in Manufacturing & Service Operations Management, and INFORMS Journal on Computing.

